MICROTUNE INC Form DEF 14A April 09, 2010

Filed by the Registrant x

# **UNITED STATES**

# SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# **SCHEDULE 14A**

Proxy Statement Pursuant to Section 14(a) of the Securities

Exchange Act of 1934 (Amendment No. )

Filed by a Party other than the Registrant "		
Check the appropriate box:		
" Preliminary Proxy Statement		
" Confidential, for Use of the Commission Only (as permitted by Rule 14a-6(e)(2))	)	
x Definitive Proxy Statement		
" Definitive Additional Materials		
" Soliciting Material Pursuant to §240.14a-12		
MICROTUNE, INC	<b>7.</b>	

(Name of Registrant as Specified In Its Charter)

 $(Name\ of\ Person(s)\ Filing\ Proxy\ Statement, if\ other\ than\ the\ Registrant)$ 

Pay	Payment of Filing Fee (Check the appropriate box):		
x	No f	ee required.	
	Fee	computed on table below per Exchange Act Rules 14a-6(i)(1) and 0-11.	
	(1)	Title of each class of securities to which transaction applies:	
	(2)	Aggregate number of securities to which transaction applies:	
	(3)	Per unit price or other underlying value of transaction computed pursuant to Exchange Act Rule 0-11 (set forth the amount on which	
		the filing fee is calculated and state how it was determined):	
	(4)	Proposed maximum aggregate value of transaction:	
	(5)	Total fee paid:	

Fee <sub>1</sub>	paid previously with preliminary materials:
	ck box if any part of the fee is offset as provided by Exchange Act Rule 0-11(a)(2) and identify the filing for which the offsetting fee paid previously. Identify the previous filing by registration statement number, or the Form or Schedule and the date of its filing.
(1)	Amount Previously Paid:
(2)	Form, Schedule or Registration Statement No:
(3)	Filing party:
(4)	Date filed:

2201 10th S	Street
-------------	--------

Plano, Texas 75074		
Dear Stockholder:		
You are cordially invited to attend the 2010 Annual Meeting of Stockholders of Microtune, Inc., which will be held on Thursday, May 20, 2010 at 2:00 p.m. (Central Daylight Time) at the Homewood Suites by Hilton, 2601 E. George Bush, Plano, TX 75074.		
Details of the business to be conducted at the Annual Meeting are given in the attached Notice of Annual Meeting of Stockholders and Proxy Statement.		
We are complying with the U.S. Securities and Exchange Commission rule that allows companies to furnish their proxy materials over the Internet. As a result, we are mailing to our stockholders a notice regarding the Internet availability of the proxy materials, instead of a paper copy of this Proxy Statement and our 2009 Annual Report. The notice contains instructions regarding how to access this Proxy Statement and our 2009 Annual Report over the Internet. The notice also contains instructions regarding how each stockholder can receive a paper copy of our proxy materials, including this Proxy Statement, our 2009 Annual Report and a form of proxy card or voting instruction card. This process reduces the printing and postage costs of distributing paper copies of our proxy materials.		
After careful consideration, our Board of Directors has approved the proposals set forth in the Proxy Statement and recommends that you vote for each proposal.		
Whether or not you plan to attend the Annual Meeting, it is important that your shares be represented and voted at the meeting. Therefore, we urge you to vote promptly. You may vote electronically over the Internet or by telephone, or, if you requested and received paper copies of the proxy materials by mail, you may also vote by mail by following the instructions on the proxy card or voting instruction card. Timely voting by any of these methods will ensure your representation at the Annual Meeting.		
We look forward to seeing you at the 2010 Annual Meeting of Stockholders of Microtune. Inc.		

Sincerely, James A. Fontaine Chief Executive Officer and President

Plano, Texas

April 9, 2010

## YOUR VOTE IS IMPORTANT

In order to ensure your representation at the meeting, you are requested to vote electronically over the Internet or by telephone as promptly as possible. If you requested and received paper copies of the proxy materials by mail, you may also vote by mail by following the instructions on the proxy card or voting instruction card.

STOCKHOLDERS WHO HAVE E-MAIL ACCOUNTS CAN ELECT TO RECEIVE MICROTUNE S ANNUAL REPORTS AND PROXY MATERIALS THROUGH OUR ONLINE DELIVERY SERVICE. TO ENROLL IN OUR ONLINE DELIVERY SERVICE, PLEASE FOLLOW THE INSTRUCTIONS AT WWW.ICSDELIVERY.COM/MICROTUNE.

#### MICROTUNE, INC.

## NOTICE OF ANNUAL MEETING OF STOCKHOLDERS

#### TO BE HELD THURSDAY, MAY 20, 2010

## **DEAR STOCKHOLDERS:**

The 2010 Annual Meeting of Stockholders of Microtune, Inc. will be held on Thursday, May 20, 2010 at 2:00 p.m. (Central Daylight Time) at the Homewood Suites by Hilton, 2601 E. George Bush, Plano, TX 75074. Only stockholders of record at the close of business on March 24, 2010 will be entitled to vote. At our Annual Meeting, we will ask our stockholders to act on the following matters:

- 1. Elect nominees to our Board of Directors to serve until the 2011 Annual Meeting of Stockholders. The nominees are (i) incumbent members James H. Clardy, Steven Craddock, James A. Fontaine, Anthony J. LeVecchio, Bernard T. Marren and A. Travis White and (ii) Robert Rast, Raghavendra (Raghu) Rau and Drew Peck;
- 2. Approve the Microtune, Inc. 2010 Stock Plan;
- 3. Approve the Microtune, Inc. 2010 Director Stock Plan;
- 4. Approve the Microtune, Inc. 2010 Employee Stock Purchase Plan;
- 5. Ratify the appointment of KPMG LLP as our independent auditor for the fiscal year ending December 31, 2010; and
- 6. Transact any other business that is properly presented at our Annual Meeting, or any adjournment or postponement of our Annual Meeting.

We have described each of these matters in more detail in the enclosed Proxy Statement that accompanies this Notice.

All stockholders are cordially invited to attend the meeting in person. Any stockholder attending the meeting may vote in person even if such stockholder has previously voted electronically over the Internet, by telephone or by returning a proxy card, provided that he has a valid proxy card in his possession.

Sincerely,
James A. Fontaine
Chief Executive Officer and President

Plano, Texas

April 9, 2010

IMPORTANT: REGARDLESS OF THE NUMBER OF SHARES YOU OWN OR WHETHER OR NOT YOU PLAN TO ATTEND THE ANNUAL MEETING, PLEASE VOTE ELECTRONICALLY OVER THE INTERNET OR BY TELEPHONE IN ORDER TO ENSURE REPRESENTATION OF YOUR SHARES. IF YOU REQUESTED AND RECEIVED PAPER COPIES OF THE PROXY MATERIALS BY MAIL, YOU MAY ALSO VOTE BY MAIL BY FOLLOWING THE INSTRUCTIONS ON THE PROXY CARD OR VOTING INSTRUCTION CARD. FOR SPECIFIC INSTRUCTIONS ON HOW TO VOTE YOUR SHARES, PLEASE REFER TO PAGES 5-6 OF THIS PROXY STATEMENT.

#### MICROTUNE, INC.

#### PROXY STATEMENT FOR

## 2010 ANNUAL MEETING OF STOCKHOLDERS

May 20, 2010

#### General

The Board of Directors of Microtune, Inc. is asking for your proxy for use at our 2010 Annual Meeting of Stockholders, and at any adjournment or postponement of our Annual Meeting. We are holding the Annual Meeting on Thursday, May 20, 2010 at 2:00 p.m. (Central Daylight Time) at the Homewood Suites by Hilton, 2601 E. George Bush, Plano, TX 75074. We are initially mailing a notice regarding the Internet availability of the proxy materials on or about April 9, 2010 to our stockholders entitled to vote at the Annual Meeting. Directions to attend our Annual Meeting can be found on our website at www.microtune.com.

#### INFORMATION ABOUT THE MICROTUNE ANNUAL MEETING

- Q: Why am I receiving these materials?
- A: Our Board of Directors is providing these proxy materials to you in connection with our Annual Meeting of Stockholders, which will take place on May 20, 2010. Our stockholders are invited to attend the Annual Meeting and are requested to vote on the proposals described in this Proxy Statement.
- Q. What is included in the proxy materials?
- A: The proxy materials include:

Our Proxy Statement for the 2010 Annual Meeting of Stockholders; and

Our 2009 Annual Report on Form 10-K for the fiscal year ended December 31, 2009.

If you requested and received a paper copy of these materials by mail, the proxy materials also include a proxy card or a voting instruction card for the Annual Meeting.

Q. Why did I receive a notice in the mail regarding the Internet availability of the proxy materials instead of a paper copy of the proxy materials?

A. We are complying with the U.S. Securities and Exchange Commission (the SEC) rule that allows companies to furnish their proxy materials over the Internet. As a result, we are mailing to our stockholders a notice about the Internet availability of the proxy materials instead of a paper copy of the proxy materials. All stockholders receiving the notice will have the ability to access the proxy materials over the Internet and request to receive a paper copy of the proxy materials by mail. Instructions regarding how to access the proxy materials over the Internet or to request a paper copy can be found on the notice. In addition, the notice contains instructions regarding how stockholders may request to receive proxy materials electronically by e-mail on an ongoing basis.

## Q. How can I access the proxy materials over the Internet?

A. Your notice about the Internet availability of the proxy materials contains instructions regarding how to:

View our proxy materials for the Annual Meeting on the Internet;

Request a paper copy of our proxy materials for the Annual Meeting; and

Instruct us to send our future proxy materials to you electronically by e-mail.

1

If you choose to receive future proxy materials by e-mail, you will receive an e-mail with instructions containing a link to the website where those materials are available and a link to the proxy voting website. Your election to receive proxy materials by e-mail will remain in effect until you terminate it.

## Q. How may I obtain a paper copy of the proxy materials?

- A. Stockholders receiving a notice about the Internet availability of the proxy materials will find instructions regarding how to obtain a paper copy of the proxy materials in their notice.
- Q: What should I do if I receive more than one notice about the Internet availability of the proxy materials or more than one paper copy of the proxy materials?
- A: You may receive more than one notice. For example, if you hold your shares in more than one brokerage account, you may receive a separate notice for each brokerage account in which you hold shares. If you are a stockholder of record and your shares are registered in more than one name, you may receive more than one notice. To vote all of your shares by proxy, you must vote over the Internet or by telephone the shares represented by each notice that you receive (unless you have requested and received a proxy card or voting instruction card for the shares represented by one or more of those notices).

### Q: What proposals will be voted on at the Annual Meeting?

A: The following five proposals are scheduled to be voted on at the Annual Meeting:

Proposal No. 1: Elect nominees to our Board of Directors to serve until the 2011 Annual Meeting of Stockholders. The nominees are (i) incumbent members James H. Clardy, Steven Craddock, James A. Fontaine, Anthony J. LeVecchio, Bernard T. Marren and A. Travis White and (ii) Robert Rast, Raghavendra (Raghu) Rau and Drew Peck;

Proposal No. 2: Approve the Microtune, Inc. 2010 Stock Plan;

Proposal No. 3: Approve the Microtune, Inc. 2010 Director Stock Plan;

Proposal No. 4: Approve the Microtune, Inc. 2010 Employee Stock Purchase Plan; and

Proposal No. 5: Ratify the appointment of KPMG LLP as our independent auditor for the fiscal year ending December 31, 2010.

Also, the Annual Meeting may involve the consideration of such other business as may properly come before the meeting or any adjournments or postponements thereof.

## Q: What is Microtune s voting recommendation?

A: Our Board of Directors recommends that you vote your shares as follows:

FOR	all nominees for election as directors under Proposal No. 1;
FOR	Proposal No. 2;
FOR	Proposal No. 3;
FOR	Proposal No. 4; and
FOR	Proposal No. 5.

#### Q: How many shares represented does Microtune need to hold the Annual Meeting?

A: We need a quorum to take action at our Annual Meeting. We will have a quorum at our Annual Meeting if a majority of the shares issued and outstanding on the record date and entitled to vote are present, either in person or by proxy. If by the date of our Annual Meeting we do not have sufficient shares represented to constitute a quorum or to approve one or more of the proposals, the Chairperson of our Annual Meeting, or the persons named as proxies, may propose one or more adjournments of our Annual Meeting to permit further solicitation of proxies. The persons named as proxies would generally exercise their authority to vote in favor of adjournment. Your shares will be counted as present at the Annual Meeting if you:

are present and vote in person at the Annual Meeting; or

have properly voted by proxy, electronically, or by telephone.

Both abstentions and broker non-votes are counted as present for the purpose of determining the presence of a quorum. Broker non-votes occur when shares held by a broker on behalf of a beneficial owner are not voted with respect to a particular proposal, which generally occurs when the broker has not received voting instructions from the beneficial owner and lacks the discretionary authority to vote the shares itself.

## Q: Who can vote at the Annual Meeting?

A: Our Board of Directors has set March 24, 2010 as the record date for the Annual Meeting. All stockholders who owned shares of our common stock on March 24, 2010 may attend and vote at the Annual Meeting. Each of these stockholders is entitled to one vote on all matters to be voted on for each share of common stock held as of such date. On March 24, 2010, 54,002,294 shares of our common stock were outstanding. For information regarding security ownership by management and more than 5% stockholders, see Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters beginning on page 41.

## Q: Which shares that I own can be voted?

A: All shares owned by you as of the close of business on March 24, 2010, the record date, may be voted by you if either (1) you held these shares directly in your name as the stockholder of record, or (2) these shares were held for you as the beneficial owner through a broker, bank or other nominee.

## Q: What is the difference between holding shares as a stockholder of record and as a beneficial owner?

A: Most stockholders of Microtune hold their shares through a broker, bank or other nominee rather than directly in their own name. As summarized below, there are some distinctions between shares held of record and those owned beneficially.

### Stockholder of Record:

If your shares are registered directly in your name with our transfer agent, Computershare Shareholder Services, you are considered the stockholder of record with respect to those shares. As the stockholder of record, you have the right to grant your voting proxy directly to us by following the instructions on the notice regarding Internet availability of the proxy materials or to vote in person at the Annual Meeting.

## Beneficial Owner:

If your shares are held in a stock brokerage account or by a bank or other nominee, you are considered the beneficial owner of shares held in street name, and your broker, banker or nominee is considered the stockholder of record with respect to those shares. As the beneficial owner, you have the right to direct your broker, bank or nominee regarding how to vote your shares and are also invited to attend the Annual

Meeting. However, since you are not the stockholder of record, you may not vote these shares in person at the Annual Meeting, unless you request, complete and deliver a legal proxy from your broker, bank or nominee. You may also vote by following the instructions on the notice regarding Internet availability of the proxy materials.

#### Q: What vote is required for each item?

A: **Proposal No. 1: Election of Directors.** A plurality of the votes cast at the Annual Meeting is required to elect each nominee. Accordingly, abstentions and broker non-votes will have no effect on the election of directors. Shares represented by proxies will be voted for the election of the nominees named below unless authority to do so is specifically withheld. If any nominee declines to serve or becomes unavailable for any reason, or if a vacancy occurs before the election (although we know of no reason to anticipate that this will occur), the proxies may be voted for such substitute nominees as the Board of Directors may designate.

If you hold your shares through a broker, bank or other nominee, it is critical that you cast your vote if you want it to count in the election of directors (Proposal 1). In the past, if you held your shares through a broker, bank or other nominee and you did not indicate how you wanted your shares voted in the election of directors, your bank, broker or other nominee was allowed to vote those shares on your behalf in the election of directors as they felt appropriate.

Recent changes in the regulations were made to take away the ability of your bank, broker or other nominee to vote your uninstructed shares in the election of directors on a discretionary basis. Thus, if you hold your shares through a bank, broker or other nominee and you do not instruct your bank, broker or other nominee how to vote in the election of directors, no votes will be cast on your behalf. Your bank, broker or other nominee will, however, continue to have discretion to vote any uninstructed shares on the ratification of our independent auditor (Proposal 5). If you are a stockholder of record and you do not cast your vote, no votes will be cast on your behalf on any of the items of business at the Annual Meeting.

**Proposal No. 2: Approval of the Microtune, Inc. 2010 Stock Plan.** The affirmative vote of the holders of a majority of the shares of our common stock present in person or represented by proxy and entitled to vote is required to approve the Microtune, Inc. 2010 Stock Plan. An abstention will be counted as a vote against approval since it is one less vote for approval of the shares present. Broker non-votes will not affect the outcome since they are not considered shares present for voting purposes. Brokers do not have discretionary authority to vote shares they hold on the behalf of a beneficial holder in favor of or against this proposal if they have not been instructed to do so by such beneficial holder. Shares represented by proxies will be voted for the approval of the Microtune, Inc. 2010 Stock Plan, unless authority to do so is specifically withheld.

**Proposal No. 3: Approval of the Microtune, Inc. 2010 Director Stock Plan.** The affirmative vote of the holders of a majority of the shares of our common stock present in person or represented by proxy and entitled to vote is required to approve the Microtune, Inc. 2010 Director Stock Plan. An abstention will be counted as a vote against approval since it is one less vote for approval of the shares present. Broker non-votes will not affect the outcome since they are not considered shares present for voting purposes. Brokers do not have discretionary authority to vote shares they hold on the behalf of a beneficial holder in favor of or against this proposal if they have not been instructed to do so by such beneficial holder. Shares represented by proxies will be voted for the approval of the Microtune, Inc. 2010 Director Stock Plan, unless authority to do so is specifically withheld.

*Proposal No. 4: Approval of the Microtune, Inc. 2010 Employee Stock Purchase Plan.* The affirmative vote of the holders of a majority of the shares of our common stock present in person or represented by proxy and entitled to vote is required to approve the Microtune, Inc. 2010 Employee Stock Purchase Plan. An abstention will be counted as a vote against approval since it is one less vote for approval of the shares

present. Broker non-votes will not affect the outcome since they are not considered shares present for voting purposes. Brokers do not have discretionary authority to vote shares they hold on the behalf of a beneficial holder in favor of or against this proposal if they have not been instructed to do so by such beneficial holder. Shares represented by proxies will be voted for the approval of the Microtune, Inc. 2010 Employee Stock Purchase Plan, unless authority to do so is specifically withheld.

**Proposal No. 5: Ratification of Appointment of Our Independent Auditor.** Ratification of KPMG LLP as our independent registered public accounting firm for the fiscal year ending December 31, 2010 requires the affirmative vote of the holders of a majority of the shares of common stock present in person or represented by proxy and entitled to vote. An abstention will be counted as a vote against approval since it is one less vote for approval of the shares present. Broker non-votes will not affect the outcome since they are not considered shares present for voting purposes; however, brokers will have authority to vote shares they hold on the behalf of beneficial holders in favor of approval if they have not been instructed otherwise. Shares represented by proxies will be voted for the ratification of KPMG LLP as our independent registered public accounting firm for the fiscal year ending December 31, 2010, unless authority to do so is specifically withheld.

### Q: How are votes counted?

A:	You may vote either FOR each	nominee for election to our B	oard of Directors under Proposal No. 1 or to	WITHHOLD your vote for that
	nominee. You may vote FOR,	AGAINST or ABSTAIN	on the other proposals. If you vote your prov	ky with no further instructions, your
	shares will be counted as a vote:			

FOR each of the nominees for election as directors under Proposal No. 1;

FOR Proposal No. 2;

FOR Proposal No. 3;

FOR Proposal No. 4; and

FOR Proposal No. 5.

A plurality of the votes cast at the Annual Meeting is required to elect each nominee under Proposal No. 1. Proposal No. 2, Proposal No. 3, Proposal No. 4 and Proposal No. 5 require the affirmative vote of the holders of a majority of the shares of common stock present in person or represented by proxy and entitled to vote.

## Q: Who will count the vote?

- A: A representative of Broadridge Financial Solutions, Inc. will tabulate the votes and act as the inspector of election at the Annual Meeting.
- Q: How can I vote my shares in person at the Annual Meeting?
- A: Shares held directly in your name as the stockholder of record may be voted in person at the Annual Meeting. If you choose to do so, please bring your notice of Internet availability of the proxy materials (or, your proxy card, if you requested and received paper copies

of the proxy materials by mail) and proof of identification to the Annual Meeting.

If you hold your shares in street name, you must request a legal proxy card from your broker, bank or nominee in order to vote in person at the Annual Meeting. Please note that if you hold your shares in street name, you will NOT be able to vote in person if you bring only your notice of Internet availability of the proxy materials (or, your voter instruction card, if you requested and received paper copies of the proxy materials by mail) to the Annual Meeting. You must have a legal proxy card to vote your shares in person at the Annual Meeting.

#### Q: How can I vote my shares without attending the Annual Meeting?

A: Whether you hold your shares directly as the stockholder of record or through a broker, bank or other nominee as the beneficial owner, you may direct how your shares are voted without attending the Annual Meeting. There are three ways to vote by proxy:

**By Internet** Stockholders may vote over the Internet by following the instructions on their notice of Internet availability of the proxy materials. Stockholders who have requested and received a paper copy of a proxy card or voting instruction card by mail may also vote over the Internet by following the instructions on the proxy card or voting instruction card.

**By Telephone** Stockholders of record may vote by telephone by calling 1 (800) 690-6903 and following the instructions. Stockholders of record will need to have the control number that appears on their notice of Internet availability of the proxy materials available when voting. Stockholders of record who have requested and received a proxy card by mail may also use the control number that appears on their proxy card when voting. Stockholders who are beneficial owners of their shares should check their notice of Internet availability of the proxy materials for telephone voting availability and instructions.

**By Mail** Stockholders who have requested and received a paper copy of a proxy card or voting instruction card by mail may submit proxies by completing, signing and dating their proxy card or voting instruction card and mailing it in the accompanying pre-addressed envelope.

### Q: How can I change my vote after I return my proxy?

A: You may change your vote at any time prior to the vote at the Annual Meeting. If you are a stockholder of record, you may change your vote by granting a new proxy bearing a later date (which automatically revokes any earlier proxy), by providing a written notice of revocation to our Corporate Secretary, Phillip D. Peterson, prior to the time your shares are voted, or by attending the Annual Meeting and voting in person. Attendance at the meeting will not cause your previously granted proxy to be revoked unless you specifically so request.

For shares you hold beneficially in street name, you may change your vote by submitting new voting instructions to your broker, bank or nominee, or, if you have obtained a legal proxy card from your broker, bank or nominee giving you the right to vote your shares, by attending the meeting and voting in person.

## Q: Where can I find the voting results of the Annual Meeting?

A: The preliminary voting results will be announced at the Annual Meeting. The final results will be published in a Current Report on Form 8-K, filed as soon as practicable after the Annual Meeting.

#### Q: What happens if additional proposals are presented at the Annual Meeting?

A: Other than the five proposals described in this Proxy Statement, we do not expect any matters to be presented for a vote at the Annual Meeting. If you grant a proxy, the persons named as proxy holders, James A. Fontaine, our Chief Executive Officer and President, and Phillip D. Peterson, our General Counsel and Corporate Secretary, will have discretion to vote your shares on any additional matters properly presented for a vote at the Annual Meeting.

#### Q: May I propose actions for consideration at next year s Annual Meeting of Stockholders?

A: You may submit proposals for consideration at future annual stockholder meetings in accordance with the procedures set forth below:

Proposals of our stockholders that are intended for inclusion in our proxy materials relating to the 2011 Annual Meeting of Stockholders must be received by us at our principal executive offices no later than 120 days prior to the anniversary of the date of this year s mailing, or December 9, 2010, and must satisfy the conditions established by the SEC, including, but not limited to, Rule 14a-8 promulgated under the Securities Exchange Act of 1934, as amended, and in our bylaws for stockholder proposals in order to be included in our proxy materials for that meeting.

Proposals of our stockholders that are not intended to be included in our proxy materials for the 2011 Annual Meeting of Stockholders but that are intended to be presented by the stockholder at such meeting must be delivered to or mailed and received at our principal executive offices no later than 120 calendar days prior to the anniversary of the date of the 2010 Annual Meeting of Stockholders; provided, however, if the date of the 2011 Annual Meeting of Stockholders has been changed by more than 30 days from the date contemplated at the time of this year s Annual Meeting, notice by the stockholder to be timely must be so delivered not earlier than the close of business on the later of the day 120 days prior to such meeting or the day 10 days after public announcement of the date of the meeting is first made.

#### Q: Who bears the costs of soliciting proxies?

A: Microtune will pay all expenses of soliciting proxies to be voted at our Annual Meeting. After the proxies are initially distributed, Microtune and/or its agents may also solicit proxies by mail, telephone or in person. We have not engaged a proxy solicitor and therefore do not expect to incur any proxy solicitation fees. After the notices regarding Internet availability of the proxy materials are initially distributed, we will ask brokers, custodians, nominees and other record holders to forward copies of the proxy materials to people for whom they hold shares of our common stock that request a paper copy of the proxy materials. We will reimburse record holders for reasonable expenses they incur in forwarding paper copies of the proxy materials to beneficial holders. We expect to incur approximately \$25,000 for services provided by Broadridge Financial Solutions, Inc. for distribution of notices, proxies, Internet and telephone voting, voting tabulation and voting reports.

### Q: How can I sign up to access future stockholder communications electronically?

A: All stockholders who have e-mail accounts can elect to access Microtune s annual reports and proxy materials online through our online delivery service.

To enroll in our online program, simply go to www.icsdelivery.com/microtune and follow the instructions.

#### MICROTUNE CORPORATE GOVERNANCE

#### **Corporate Governance**

Our Board of Directors and members of management are committed to following high standards of corporate governance to ensure that the long term interests of our stockholders are served. We have in place a variety of policies and practices to promote high standards of corporate governance. Our Board of Directors has determined that, with the exception of James A. Fontaine, all of our current directors, and all director nominees, with the exception of Drew Peck, are independent in accordance with the rules of The NASDAQ Stock Market, and all of the members of our Audit Committee, Compensation Committee and Nominating and Corporate Governance Committee also meet the guidelines of The NASDAQ Stock Market for independence. Compensation of our Chief Executive Officer is approved by our Board of Directors, upon the recommendation of our Compensation Committee, which evaluates our CEO s performance in light of preset corporate goals and objectives. In addition, we have established:

disclosure control policies and procedures in accordance with the requirements of the Sarbanes-Oxley Act of 2002 (the Sarbanes-Oxley Act );

an Equity Compensation Award Policy to help ensure that grants of equity awards will comply with applicable legal and accounting guidance;

a procedure for receipt, retention and treatment of anonymous and confidential complaints or concerns regarding accounting, internal controls and auditing matters; and

a Code of Ethics and Business Conduct applicable to our directors, officers and employees.

Copies of our Code of Ethics and Business Conduct and the current charters for the Audit, Compensation and Nominating and Corporate Governance Committees can be found in the Investors Corporate Governance section of our website at www.microtune.com.

Corporate Governance Provisions Adopted as a Result of the Settlement of Stockholder Derivative Litigation in 2005. On January 11, 2005, we announced that we had reached an agreement to settle the consolidated stockholder derivative litigation, pending in the U.S. District Court for the Eastern District of Texas, against a number of our current and former officers and directors and Microtune as a nominal defendant. The court approved the terms of the derivative litigation settlement and issued a final judgment and order dismissing the action with prejudice on March 31, 2005. Pursuant to paragraph 4(a) of the Stipulation and Agreement of Settlement, dated January 10, 2005, we have adopted certain corporate governance provisions. Some of the more notable provisions are as follows:

we recommended and presented a proposal to our stockholders at the 2005 Annual Meeting to amend our Second Amended and Restated Certificate of Incorporation and our Second Amended and Restated Bylaws to declassify our Board of Directors such that each director now serves a one-year term and stands for election or reelection every year;

we adopted a variety of compensation principles for our executive officers and directors, including linking compensation to performance, requiring full board approval of any discretionary bonuses and implementing restrictions on stock sales following the exercise of options;

our Board of Directors, as a whole, evaluates each member of the Board on a periodic basis and a consideration in such evaluation is whether each individual director has personally attended at least 50% of Board and committee meetings;

we designated a trading compliance officer who is responsible for overseeing a comprehensive insider trading compliance program designed to ensure compliance with our trading policies;

we vest equity awards granted to any director or executive officer over a minimum of a three-year period;

8

we have designated a corporate ethics officer who is responsible for devising, implementing and improving our ethical standards and for providing a vehicle for corporate employees to confidentially report suspected wrongdoing or non-compliance with our corporate policies and procedures;

all stockholder proposals that are properly submitted will be evaluated by the Nominating and Corporate Governance Committee; and our full Board of Directors will include a recommendation for or against such stockholder proposals and the reasons for such recommendation in the proxy statement;

our Board of Directors established a procedure for stockholder-nominated director elections and allowed stockholders, in accordance with this procedure, to nominate two directors to our Board who were both subsequently elected; and

we rotated our independent auditor by selecting KPMG LLP to perform the audit of our financial statements for the fiscal years ending December 31, 2008 and December 31, 2009.

These provisions must remain in effect for five years from the date of adoption, which was May 25, 2005, and will continue to be in effect thereafter until modified by the Board of Directors.

In addition to the foregoing, by majority vote of our independent directors, our Board of Directors elected A. Travis White as Lead Independent Director, who works with our Chief Executive Officer to perform a variety of functions related to our corporate governance, including coordinating the Board s activities, scheduling meetings of the full Board, scheduling executive sessions of the non-employee directors, setting relevant items on the agenda (in consultation with the Chief Executive Officer as necessary or appropriate) and ensuring adequate communication between the Board and management. The Board believes this leadership structure ensures the Board s oversight of, and independence from, our management, the ability of the Board to carry out its roles and responsibilities on behalf of our stockholders, and our overall corporate governance. Furthermore, this structure permits the Chief Executive Officer to focus on the management of the company s day-to-day operations.

Equity Award Policy. In connection with our Audit Committee's review of our internal stock option granting practices, we adopted an Equity Compensation Award Policy to enact certain improvements to our equity award granting procedures and our control environment, including our controls over the administration of and accounting for our equity compensation awards, which are described below in Compensation Discussion and Analysis-Long-Term Equity Incentive Compensation-Equity Awards and Practices. Among other things, the policy provides that any deviation from the policy must receive prior approval from our Board. Such Board approval may only be given after full consideration of the market timing issues and legal risks associated with any such deviation. The policy generally provides that concerns regarding the timing of any grant should be communicated to our General Counsel and Chief Financial Officer by the Chairperson of the Compensation Committee. Finally, the policy provides for documentation of any such deviation.

## **Board of Directors**

Our Board of Directors is currently comprised of nine members, of which Messrs. Clardy, Craddock, Fontaine, LeVecchio, Marren and White will stand for election at our 2010 Annual Meeting of Stockholders. Messrs. Ciciora, Schueppert and Tai previously notified the Board of Directors that they will not stand for re-election at the 2010 Annual Meeting. Our Board of Directors currently has no vacancies.

#### **Director Independence**

Our Board of Directors has determined that eight of our current Board members, Messrs. Ciciora, Clardy, Craddock, LeVecchio, Marren, Schueppert, Tai and White (excluding Mr. Fontaine), and two of our new nominees for director, Messrs. Rast and Rau (excluding Mr. Peck), are independent directors as defined under the current listing standards of The NASDAQ Stock Market. Mr. White was appointed Lead Independent Director of our Board of Directors on January 18, 2005, and has been reappointed every year thereafter.

## **Board Meetings and Committees; Annual Meeting Attendance**

Our Board and its committees have the authority to conduct investigations and to retain outside advisors of their choosing, at Microtune s expense. During fiscal year 2009, our Board of Directors met 13 times. Each director attended at least 75% of the aggregate of these Board meetings and of the meetings of the committees on which he served, except that Mr. Tai attended eight of the 13 Board meetings. Members of our Board of Directors are encouraged to attend our Annual Meeting. Last year, three members of our Board of Directors attended our Annual Meeting.

Our Board of Directors currently has four standing committees an Audit Committee, a Compensation Committee, a Nominating and Corporate Governance Committee and a Technical Advisory Committee.

Audit Committee. The Audit Committee s principal function is to assist the Board of Directors in fulfilling its responsibilities for oversight of:

the integrity of our financial statements;

the adequacy of our system of internal controls;

our compliance with legal and regulatory requirements;

the qualifications and independence of our independent auditor; and

the performance of our independent auditor and internal audit function.

In accordance with its charter, our Audit Committee is responsible for:

#### Financial Reporting

reviewing with management and our independent auditor the significant judgments and estimates used in developing our financial reports and the major issues addressed;

reviewing the accounting and reporting treatment of significant transactions outside our ordinary operations;

reviewing with management and our independent auditor significant changes to our accounting principles or their application as reflected in our financial reports;

meeting periodically with our independent auditor (outside of the presence of management, as appropriate):

to review its reasoning in accepting or questioning significant decisions made by management in preparing our financial reports;

to review any audit problems or difficulties and management s response;

to review any outstanding disagreements with management that would cause it to issue a non-standard report on our financial statements;

to examine the appropriateness of our accounting principles (including the quality, not just the acceptability, of accounting principles) and the clarity of disclosure practices used or proposed;

to determine if any restrictions have been placed by management on the scope of its audit; and

to discuss any other matters the Audit Committee deems appropriate;

reviewing and discussing our earnings press releases with management and our independent auditor, including the use of non-GAAP financial information;

reviewing (by the Audit Committee or its Chairperson) earnings press releases, as well as financial information and earnings guidance provided to analysts and rating agencies;

reviewing quarterly and annual financial statements and discussing their appropriateness with management and our independent auditor and, in the case of our annual financial statements, recommending to the Board of Directors whether the audited financial statements should be included in our Annual Report on Form 10-K; and

in connection with each quarterly and annual report, reviewing:

management s disclosure to the Audit Committee and our independent auditor under Section 302 of the Sarbanes-Oxley Act, including identified changes in internal control over financial reporting; and

the contents of the Chief Executive Officer and Chief Financial Officer certificates to be filed under Sections 302 and 906 of the Sarbanes-Oxley Act and the process conducted to support the certifications;

## Relationship with Independent Auditor

bearing primary responsibility for overseeing our relationship with our independent auditor. In carrying out this responsibility, the Audit Committee shall:

have sole authority to appoint, determine funding for and oversee our outside auditor;

be directly responsible for the compensation and oversight of the work of the independent auditor for the purpose of preparing or issuing an audit report or related work;

review the scope and plan for our independent auditor s annual audit and review and annually pre-approve, in advance, the fees to be charged by our independent auditor for its audit services;

review with our independent auditor the extent of non-audit services provided and related fees, and approve any internal control-related services and permitted non-audit services or other non-audit relationships;

at least annually, obtain and review a report from the independent auditor delineating all relationships between the independent auditor and Microtune, consistent with the applicable requirements of the Public Company Accounting Oversight Board for independent auditor communications with Audit Committees concerning independence;

determine whether the Audit Committee believes the outside auditor is independent;

at least annually, obtain and review a report by Microtune s independent auditor describing the independent auditor firm s internal quality-control procedures; material issues raised by the most recent internal quality-control review, or peer review, of the firm, or by any inquiry or investigation by governmental or professional authorities, within the preceding five years, respecting one or more independent audits carried out by the firm, and any steps taken to deal with any such issues; and (to assess the auditor s independence) all relationships between the independent auditor and Microtune;

inquire of management, internal auditors and the independent auditor concerning any deficiencies in the policies and procedures that could adversely affect the adequacy of internal controls and the financial reporting process and reviewing the timeliness and reasonableness of proposed corrective actions and the adequacy of disclosure about changes in internal control over financial reporting;

review significant management audit findings and recommendations, and management s responses thereto; and

inquire of our independent auditor annually as to whether any illegal act has been detected or has otherwise come to its attention in the course of its audit, unless the illegal act is clearly inconsequential;

11

#### Complaint Procedures and General Oversight by Audit Committee

establishing procedures for the receipt, retention and treatment of complaints received by Microtune regarding accounting, internal accounting controls or auditing matters, and the confidential, anonymous submission by employees of concerns regarding questionable accounting or auditing matters;

recommending to Microtune, as necessary, appropriate remedial measures or actions with respect to the complaints or concerns described immediately above;

discussing with management, Microtune s internal audit personnel and our independent auditor whether Microtune and its subsidiaries are in conformity with applicable legal requirements and Microtune s Code of Ethics and Business Conduct and advise the Board of Directors with respect to its policies and procedures regarding compliance with applicable laws and regulations and with the Code of Ethics and Business Conduct;

reviewing management s responses to recommendations for improving internal controls in the independent auditor s management letters;

reviewing Microtune s policies and practices with respect to risk assessment and risk management;

reviewing Microtune s policies and practices related to compliance with laws, ethical conduct and conflicts of interest;

reviewing all related party transactions for potential conflict of interest situations on an ongoing basis and approving any such transactions (the term related party transaction refers to transactions required to be disclosed pursuant to Item 404 of Regulation S-K); provided, any such transaction may be alternatively approved by another independent body of our Board;

reviewing material cases of conflicts of interest, misconduct or fraud;

reviewing material issues between Microtune and regulatory agencies; and

reviewing as appropriate material litigation involving Microtune;

## Approval of Reports

preparing and approving the Audit Committee s report required by the rules of the SEC to be included in the proxy statement for our annual meeting of stockholders, and such other reports as may from time to time be necessary or appropriate;

#### Annual Performance Review

conducting an annual evaluation of the Audit Committee s performance in carrying out its responsibilities; and

reviewing and reassessing the adequacy of the Audit Committee s charter annually and recommending any proposed changes to the Board of Directors for approval.

Mr. LeVecchio, Mr. Craddock and Mr. White are the current members of the Audit Committee. Mr. LeVecchio became a member of the Audit Committee in August 2003. Mr. Craddock and Mr. White became members of the Audit Committee in January 2004. The Audit Committee discusses general financial and accounting-related matters at its regular quarterly meetings that coincide with full Board meetings, and it discusses our quarterly financial performance and associated press releases describing our quarterly and annual financial results at such meetings. The Audit Committee met five times during fiscal year 2009. For more information about our Audit Committee, see the Report of the Audit Committee of the Board of Directors on page 65.

Our Board of Directors has determined that each of the members of the Audit Committee who served during fiscal year 2009 is independent as defined under NASDAQ Listing Rule 5605(a)(2), and met the independence requirements of Rule 10A-3(b)(1) of the Securities Exchange Act of 1934, as amended (Exchange

Act), as well as the requirements of NASDAQ Listing Rule 5605(c)(2). Mr. LeVecchio serves as Chairperson of the Audit Committee, and our Board of Directors has determined that Mr. LeVecchio qualifies as a financial expert as defined by the rules of the SEC. Mr. LeVecchio currently serves on the audit committees of three other public companies. However, our Board has noted the significant time expended by Mr. LeVecchio managing the Audit Committee s responsibilities and that his responsibilities to other boards have not impaired his ability to contribute extraordinary time and efforts to our Audit Committee. Our Board has therefore determined that Mr. LeVecchio s simultaneous service on three other audit committees in addition to our Audit Committee does not impair, and will not impair, the ability of Mr. LeVecchio to effectively serve as Chairperson of our Audit Committee.

Compensation Committee. The Compensation Committee reviews and makes recommendations to the Board of Directors regarding all forms of compensation to be provided to our executive officers and directors. In addition to addressing executive compensation, the Compensation Committee provides guidance to management on general compensation and organizational development issues and also administers our equity compensation plans. The philosophy of the Compensation Committee is to provide compensation to our officers and directors in such a manner as to attract and retain the best available personnel for positions of substantial responsibility, to provide incentives for such persons to perform to the best of their abilities, and to promote the success of our business. The Compensation Committee may retain or employ professional firms and experts to assist it in the discharge of its duties. Further, the Compensation Committee has the authority to appoint and delegate any of its duties to a sub-committee comprised of one or more Compensation Committee members, as it deems necessary and appropriate.

In accordance with its charter, the Compensation Committee is also responsible for:

determining our overall compensation philosophy;

reviewing and making determinations regarding the corporate goals and objectives relevant to the Chief Executive Officer s compensation, evaluating our Chief Executive Officer s performance in light of those goals and objectives, and recommending appropriate compensation to our Board of Directors based on the evaluation;

reviewing and making recommendations to the Board of Directors regarding our compensation policy for officers and directors;

reviewing and making recommendations to the Board of Directors regarding all forms of compensation to be provided to our officers;

reviewing all components of compensation for non-employee directors;

working with the Chief Executive Officer to review and make recommendations to the Board of Directors regarding general compensation goals for non-officer employees and the criteria by which bonuses are awarded to non-officer employees;

administering our equity compensation plans and making recommendations to the Board of Directors regarding any amendments to our equity compensation plans;

reviewing and approving any agreement to be entered into between Microtune and any employee that provides for a fixed term of employment or for severance or other termination benefits;

producing an annual report of executive compensation to be provided in the Annual Report on Form 10-K or proxy statement, as applicable;

reviewing and discussing the Compensation Discussion and Analysis disclosure prepared by management for inclusion in our Annual Report on Form 10-K and, as applicable, Proxy Statement, and once satisfied with the disclosure, recommending its inclusion in the Annual Report on Form 10-K and, as applicable, the Proxy Statement to our Board of Directors;

reviewing and evaluating its performance annually;

reviewing and reassessing the Compensation Committee charter and our Equity Compensation Award Policy; and

consulting with management regarding regulatory compliance with respect to compensation matters.

As part of our settlement of the consolidated derivative litigation in 2005, our Board of Directors agreed to give the Compensation Committee the following additional responsibilities:

evaluating our Chief Executive Officer s performance against annual and long-term performance goals and the Philadelphia Semiconductor Index:

coordinating evaluation of our Chief Executive Officer s and Chief Financial Officer s performance with our Lead Independent Director;

recommending Chief Executive Officer and executive officer bonus compensation for the last fiscal year to the Board of Directors, which is based on the achievement of goals set in advance (discretionary awards can be made only if beneficial to Microtune and approved by the Board of Directors);

annually reviewing current compensation packages with consideration given to performance goals and incentive awards and overall compensation packages; and

reviewing director compensation annually.

Our Board of Directors has determined that each of the members of the Compensation Committee is a *non-employee director*, as defined in Rule 16b-3 promulgated under the Exchange Act, and an *outside director*, as defined pursuant to Section 162(m) of the Internal Revenue Code of 1986, as amended (the Code ). Our Board of Directors has also determined that the members of the Compensation Committee who served during fiscal year 2009 are independent as defined under NASDAQ Listing Rule 5605(a)(2). Messrs. Clardy, Marren and Schueppert are the current members of the Compensation Committee and all served on the committee throughout fiscal 2009. Mr. Clardy is the chairperson of the Compensation Committee. The Compensation Committee met 12 times during fiscal 2009. For more information on our Compensation Committee, see our Compensation Committee Report on page 56.

*Nominating and Corporate Governance Committee*. The Nominating and Corporate Governance Committee is appointed by the Board of Directors to fulfill the following general responsibilities:

identifying individuals qualified to become directors;

recommending to the Board of Directors a slate of director nominees to be elected by the stockholders at the next annual meeting of stockholders and, when appropriate, director appointees to take office between annual meetings;

developing, reviewing and recommending to the Board of Directors the corporate governance guidelines applicable to Microtune;

leading the Board of Directors in its annual evaluation of the performance and independence of the Board and the performance of executive management; and

recommending to the Board of Directors membership on our standing board committees.

In accordance with its charter, the Nominating and Corporate Governance Committee is responsible for the following:

identifying and evaluating potential new Board members, providing information about potential nominees for the full Board of Directors to consider;

recommending the slate of qualified director nominees for the Board of Directors (based upon the Nominating and Corporate Governance Committee sassessment of their personal and professional integrity, experience, skills, independence, ability and willingness to devote the time and effort

necessary to be an effective board member and commitment to acting in the best interests of Microtune and our stockholders) in connection with each annual meeting and any directors to be elected by the Board to fill vacancies between annual meetings;

recommending committee assignments for directors;

evaluating the composition, organization and performance of the Board of Directors and its committees;

evaluating and making recommendations to the Board of Directors as to the determination of each director s independence;

reviewing and reporting to the Board of Directors regarding succession planning with respect to the Chief Executive Officer;

developing, recommending to the Board of Directors, and annually reviewing the corporate governance guidelines applicable to Microtune:

developing and recommending to the Board of Directors a self-evaluation process for the Board of Directors and its committees;

overseeing the Board of Directors and committee self-evaluation process and reporting its findings to the Board of Directors following the end of each fiscal year; and

conducting a preliminary review of the independence of each Board member and providing its findings and making recommendations to the full Board regarding the independence of each Board member.

Messrs. Ciciora, Clardy and White are the current members of the Nominating and Corporate Governance Committee and served on the committee throughout fiscal year 2009. Mr. White is the Chairperson of the Nominating and Corporate Governance Committee. The Nominating and Corporate Governance Committee met two times during fiscal year 2009.

Technical Advisory Committee. The Technical Advisory Committee was established by the Board of Directors on May 17, 2004. The Technical Advisory Committee is responsible for assisting management in the strategic development of new technology and the commercialization and marketing of existing technology. Messrs. Ciciora, Craddock and Tai are the current members of the Technical Advisory Committee and all served on the committee throughout fiscal year 2009. Messrs. Craddock and Ciciora are the Co-Chairpersons of the Technical Advisory Committee. The Technical Advisory Committee did not meet during fiscal year 2009. The Technical Advisory Committee does not have a formal written charter.

#### **Director Qualifications**

When evaluating director nominees, our Nominating and Corporate Governance Committee generally seeks to identify individuals with diverse, yet complementary backgrounds. Although we have no formal policy regarding diversity, our Nominating and Corporate Governance Committee considers both the personal characteristics and experience of director nominees, including each nominee s independence, diversity, age, skills, expertise, time availability, and industry backgrounds in the context of the needs of the Board and the company. The committee believes that director nominees should exhibit proven leadership capabilities and experience at a high level of responsibility within their chosen fields, and have the experience and ability to analyze complex business issues facing Microtune, specifically, those issues that are inherent in the fabless semiconductor industry. In addition to business expertise, the committee requires that director nominees have the highest personal and professional ethics, integrity and values and, above all, are committed to representing the long-term interests of our stockholders and other

stakeholders. In general, candidates who hold or who have held an executive-level position in an established high technology company are preferred.

When considering a candidate for director, the Nominating and Corporate Governance Committee takes into account a number of factors,

_	
i	independence;
	depth of understanding of technology, manufacturing, sales and marketing, finance and/or other elements directly relevant to our business;
6	education and professional experience;
j	judgment, integrity and reputation;
6	existing commitments to other businesses as a director, executive or owner;
1	personal conflicts of interest, if any; and
1	the size, composition and skills of the existing Board of Directors.

Prior to nominating a sitting director for re-election at an annual meeting of stockholders, the Nominating and Corporate Governance Committee will consider the director s past attendance at, and participation in, meetings of the Board of Directors and its committees and the director s formal and informal contributions to the work of the Board of Directors and its committees.

When seeking candidates for director, the Nominating and Corporate Governance Committee may solicit suggestions from incumbent directors, management, stockholders and others. Additionally, the committee may in the future use the services of third party search firms to assist in the identification of appropriate candidates. After conducting an initial evaluation of a prospective candidate, the committee will interview that candidate if it believes the candidate might be suitable to be a director. The committee may also ask the candidate to meet with management. If the committee believes a candidate would be a valuable addition to the Board of Directors, it may recommend to the Board that candidate s appointment or election.

#### Stockholder Recommendations for Nominations to the Board of Directors

including the following:

In connection with the settlement of our consolidated stockholder derivative litigation in 2005, we agreed to establish a procedure for stockholders to nominate two directors for election to our Board of Directors. The committee evaluated such recommendations by applying its regular nominee criteria described above and the additional procedures described below.

In 2005 and 2006, a corporate governance consultant selected by lead plaintiffs—counsel, working in conjunction with us, identified potential directors by contacting stockholders who had held 1% or more of our common stock for at least nine months to solicit names of candidates for election to our Board of Directors. Our Nominating and Corporate Governance Committee then reviewed the candidate—s qualifications and selected the stockholders—nominee. This procedure resulted in the appointment of Bernard T. Marren on August 30, 2005 and the appointment of Michael T. Schueppert on August 21, 2006, all in accordance with the derivative litigation settlement and our Third Amended and Restated

Bylaws.

#### **Communications with Directors**

Stockholders may communicate with any and all members of the Board of Directors by transmitting correspondence by mail or facsimile addressed to one or more directors by name (or to the Lead Independent Director, for a communication addressed to the entire Board of Directors) at the following address and fax number:

Name of Director(s)

c/o Corporate Secretary

Microtune, Inc.

2201 10th Street

Plano, Texas 75074

Fax: (972) 673-1815

Communications from our stockholders to one or more directors will be collected and organized by our Corporate Secretary under procedures approved by our independent directors. The Corporate Secretary will forward all communications to the Board or to the identified director(s) as soon as practicable, although communications that are abusive, in bad taste or that present safety or security concerns may be handled differently. If multiple communications are received on a similar topic, the Corporate Secretary may, in his or her discretion, forward only representative correspondence.

The Lead Independent Director will determine whether any communication addressed to the entire Board should be properly addressed by the entire Board or a committee thereof. If a communication is sent to the Board or a committee, the Board of Directors or the chairperson of that committee, as the case may be, will determine whether a response to the communication is warranted. If a response to the communication is warranted, the content and method of the response will be coordinated with our General Counsel.

#### PROPOSAL NO. 1

## **ELECTION OF DIRECTORS**

The following table lists all of Microtune s current directors and nominees for director. The information set forth in the table is as of April 9, 2010. All of Microtune s current directors, with the exception of Messrs. Ciciora, Schueppert and Tai, will stand for re-election at the 2010 Annual Meeting. In addition, Messrs. Peck, Rast and Rau, none of whom previously served as a director of Microtune, are nominees standing for election at the 2010 Annual Meeting. Each nominee, if elected, will serve until the 2011 Annual Meeting of Stockholders or until a qualified successor is elected, unless the nominee resigns or is removed from the Board before then. In the event any such nominee, who has expressed an intention to serve if elected, fails to stand for election, the persons named in the proxy presently intend to vote for a substitute nominee designated by the Board of Directors.

Name	Age	Principal Occupation	Director Since	Audit Committee	Compensation Committee	Nominating and Corporate Governance Committee	Technical Advisory Committee
Walter S. Ciciora <sup>(4)</sup>	67	Independent Consultant	1996	Committee	Committee	X	$\mathbf{C}^{(1)}$
James H. Clardy	75	Retired Venture Partner,	1,,,0				
		Austin Ventures <sup>(3)</sup>	1996		C	X	
Steven Craddock	61	Independent Consultant	2002	X			$C^{(1)}$
James A. Fontaine	52	President and Chief Executive Officer, Microtune, Inc.	2003				
Anthony J. LeVecchio	63	President and Owner,					
		The James Group	2003	C			
Bernard T. Marren	74	President and Chief Executive Officer, OPTi, Inc.	2005		X		
Drew Peck <sup>(5)</sup>	53	Co-Founder and Partner, Steel Horse Advisors					
Robert Rast <sup>(5)</sup>	68	Principal, Tsar Digital					
Raghavendra (Raghu) Rau <sup>(5)</sup>	60	Member of Marketing Advisory Board, Cleversafe, Inc., and Strategic Advisory Board, IOCOM Integrated Communications					
Michael T. Schueppert <sup>(4)</sup>	44	Chief Executive Officer,					
		FluteRadio, LLC	2006		X		
William P. Tai <sup>(4)</sup>	47	General Partner,					
		Charles River Ventures	1998				X
A. Travis White <sup>(2)</sup>	65	Independent Consultant	2004	X		C	

- X Member of the committee.
- C Chairperson of the committee.
- (1) Co-Chairperson of the Technical Advisory Committee.
- (2) Lead Independent Director.
- (3) Retired from Austin Ventures on December 31, 2008.
- (4) Will not stand for re-election to the Board of Directors at the 2010 Annual Meeting.
- (5) Nominee standing for election to the Board of Directors at the 2010 Annual Meeting.

The following is a brief description of the business experience and educational background of each current director and each of the nominees for director, including the capacities in which each has served during the past five years:

Walter S. Ciciora became a director of Microtune in November 1996. Mr. Ciciora has been an independent consultant for companies in the cable, television, consumer electronics and telecommunications industries since October 1993. Mr. Ciciora holds a B.S., M.S. and Ph.D. in electrical engineering from the Illinois Institute of

Technology and a M.B.A. from the University of Chicago. We believe that Mr. Ciciora s financial and business expertise, including a diversified background in the cable television industry gave him the qualifications and skills to serve as a director. Mr. Ciciora is not standing for re-election at the 2010 Annual Meeting.

James H. Clardy became a director of Microtune in August 1996 and served as a member of the Office of the President from June 2003 to August 2003. Mr. Clardy was a venture partner of Austin Ventures, a venture capital firm, from January 1998 to December 2008. Mr. Clardy assumed the role of interim Chief Executive Officer of D2 Audio in August 2004 and left in 2005. Additionally, he served as President and Chief Executive Officer of nanoCoolers from July 2005 until November 2007. Mr. Clardy holds a B.S. in electrical engineering from the University of Tennessee. We believe that Mr. Clardy s financial and business expertise, including a diversified background in the semiconductor industry give him the qualifications and skills to serve as a director.

Steven Craddock became a director of Microtune in April 2002. Mr. Craddock has been an independent consultant for companies in the cable television and telecommunications industries since retiring from Comcast Corporation in July 2008. Previously, he was the Senior Vice President Technology for Comcast Corp. from June 1994, where he was responsible for the evaluation and development of new technologies for voice, data and video and their integration into the Comcast infrastructure. He also served as Comcast s representative to CableLabs, a research and development consortium sponsoring industry initiatives such as the DOCSIS, PacketCable and CableHome standards. He currently serves on a number of technical advisory boards for high technology companies and also serves as a director on the boards of several privately-held companies. Mr. Craddock is a licensed professional engineer (PE) and holds a B.S. in civil engineering and electrical engineering from the Virginia Military Institute. We believe that Mr. Craddock s financial and business expertise, including a diversified background in the cable television and telecommunications industries give him the qualifications and skills to serve as a director.

James A. Fontaine became a director, Chief Executive Officer and President of Microtune in August 2003. Mr. Fontaine was retired from May 2002 until his return to Microtune in August 2003. Mr. Fontaine previously served as Microtune s Chief Strategy Officer from October 2001 until May 2002, Office of the President from August 2001 until September 2001, President from February 1999 until August 2001, and Executive Vice President of Sales and Marketing from August 1998 until February 1999. Mr. Fontaine holds a B.S. in electrical engineering from Marquette University. We believe that Mr. Fontaine s financial and business expertise, including a diversified background in the semiconductor industry give him the qualifications and skills to serve as a director.

Anthony J. LeVecchio became a director of Microtune in August 2003. Mr. LeVecchio has been the President and Owner of The James Group, Inc., a general business consulting firm, since 1988. He currently serves as a director and as chairperson of the audit committees of DG Fast Channel Inc. (NASDAQ: DGIT), Ascendant Solutions (OTC: ASDS.PK) and ViewPoint Financial Group (NASDAQ: VPFG) and serves as a member of the board of directors of several privately-held companies. In March 2010, Mr. LeVecchio participated in the West Coast Peer Exchange Conference for audit committee chairmen sponsored by Corporate Board Member Magazine. In addition, he is a Director in Residence and Advisor to the University of Texas at Dallas s Institute for Corporate Governance. He also is a guest lecturer in the UTD MBA program. Mr. LeVecchio holds a Bachelor of Economics and a M.B.A. in Finance from Rollins College. We believe that Mr. LeVecchio s financial and business expertise, including a diversified background in finance and accounting give him the qualifications and skills to serve as a director.

Bernard T. Marren became a director of Microtune in August 2005. Mr. Marren is currently the President and CEO of OPTi, Inc. (OTC BB: OPTI.OB), a company that licenses intellectual property for logic chips. He currently serves as a director of UniPixel Displays, Inc. (OTC BB: UNXL.OB), a company that develops new display panel technologies. From April 2007 until its acquisition in May 2009, Mr. Marren served on the board of InFocus Corporation (NASDAQ: INFS), a company that manufactured digital projectors. Mr. Marren was the founder and first president of the Semiconductor Industry Association (SIA). Mr. Marren holds a B.S. in electrical engineering from Illinois Institute of Technology (IIT). We believe that Mr. Marren s financial and business expertise, including a diversified background in management and knowledge of intellectual property matters related to the semiconductor industry, give him the qualifications and skills to serve as a director.

Drew Peck is a nominee standing for election to the Board of Directors at the 2010 Annual Meeting. Mr. Peck has not previously served as a director of Microtune. Mr. Peck is a semiconductor analyst and consultant. He is a partner and co-founder of Steel Horse Advisors, founded in 2007, a firm that provides strategic consulting and advises on mergers/acquisitions and capital raising. Mr. Peck is a 25-year veteran of Wall Street with more than 20 years as an equity research analyst. Previously, Mr. Peck was a partner with Crimson Investment, a venture capital and private equity firm, where he focused on Asian technology investments. Prior to Crimson, Mr. Peck was with Cowen & Company, leading its Electronics Research Team. At Cowen, Mr. Peck conducted, managed and directed research coverage of semiconductor companies. Prior to Cowen, he was with DLJ, as a senior semiconductor research analyst and with Gartner Securities, as a semiconductor industry analyst. Before entering into the financial services industry, Mr. Peck was a member of MIT s Lincoln Laboratory s research staff, where he published numerous papers in the fields of quantum electronics and submillimeter physics. Mr. Peck graduated from Brandeis University with a BA in Physics and received a MS in Management and a MSEE from the Massachusetts Institute of Technology. We believe that Mr. Peck s financial and business expertise, including a diversified background as a financial analyst of the semiconductor industry, give him the qualifications and skills to serve as a director.

Robert Rast is a nominee standing for election to the Board of Directors at the 2010 Annual Meeting. Mr. Rast has not previously served as a director of Microtune. Mr. Rast is currently a principal at Tsar Digital. He recently served as Chairman of the Board of the Advanced Television Systems Committee, where he both revitalized the ATSC standard and led the development of the ATSC strategic plan (2006). As senior executive, Mr. Rast led the DTV technical and business initiatives at multiple companies, including Micronas, LINX Electronics, DemoGraFX, Dolby Laboratories, and General Instruments. He has served on a number of industry and technical committees, including the Digital Cinema Advisory Committee, the EIA and NCTA. Mr. Rast has a BSEE from the University of Maryland with additional study for a MSEE at the University of Pennsylvania. We believe that Mr. Rast s financial and business expertise, including a diversified background in the television and telecommunications industries, give him the qualifications and skills to serve as a director.

Raghavendra (Raghu) Rau is a nominee standing for election to the Board of Directors at the 2010 Annual Meeting. Mr. Rau has not previously served as a director of Microtune. Mr. Rau is a strategic advisor specializing in global marketing and business strategy and venture capital and market development for high-technology, early revenue companies. Mr. Rau currently serves on the Marketing Advisory Board of Cleversafe, Inc., a provider of dispersed data storage technologies and on the Strategic Advisory Board of IOCOM Integrated Communications, a provider of software and related services to companies, research labs, and government institutions. Mr. Rau served as Senior Vice President of the Mobile TV Solutions Business of Motorola, Inc. (Motorola), a provider of technologies, products and services in the communications industry, from May 2007 until January 2008, and as Senior Vice President of Strategy and Business Development, Networks & Enterprise of Motorola from March 2006 until May 2007. Mr. Rau served as Corporate Vice President of Global Marketing and Strategy for Motorola from 2005 until 2006 and as Corporate Vice President, Marketing and Professional Services, from 2001 until 2005. From October 1992 until 2001, Mr. Rau served in various positions within Motorola, including as Vice President of Strategic Business Planning and Vice President of Sales and Operations and held positions in Asia and Europe. Mr. Rau is a former Chairman of the QuEST Forum, a collaboration of service providers and suppliers dedicated to telecom supply chain quality and performance, and was a Director of the Center for Telecom management at the University of Southern California. Mr. Rau also served on the Motorola Partnership Board of France Telecom. Mr. Rau holds a Bachelor degree in Engineering from the University of Mysore, India and an MBA from the Indian Institute of Management in Ahmedabad. We believe that Mr. Rau s financial and business expertise, including a diversified background in global marketing and business strategy and venture capital and market development for high-technology companies, give him the qualifications and skills to serve as a director.

Michael T. Schueppert became a director of Microtune in August 2006. Mr. Schueppert is the founder and CEO of FluteRadio, LLC. From January 2005 to December 2006, Mr. Schueppert was President of Modeo<sup>TM</sup> LLC and from 1997 to 2006, he served as Senior Vice President, Business Development, of Crown Castle International Corp. From 1994 to 1997, Mr. Schueppert served in a variety of international sales and marketing roles for Cable & Wireless plc. Mr. Schueppert is a graduate of Cambridge University, UK, and holds a M.B.A.

with distinction from INSEAD, France. We believe that Mr. Schueppert s financial and business expertise, including a diversified background in the telecommunications industry gave him the qualifications and skills to serve as a director. Mr. Schueppert is not standing for re-election at the 2010 Annual Meeting.

William P. Tai became a director of Microtune in June 1998. Mr. Tai has been a general partner of Charles River Ventures since June 2002. He also has been a general partner/managing director of Institutional Venture Partners, a venture capital firm, since July 1997. From 1995 to 2008, Mr. Tai served on the board of Transmeta Corp. (NASDAQ: TMTA), a provider of microprocessors. Mr. Tai also serves on the boards of directors of several privately-held companies. Mr. Tai holds a B.S. in electrical engineering from the University of Illinois and a M.B.A. from Harvard Business School. We believe that Mr. Tai s financial and business expertise, including a diversified background in venture capital and market development for high-technology companies gave him the qualifications and skills to serve as a director. Mr. Tai is not standing for re-election at the 2010 Annual Meeting.

A. Travis White became a director of Microtune in January 2004. Mr. White has been semi-retired since January 2000. Prior to his retirement, Mr. White was President and Chief Executive Officer of Centillium Communications, Inc. from April 1998 through January 2000. Prior to April 1998, he was President of Sony Semiconductor Company of America and also Senior Vice President of Sony Corporation, Japan. He currently serves on the board of one other public company, Entorian Technologies, Inc. (NASDAQ: ENTN) (formerly Staktek Holdings, Inc.), where he serves as chairperson of the audit committee, and as a director on two non-profit organizations and two privately-held companies. In addition, Mr. White is involved in a number of chief executive officer coaching assignments. Mr. White holds a B.S. in biological sciences and chemistry from the University of Texas, El Paso. We believe that Mr. White s financial and business expertise, including a diversified background in management and the semiconductor industry give him the qualifications and skills to serve as a director.

### Agreement with Certain Entities and Individuals Associated with Ramius LLC

On March 23, 2010, Microtune entered into an agreement (the Ramius Agreement ) with certain entities and individuals associated with Ramius LLC (collectively, Ramius ), as further described in the Form 8-K filed by Microtune with the Securities and Exchange Commission on March 24, 2010.

Pursuant to the Ramius Agreement, Microtune agreed, among other things, to nominate three new directors, who include Robert Rast, Raghavendra (Raghu) Rau and Drew Peck, for election to the Board of Directors at the 2010 Annual Meeting and grant Ramius certain rights with regard to director nominations for the 2011 Annual Meeting.

Ramius agreed to withdraw the letter previously submitted to Microtune nominating members to the Board of Directors and to support Microtune s director nominations at the 2010 Annual Meeting and not to submit any nominations for directors at the 2011 Annual Meeting other than in accordance with the terms of the Ramius Agreement. Additionally, Ramius agreed to certain limited standstill restrictions which, subject to certain exceptions relating to the 2011 Annual Meeting, will expire prior to the deadline for nominations of directors for election or submission of proposals to be considered at the 2012 Annual Meeting, but in any event no later than October 31, 2011.

In connection with entering into the Ramius Agreement, Messrs. Ciciora and Tai agreed that they would not stand for re-election at the 2010 Annual Meeting. In addition, Mr. Schueppert will not stand for re-election at the 2010 Annual Meeting. Messrs. Ciciora, Tai and Schueppert will continue to serve as directors until the 2010 Annual Meeting.

### **Required Vote**

A plurality of the votes cast at the Annual Meeting is required to elect each nominee. Accordingly, abstentions and broker non-votes will have no effect on the election of directors. Shares represented by proxies will be voted for the election of the nominees named above unless authority to do so is specifically withheld. If any nominee declines to serve or becomes unavailable for any reason, or if a vacancy occurs before the election, the proxies may be voted for such substitute nominees as the Board of Directors may designate.

If you hold your shares through a broker, bank or other nominee, it is critical that you cast your vote if you want it to count in the election of directors. In the past, if you held your shares through a broker, bank or other nominee and you did not indicate how you wanted your shares voted in the election of directors, your bank, broker or other nominee was allowed to vote those shares on your behalf in the election of directors as they felt appropriate.

Recent changes in the regulations were made to take away the ability of your bank, broker or other nominee to vote your uninstructed shares in the election of directors on a discretionary basis. Thus, if you hold your shares through a bank, broker or other nominee and you do not instruct your bank, broker or other nominee how to vote in the election of directors, no votes will be cast on your behalf. If you are a stockholder of record and you do not cast your vote, no votes will be cast on your behalf on any of the items of business at the Annual Meeting.

#### **Recommendation of the Board of Directors**

The Board of Directors unanimously recommends a vote FOR the election of each of the nine nominees listed above. Unless otherwise instructed, the proxy holders named in each proxy will vote the shares represented thereby FOR the election of the nine nominees named in Proposal No. 1.

## PROPOSAL No. 2

## APPROVAL OF THE MICROTUNE, INC. 2010 STOCK PLAN

The Board of Directors has adopted, and recommends to our stockholders for approval, the Microtune, Inc. 2010 Stock Plan (the 2010 Stock Plan ).

On April 9, 2010, our Board adopted, subject to stockholder approval, the 2010 Stock Plan. The 2010 Stock Plan is intended to replace our Amended and Restated 2000 Stock Plan (the Prior Stock Plan ) and to serve the same purpose of enabling us to deliver a portion of our employee, director and consultant compensation in the form of grants of stock and cash-based incentive awards. The Board believes that an effective and stockholder-friendly compensation program should include long-term incentive compensation, the ultimate value of which is tied to our financial performance. Approval of the 2010 Stock Plan will allow us to continue to attract, motivate and retain qualified employees, directors and consultants and provide them with the incentive to maximize long-term stockholder returns and achieve long-term objectives that will inure to the benefit of our stockholders.

We are proposing the adoption of the 2010 Stock Plan to (i) add to the number of shares currently available for equity grants to our employees, directors and consultants, (ii) provide greater flexibility in the terms and conditions of awards and (iii) conform our stock incentive plan to current best practices and changes in federal tax law. As of March 31, 2010, there was an aggregate total of 2,593,477 shares remaining under the Prior Stock Plan. Since such time, none of these shares have been awarded through our annual equity compensation process, leaving 2,593,477 shares available for grants under the Prior Stock Plan. We intend, however, to make our regular annual equity awards effective May 15, 2010 under the Prior Stock Plan. The shares remaining under the Prior Stock Plan will not be made available for grants under the 2010 Stock Plan and are not included in the aggregate 3,000,000 shares that would be available under the 2010 Stock Plan. If the 2010 Stock Plan is approved by our stockholders, the Board will take such steps as are necessary to terminate the Prior Stock Plan as soon as practicable following such approval.

The 2010 Stock Plan also is intended to enable us to deliver a portion of our employee and director compensation in the form of grants of stock and cash-based incentive awards, which awards, in the discretion of the Compensation Committee, may be designed to satisfy the performance-based compensation exception to the \$1 million limit on deductible compensation under Section 162(m) of the Code.

The primary features of the 2010 Stock Plan are summarized below. The summary is qualified by, and subject to, the provisions of the 2010 Stock Plan, a copy of which is attached as **APPENDIX A**, and should be referred to for a complete statement of the terms of the 2010 Stock Plan. The meaning of capitalized terms not defined in this summary can be found in the Definitions section of the 2010 Stock Plan.

Any stockholder of Microtune who wishes to obtain a copy of the original plan document may do so upon written request to us at: Microtune, Inc., 2201 10th Street, Plano, Texas 75074.

## Summary of the 2010 Stock Plan

Shares Available Under the 2010 Stock Plan. Upon approval of the 2010 Stock Plan by our stockholders, there will be an aggregate of 3,000,000 shares of our common stock, par value \$0.001 per share, available for issuance pursuant to awards under the 2010 Stock Plan. Each share issued to settle awards will reduce the number of shares available for issuance on a 1-for-1 basis. The aggregate number of shares that may be issued to settle awards that are intended to satisfy the performance-based compensation exception of Section 162(m) of the Code will not exceed certain amounts as described below under Section 162(m) Conditions. Upon approval of the 2010 Stock Plan by our stockholders, the shares will be subsequently registered with the Securities and Exchange Commission on a Form S-8.

Administration of the 2010 Stock Plan. In general, the 2010 Stock Plan is administered by the Compensation Committee, which is composed entirely of independent directors. The administrative responsibilities of the Compensation Committee include selecting the individuals eligible to participate in the 2010 Stock Plan, the types of awards that may be granted, the time(s) at which awards may be granted and the number of shares to be covered by each award granted. The Compensation Committee also has the authority to interpret and administer the 2010 Stock Plan, to determine the terms and conditions of awards and to make all other determinations relating to the 2010 Stock Plan that it deems necessary or desirable for the administration of the plan. In addition, the Compensation Committee may appoint a plan administrator who will have the responsibility to administer the 2010 Stock Plan on a daily basis.

Term of Plan. The 2010 Stock Plan will terminate on May 20, 2020, the tenth anniversary of the date it was approved by our stockholders (if such approval occurs) except with respect to awards then outstanding.

Eligibility. Eligible participants under the 2010 Stock Plan include all of our employees, directors and consultants as selected by the Compensation Committee. Approximately 222 persons currently participate in our Prior Stock Plan and are considered to be eligible to participate in the 2010 Stock Plan.

Awards. Awards include options, stock appreciation rights, restricted stock, restricted stock units, performance awards and other share-based awards and any other right, interest or option relating to shares or cash granted pursuant to the 2010 Stock Plan. Each award granted under the 2010 Stock Plan will be evidenced by a written or electronic award agreement in a form, and containing such terms and conditions, as the Compensation Committee determines. The Compensation Committee may grant awards to any eligible participant at any time. The Compensation Committee may delegate to one or more executive officers the authority to grant awards to employees, who are not directors or executive officers, and consultants.

Terms and Conditions of Options. Options may be granted alone or in addition to other awards. Options may be nonqualified stock options or incentive stock options within the meaning of Section 422 of the Code; provided, that incentive stock options may only be granted to employees and are subject to the requirements of Section 422 of the Code as explained in the 2010 Stock Plan.

The exercise price for shares purchasable under an option will not be less than 100% of the fair market value of the shares on the date of grant. Options will vest and be exercisable as determined by the Compensation Committee and may be exercised during a term not to exceed 10 years from the date of grant. The Committee may permit participants to use shares they own in order to pay the exercise price and may provide that the shares to be issued upon an option s exercise will be in the form of restricted stock or other similar securities.

Terms and Conditions of Stock Appreciation Rights. A stock appreciation right may be granted in connection with, or without relationship to, all or part of any award, either at the time of grant or at any time during the term of the award. A stock appreciation right will have (i) a grant price not less than fair market value on the date of grant (subject to the requirements of Section 409A of the Code with respect to stock appreciation rights granted in conjunction with, but subsequent to, an option), and (ii) a term not greater than 10 years. The Compensation Committee may impose such other conditions or restrictions on the terms of the exercise and the grant price of any stock appreciation right as it deems appropriate.

Upon the exercise of a stock appreciation right, the holder will have the right to receive the excess of (i) the fair market value of the shares on the date of exercise (or such amount less than the fair market value as the Compensation Committee determines at any time during a specified period before the date of exercise) over (ii) the grant price of the stock appreciation right on the date of grant, which, except as provided in the 2010 Stock Plan with respect to substitute awards or share adjustments, will not be less than the fair market value of the shares on such date of grant of the stock appreciation right. The Compensation Committee may determine, in its sole discretion, whether payment of a stock appreciation right will be made in cash, in whole shares, or any combination thereof.

Terms and Conditions of Restricted Stock and Restricted Stock Units. Restricted stock and restricted stock units may be granted either alone or in addition to other awards and as a form of payment of performance awards and other earned cash-based incentive compensation. The Compensation Committee will determine the vesting period, which may provide for pro rata vesting over such time (which may be accelerated under certain limited circumstances such as death, disability, retirement or a change in control).

Unless otherwise provided in the applicable award agreement, beginning on the date of the grant of restricted stock, the participant will become a stockholder of the company with respect to all shares subject to the award agreement and will have all of the rights of a stockholder, including the right to vote such shares and the right to receive distributions made with respect to such shares. A participant receiving restricted stock units will not possess voting rights with respect to such award. Generally, shares or any other property (other than cash) distributed as a dividend or otherwise with respect to any unvested awards of restricted stock or restricted stock units will be subject to the same restrictions as the award itself. The shares representing a grant of restricted stock will remain in the physical custody of the company until such restricted stock has vested and any other restrictions are removed or expire.

Terms and Conditions of Performance Awards. Performance awards may be granted for no consideration or for such minimum consideration as may be required by applicable law, either alone or in addition to other awards. The amount subject to the performance award, the performance goals to be achieved during any performance period and the length of the performance period will be determined by the Compensation Committee. Except as provided otherwise in an award agreement or in the event of a change in control, performance awards will be distributed only after the end of the relevant performance period. Performance awards may be paid in cash, shares, or any combination thereof, as determined in the sole discretion of the Compensation Committee. Performance awards may also be paid in lump sum or in installments following the close of the performance period, or, in accordance with procedures established by the Compensation Committee, on a deferred basis subject to Section 409A of the Code.

Terms and Conditions of Other Share-Based Awards. Other share-based awards may be granted to participants either alone or in addition to other awards. Other share-based awards are also available as a form of payment of other awards and other earned cash-based compensation. The same vesting conditions noted for

restricted stock and restricted stock units above generally apply with respect to other share-based awards. Other share-based awards may be paid in cash, shares, or any combination thereof, as determined by the Compensation Committee in its sole discretion. Payments with respect to other share-based awards may be made in a lump sum, in installments or on a deferred basis subject to Section 409A of the Code.

Effect of Termination of Employment or Services. The Compensation Committee will determine and set forth in each award agreement whether any awards will, as applicable, continue to be exercisable and the terms of such exercise, or cease to be subject to restrictions and the timing of when such restrictions will lapse, on and after the date that a participant ceases to be employed by or to provide services to us (including as a director or consultant), whether by reason of death, disability, voluntary or involuntary termination of employment or services, or otherwise. The date of termination of a participant semployment or services will be determined by the Compensation Committee, which determination will be final.

Section 162(m) Conditions. If the Compensation Committee determines that at the time an award is granted to a participant who is, or is likely to be a covered employee (as of the end of the tax year in which the company would claim a tax deduction in connection with such award), then the Compensation Committee may provide that the provisions of the 2010 Stock Plan relating to Section 162(m) of the Code will be applicable to such award (such award is referred to herein as a 162(m) Award). In such case, (i) the vesting of an option or stock appreciation right may be subject to objective performance criteria (as described below) established by the Compensation Committee within the time required by Section 162(m), (ii) the vesting of an award (other than an option or stock appreciation right) and the distribution of cash, shares or other property pursuant thereto, as applicable, will be subject to the achievement of one or more objective performance criteria (as described below) established by the Compensation Committee within the time required by Section 162(m), and (iii) certain limitations with respect to awards will apply. Specifically, during any consecutive five-year period, no participant may with respect to 162(m) Awards (i) be granted options or stock appreciation rights with respect to more than an average of 700,000 shares per year, or (ii) earn more than an average of (A) 700,000 shares per year under other awards or (B) \$5,000,000 per year with respect to performance awards. If a 162(m) Award is cancelled, the cancelled award will continue to be counted toward the foregoing limitations as applicable.

The performance criteria for 162(m) Awards are limited to the following performance measures, taken alone or in conjunction with each other, each of which may be adjusted by the Compensation Committee to exclude the before-tax or after-tax effects of any significant acquisitions or dispositions not included in the calculations made in connection with setting the performance criteria for the relevant award:

- (i) Basic or diluted earnings per share, which may be calculated (A) as income calculated in accordance with (iv) below, divided by (1) the weighted average number of shares, in the case of basic earnings per share, and (2) the weighted average number of shares and shares equivalents of common stock, in the case of diluted earnings per share, or (B) using such other method as may be specified by the Compensation Committee:
- (ii) Cash flow, which may be calculated or measured in any manner specified by the Compensation Committee;
- (iii) Economic value added, which is after-tax operating profit less the annual total cost of capital;
- (iv) Income, which may include, without limitation, net income, operating income, volume measures and expense control measures, and which may be calculated or measured (A) before or after income taxes, including or excluding interest, depreciation and amortization, minority interests, extraordinary items and other material non-recurring items, discontinued operations, the cumulative effect of changes in accounting policies and the effects of any tax law changes; or (B) using such other method as may be specified by the Compensation Committee;

(v) Business performance or return measures (including, but not limited to, market share, debt reduction, return on assets, capital, equity, or sales), which may be calculated or measured in any manner specified by the Compensation Committee;

(vi) The price of the company s common stock (including, but not limited to, growth measures and total stockholder return), which may be calculated or measured in any manner specified by the Compensation Committee; or

(vii) Any of the above performance criteria, determined on an absolute or relative basis or as compared to the performance of a published or special index deemed applicable by the Compensation Committee including, but not limited to, the Russell 2000 or the Russell 3000 Index, as applicable, or a group of companies deemed by the Compensation Committee to be comparable to the company.

All determinations regarding the achievement of performance goals are made by the Compensation Committee. The Compensation Committee may adjust downwards, but not upwards, the amount payable pursuant to a 162(m) Award; provided, however, that no such adjustment shall be made if it would cause the 2010 Stock Plan or an award to fail to comply with or be exempt from the requirements of Section 409A of the Code. The Compensation Committee may not waive the achievement of the applicable performance goals, except in the case of the death or disability of the participant or as otherwise determined by the Compensation Committee in special circumstances, subject to the requirements of Section 162(m) of the Code.

Adjustment Provisions. Awards may be adjusted in the event of unusual events such as distributions in connection with a merger, reorganization or stock split. In addition, in the event of a change in control, the treatment of outstanding awards granted to employees, directors or consultants generally depends on whether: (i) the successor company assumes or substitutes the applicable award, in which case such awards will continue pursuant to their terms; and/or (ii) the participant incurs a qualifying termination (subject to the terms and conditions in the participant s award agreements), in which case such outstanding awards will become immediately vested and exercisable or transferable, depending on the type of award. If the successor company does not assume or substitute the applicable award, such award will become fully vested and exercisable, depending on the type of award, immediately prior to the change in control irrespective of whether the participant incurs a qualifying termination.

In addition, the Compensation Committee may make other determinations, as described in the 2010 Stock Plan, regarding the effect of a change in control on awards.

Transferability of Awards. Generally, awards granted under the 2010 Stock Plan are not transferable other than (i) by will or the laws of descent and distribution, or (ii) pursuant to a qualified domestic relations order. The Compensation Committee in its discretion may permit other transfers of awards pursuant to the terms of the 2010 Stock Plan.

Amendment and Termination of the 2010 Stock Plan. The Compensation Committee and the Board have the authority to amend the 2010 Stock Plan at any time, and the company may, by action of the Board, terminate the 2010 Stock Plan at any time and for any reason. The 2010 Stock Plan, however, may not be amended in any respect that materially increases the benefits available under the 2010 Stock Plan without the approval of our stockholders, including, (i) increasing the number of shares that may be the subject of awards under the 2010 Stock Plan (ii) expanding the types of awards available under the 2010 Stock Plan, (iii) materially expanding the class of persons eligible to participate in the 2010 Stock Plan, (iv) amending any provision of the 2010 Stock Plan regarding changes in the exercise price of options and stock appreciation rights, (v) increasing the maximum permissible term of any option or the maximum permissible term of a stock appreciation right, or (vi) increasing the limitations set forth above under Section 162(m) Conditions .

### **Certain Federal Income Tax Consequences**

The following is a general description of the U.S. federal income tax consequences to participants and the company relating to awards. This discussion does not purport to cover all tax consequences relating to the awards, and assumes, with respect to deductibility of compensation by the company, that to the extent applicable, the requirements of Section 162 and Section 162(m) of the Code, as applicable, have been satisfied. Also, our ability to obtain a deduction for future payments under the 2010 Stock Plan could be limited by the golden parachute rules of Section 280G of the Code, which prevent the deductibility of certain excess parachute payments made in connection with a change of control of the employer corporation. The tax treatment may also vary depending on the participant s particular situation and may, therefore, be subject to special rules not discussed below.

A participant who receives options or stock appreciation rights generally will not recognize any income, nor will the company be entitled to any tax deduction, at the time of the grant. At the time that a nonqualified stock option or stock appreciation right is exercised, the participant will recognize ordinary income in an amount equal to the excess of (i) the fair market value of the shares purchased (or subject to a stock appreciation right) over (ii) the exercise price of the option for such shares (or the grant price of a stock appreciation right). We generally will be entitled to a tax deduction in an amount equal to the amount includible in the income of the participant in the taxable year in which the participant is required to recognize the income. A participant who disposes of shares received upon the exercise of a nonqualified stock option will recognize capital gain or loss in an amount equal to the difference between (i) the amount realized on the disposition of the shares, and (ii) the fair market value of the shares on the date on which the option was exercised. The capital gain or loss will be considered long-term if the shares received upon exercise of the nonqualified stock option are held for more than one year after the option was exercised. We are not entitled to any deduction for federal income tax purposes upon a participant s disposition of stock received upon the exercise of an option (other than, in the circumstances described below, an incentive stock option).

A participant will recognize no income for federal income tax purposes upon the grant or the exercise of an incentive stock option, provided that the exercise occurs during employment or within three months after termination, other than in the case of death or disability. If the shares acquired upon the exercise are held for a minimum of both (i) two years from the date of grant and (ii) one year from the date of exercise, then any gain or loss recognized by the participant on the sale of such shares will be treated as a long-term capital gain or loss, and the company will not be entitled to any deduction for federal income tax purposes. If the shares acquired are not held for these minimum periods, then the participant will be required to recognize ordinary income in the year of the disposition to the extent that the fair market value of the shares on the date of exercise or the sale price, whichever is less, exceeds the exercise price for the shares. We generally will be entitled to a deduction for federal income tax purposes equal to the amount the participant is required to recognize as ordinary income.

A participant who receives awards payable in cash or restricted stock, restricted stock units or performance awards, will not recognize income for federal income tax purposes until the awards are settled. At that time, the participant will recognize ordinary income on the amount of cash received, if any, or for awards delivered in shares, the excess of (i) the fair market value of the shares on the settlement date over (ii) the amount, if any, paid for the shares. We will be entitled to take a tax deduction in an amount equal to the ordinary income recognized by the participant.

The 2010 Stock Plan allows the Compensation Committee to permit the transfer of awards in limited circumstances. See Transferability of Awards above. For income and gift tax purposes, certain transfers of nonqualified stock options and stock appreciation rights generally should be treated as completed gifts, subject to gift taxation.

An employee participant will be subject to withholding for federal and, if applicable, state and local, income taxes at the time the participant recognizes income under the rules described above with respect to shares or cash received. As such, we will have the right to make all payments or distributions to a participant net of any taxes required to be paid at such time. We will have the right to withhold from wages or other amounts otherwise payable such withholding taxes as may be required by law, to otherwise require the participant to pay such withholding taxes or to take such other action as may be necessary to satisfy such withholding obligations. Non-employee

directors and consultants are not subject to withholding by us and must make their own arrangements for satisfying any tax obligations they may have in connection with the grant or exercise of an award under the 2010 Stock Plan.

Section 409A of the Code imposes an additional 20% tax and interest on an individual receiving nonqualified deferred compensation, as defined in Section 409A, under an arrangement that fails to satisfy certain requirements. Awards made pursuant to the 2010 Stock Plan are designed to comply with the requirements of Section 409A to the extent such awards are not otherwise exempt from coverage. However, if the 2010 Stock Plan fails to comply with Section 409A in operation, a participant could be subject to the additional taxes and interest.

#### **New Plan Benefits**

Future benefits that may be awarded under the 2010 Stock Plan will be subject to the discretion of the Compensation Committee and, therefore, are not currently determinable. The following table sets forth (i) the aggregate number of shares subject to restricted stock units awarded under the Prior Stock Plan during the last fiscal year and (ii) the dollar value of such shares based on a closing price of our common stock on the applicable date of grant.

Name of Individual or Group	Number of Restricted Stock Units (#)	Dollar Value of Restricted Stock Units (\$)
James A. Fontaine, Chief Executive Officer and President	80,000	172,000
Robert S. Kirk, Vice President of Worldwide Sales	28,000	60,200
Barry F. Koch, Executive Vice President and Managing Director, Microtune GmbH & Co. KG*	24,000	51,600
Justin M. Chapman, Chief Financial Officer and Former Interim Chief Financial Officer**	5,900	12,685
Phillip D. Peterson, General Counsel	16,000	34,400
All Executive Officers as a Group	153,900	330,885
Non-Executive Directors as a Group		
Non-Executive Officer Employees as a Group	562,110	1,172,387

- \* Mr. Koch was promoted to the position of Executive Vice President effective June 10, 2009. Mr. Koch continues to serve as Managing Director of Microtune GmbH & Co. KG.
- \*\* Mr. Chapman was appointed to the position of Interim Chief Financial Officer effective December 4, 2009. Mr. Chapman was appointed to the position of Chief Financial Officer effective February 11, 2010.

## **Required Vote**

The affirmative vote of the holders of a majority of the shares of common stock present in person or represented by proxy and entitled to vote is required to approve the 2010 Stock Plan. An abstention will be counted as a vote against approval since it is one less vote for approval. Broker non-votes will not affect the outcome since they are not considered shares present for voting purposes.

### Recommendation of the Board of Directors

The Board of Directors unanimously recommends a vote FOR the approval of the Microtune, Inc. 2010 Stock Plan. Unless otherwise instructed, the proxy holders named in each proxy will vote the shares represented thereby FOR the ratification and approval of the Microtune, Inc. 2010 Stock Plan as provided in Proposal No. 2.

## PROPOSAL No. 3

## APPROVAL OF THE MICROTUNE, INC. 2010 DIRECTOR STOCK PLAN

The Board of Directors has adopted, and recommends to our stockholders for approval, the Microtune, Inc. 2010 Director Stock Plan (the 2010 Director Plan ).

On April 9, 2010, our Board adopted, subject to stockholder approval, the 2010 Director Plan. The 2010 Director Plan is intended to replace our Amended and Restated 2000 Director Stock Plan (the Prior Director Plan ) and to serve the same purpose of enabling us to attract and retain the best available personnel for service as non-employee directors, to provide additional incentive to non-employee directors to serve as directors and to encourage their continued service on our Board. The 2010 Director Plan is intended to enable us to deliver a portion of non-employee director compensation in the form of grants of stock pursuant to restricted stock unit awards. The Board believes that an effective and stockholder-friendly compensation program should include long-term incentive compensation, the ultimate value of which is tied to our financial performance. Approval of the 2010 Director Plan will allow us to continue to attract, motivate and retain non-employee directors and provide them with the incentive to maximize long-term stockholder returns and achieve long-term objectives that will inure to the benefit of our stockholders.

We are proposing the adoption of the 2010 Director Plan to (i) add to the number of shares currently available for equity grants to our non-employee directors, and (ii) conform our non-employee director plan to current best practices and changes in federal tax law. As of March 31, 2010, there was an aggregate total of 226,800 shares remaining under the Prior Director Plan. Since such time, none of these shares have been awarded under the Prior Director Plan, leaving 226,800 shares available for grant under the Prior Director Plan. We intend, however, to make our regular annual director equity awards effective May 20, 2010 under the Prior Director Plan. The shares remaining under our Prior Director Plan will not be made available for grants under the 2010 Director Plan and are not included in the aggregate 250,000 shares that would be available under the 2010 Director Plan is approved by our stockholders, the Board will take such steps as are necessary to terminate the Prior Director Plan as soon as practicable following such approval.

The primary features of the 2010 Director Plan are summarized below. The summary is qualified by, and subject to, the provisions of the 2010 Director Plan, a copy of which is attached as **APPENDIX B**, and should be referred to for a complete statement of the terms of the 2010 Director Plan. The meaning of capitalized terms not defined in this summary can be found in the Definitions section of the 2010 Director Plan.

Any stockholder of Microtune who wishes to obtain a copy of the original plan document may do so upon written request to us at: Microtune, Inc., 2201 10th Street, Plano, Texas 75074.

Shares Available Under the 2010 Director Plan. Upon approval of the 2010 Director Plan by our stockholders, there will be an aggregate of 250,000 shares of our common stock, par value \$0.001 per share, available for issuance pursuant to awards under the 2010 Director Plan. Each share issued to settle awards will reduce the number of shares on a 1-for-1 basis. Upon approval of the 2010 Director Plan by our stockholders, the shares will be subsequently registered with the Securities and Exchange Commission on a Form S-8.

Administration of the 2010 Director Plan. In general, the 2010 Director Plan is administered by the Compensation Committee of the Board. The Compensation Committee has limited administrative powers mainly involving the authority to interpret and administer the 2010 Director Plan and to make all other determinations relating to the 2010 Director Plan that it deems necessary or desirable for the administration of the plan. Grants of awards under the 2010 Director Plan are generally intended to be automatic and nondiscretionary. Accordingly, the Compensation Committee has no authority to select the individuals eligible to participate in the 2010 Director Plan, the types of awards granted, the time(s) at which awards may be granted, the time(s) at which awards become vested, and the number of shares to be covered by each award granted (subject to the adjustment provisions in the 2010 Director Plan).

Term of Plan. The 2010 Director Plan will terminate on May 20, 2020, the tenth anniversary of the date it was approved by our stockholders (if such approval occurs) except with respect to awards then outstanding.

Eligibility. Eligible participants under the 2010 Director Plan include non-employee directors. After our 2010 Annual Meeting of Stockholders, eight persons are considered to be eligible to participate in the 2010 Director Plan, pending stockholder approval.

Awards. Awards that may be granted under the 2010 Director Plan only include restricted stock units, and no other form of award may be granted under the 2010 Director Plan. Each award granted under the 2010 Director Plan will be evidenced by a written or electronic award agreement in a standardized form for all participants, and containing such terms and conditions that are consistent with the provisions of the 2010 Director Plan.

Timing of Awards and Number of Underlying Restricted Stock Units. Effective for periods commencing after May 20, 2010, a non-employee director shall receive an award with respect to 5,250 restricted stock units on the date he or she first becomes a non-employee director. Similarly, after May 20, 2010, each non-employee director will also receive an award with respect to 8,400 restricted stock units on the date of our annual stockholders meeting of each year thereafter; provided, that he or she is a non-employee director on such date and he or she has served as a director for at least the preceding six months.

Vesting and Forfeiture of Awards. Each award becomes vested as to 33 \(^1/3\)% of the restricted stock units subject to such award on each anniversary of its date of grant, provided that the participant continues to serve as a director on such dates. Except as otherwise provided in the 2010 Director Plan, an award agreement or in any other agreement between the participant and the Company, if the termination of the participant s continuous status as a director occurs for any reason (including death or disability) or there is a liquidation or dissolution of the company, all unvested restricted stock units shall be forfeited as of the termination, liquidation or dissolution date (as applicable) and returned to the 2010 Director Plan.

Settlement of Awards. Except as otherwise provided in the 2010 Director Plan or an award agreement, to the extent restricted stock units become vested, the company will settle such vested restricted stock units, and as a result thereof issue and deliver RSU shares to the participant (i.e., one share for each such vested restricted stock unit). Shares of common stock representing the number of RSU shares will be issued to the participant as soon as practicable after settlement of the vested restricted stock units, but in no event later than seventy-five (75) days after such settlement.

Rights of Holders of Restricted Stock Units. A participant receiving restricted stock units will not possess voting rights with respect to such award. Until the issuance of the RSU shares and the stock certificate evidencing such shares, no right to vote or receive dividends or any other rights as a stockholder exist with respect to any restricted stock units, even if the time or event that gives rise to the settlement of such restricted stock units has occurred.

Adjustment Provisions. Awards may be adjusted in the event of unusual events such as distributions in connection with a merger, reorganization or stock split. In addition, in the event of a change in control, the treatment of outstanding awards granted to non-employee directors generally depends on whether: (i) the

successor company assumes or substitutes the applicable award, in which case such awards will continue pursuant to their terms to the extent the participant continues his or her status as a director with the successor company. In the event the participant s status as a director with the successor company ceases for any reason other than voluntary resignation, or the participant is not offered a director-level position with the successor company, then all unvested restricted stock units will become fully vested as of the date of termination or immediately prior to the change in control (as applicable). If the successor company does not assume or substitute the applicable award, such award will become fully vested immediately prior to the change in control irrespective of whether the participant continues his or her status as a director with the successor company. The Compensation Committee is authorized to make certain determinations, as described in the 2010 Director Plan, regarding the effect of a change in control on awards.

Transferability of Awards. Generally, awards granted under the 2010 Director Plan are not transferable other than (i) by will or the laws of descent and distribution, or (ii) pursuant to a qualified domestic relations order. The Compensation Committee in its discretion may permit other transfers of awards pursuant to the terms of the 2010 Director Plan.

Amendment and Termination of the 2010 Director Plan. The Compensation Committee and the Board have the authority to amend the 2010 Director Plan at any time, and the company may, by action of the Board, terminate the 2010 Director Plan at any time and for any reason. The 2010 Director Plan, however, may not be amended in any respect that materially increases the benefits available under the 2010 Director Plan without the approval of our stockholders, including, (i) increasing the number of restricted stock units (subject to the plan s adjustment provisions), (ii) expanding the types of awards available under the 2010 Director Plan or (iii) expanding the class of persons eligible to participate in the 2010 Director Plan.

#### **Certain Federal Income Tax Consequences**

The following is a general description of the U.S. federal income tax consequences to participants and the company relating to awards. This discussion does not purport to cover all tax consequences relating to the awards, and assumes, with respect to deductibility of compensation by the company, that to the extent applicable, the requirements of Section 162 of the Code have been satisfied. Also, our ability to obtain a deduction for future payments under the 2010 Director Plan could be limited by the golden parachute rules of Section 280G of the Code, which prevent the deductibility of certain excess parachute payments made in connection with a change of control of the company. The tax treatment may also vary depending on the participant s particular situation and may, therefore, be subject to special rules not discussed below.

A participant who receives awards comprised of restricted stock units will not recognize income for federal income tax purposes until the awards are settled. At that time, the participant will recognize ordinary income on the amount of cash received, if any, or for awards delivered in shares, the excess of (i) the fair market value of the shares on the settlement date over (ii) the amount, if any, paid for the shares. We will be entitled to take a tax deduction in an amount equal to the ordinary income recognized by the participant.

A participant who becomes an employee will be subject to withholding for federal and, if applicable, state and local, income taxes at the time the participant recognizes income under the rules described above with respect to shares or cash received. As such, we will have the right to make all payments or distributions to an employee participant net of any taxes required to be paid at such time. We will have the right to withhold from wages or other amounts otherwise payable such withholding taxes as may be required by law, to otherwise require the employee participant to pay such withholding taxes or to take such other action as may be necessary to satisfy such withholding obligations. Non-employee directors are not subject to withholding by us and must make their own arrangements for satisfying any tax obligations they may have in connection with the settlement of an award under the 2010 Director Plan.

Section 409A of the Code imposes an additional 20% tax and interest on an individual receiving nonqualified deferred compensation, as defined in Section 409A, under an arrangement that fails to satisfy certain requirements. Awards made pursuant to the 2010 Director Plan are designed to comply with the requirements of Section 409A to the extent such awards are not exempt from coverage. However, if the 2010 Director Plan fails to comply with Section 409A in operation, a participant could be subject to the additional taxes and interest.

## **New Plan Benefits**

The company currently anticipates that it will grant 8,400 RSU awards to each non-employee director on the date of each annual meeting, pursuant to the 2010 Director Plan, however, the value of such benefits is not currently determinable. Only non-employee directors are eligible to participate in the 2010 Director Plan. The following table sets forth (i) the aggregate number of shares issued pursuant to awards of restricted stock units under the Prior Director Plan during the last fiscal year and (ii) the dollar value of such shares based on a closing price of \$1.95 per share on April 24, 2009.

Name of Individual or Group	Number of Restricted Stock Units (#)	Dollar Value of Restricted Stock Units (\$)
All Executive Officers as a Group		
Non-Executive Directors as a Group	67,200	131,040
Non-Executive Officer Employees as a Group		

## **Required Vote**

The affirmative vote of the holders of a majority of the shares of common stock present in person or represented by proxy and entitled to vote is required to approve the 2010 Director Plan. An abstention will be counted as a vote against approval since it is one less vote for approval. Broker non-votes will not affect the outcome since they are not considered shares present for voting purposes.

#### Recommendation of the Board of Directors

The Board of Directors unanimously recommends a vote FOR the approval of the Microtune, Inc. 2010 Director Stock Plan. Unless otherwise instructed, the proxy holders named in each proxy will vote the shares represented thereby FOR the ratification and approval of the Microtune, Inc. 2010 Director Stock Plan as provided in Proposal No. 3.

#### PROPOSAL No. 4

## APPROVAL OF THE MICROTUNE, INC.

## 2010 EMPLOYEE STOCK PURCHASE PLAN

The Board of Directors has adopted, and recommends to our stockholders for approval, the Microtune, Inc. 2010 Employee Stock Purchase Plan (the 2010 Stock Purchase Plan ).

Effective as of April 30, 2010, our Board adopted, subject to stockholder approval, the 2010 Stock Purchase Plan. The 2010 Stock Purchase Plan is intended to replace our Amended and Restated 2000 Employee Stock Purchase Plan (the Prior Stock Purchase Plan ) and will serve the same purpose of enhancing stockholder value and promoting the attainment of our significant business objectives by allowing our employees to purchase the company s common stock at a discount, thereby giving employees an interest in common with that of our stockholders. Approval of the 2010 Stock Purchase Plan will allow us to continue to attract, motivate and retain qualified employees and to encourage them to enhance stockholder value.

We are proposing the adoption of the 2010 Stock Purchase Plan to (i) add to the number of shares currently available for purchase under our employee stock purchase program, and (ii) conform our employee stock purchase program to current best practices and changes in federal tax law. If stockholder approval is not obtained by October 29, 2010, the provisions of the 2010 Stock Purchase Plan will be null and void and any contributions made by participants to purchase shares thereunder will be refunded to such participants as provided in the 2010 Stock Purchase Plan. As of March 31, 2010, there was an aggregate total of 105,140 shares remaining available for purchase under the Prior Stock Purchase Plan. None of the remaining shares under the Prior Stock Purchase Plan will be made available for purchase under the 2010 Stock Purchase Plan and such shares are not included in the aggregate 1,000,000 shares available for purchase under the 2010 Stock Purchase Plan. The Prior Stock Purchase Plan will terminate on August 4, 2010 unless terminated earlier after the approval of the 2010 Stock Purchase Plan. If the 2010 Stock Purchase Plan as soon as practicable following such approval.

The primary features of the 2010 Stock Purchase Plan are summarized below. The summary is qualified by, and subject to, the provisions of the 2010 Stock Purchase Plan, a copy of which is attached as **APPENDIX** C and should be referred to for a complete statement of the terms of the 2010 Stock Purchase Plan. The meaning of capitalized terms not defined in this summary can be found in the Definitions section of the 2010 Stock Purchase Plan.

Any stockholder of Microtune who wishes to obtain a copy of the original plan document may do so upon written request to us at: Microtune, Inc., 2201 10th Street, Plano, Texas 75074.

## Summary of the 2010 Stock Purchase Plan

Number of Shares Available. Upon approval of the 2010 Stock Purchase Plan, there will be an aggregate of 1,000,000 shares of our common stock, par value \$0.001 per share, available for issuance under the 2010 Stock Purchase Plan. Each share purchased under the 2010 Stock Purchase Plan will reduce the number of shares on a 1-for-1 basis. Upon approval of the 2010 Stock Purchase Plan by our stockholders, the shares will be subsequently registered with the Securities and Exchange Commission on a Form S-8.

Administration of the 2010 Stock Purchase Plan. The 2010 Stock Purchase Plan is generally administered by the Compensation Committee of the Board, which is composed solely of independent directors who are not eligible to participate in the 2010 Stock Purchase Plan. However, the Compensation Committee may delegate certain of its administrative functions including the day-to-day administration of the 2010 Stock Purchase Plan to a plan administrator. We will bear the costs of administration of the 2010 Stock Purchase Plan.

Term of Plan. The 2010 Stock Purchase Plan will terminate on April 30, 2020, the tenth anniversary of its effective date pending stockholder approval.

Eligibility. Any employee who customarily works at least 20 hours per week and more than five months per calendar year is eligible to participate in the 2010 Stock Purchase Plan with respect to a particular offering period if he or she works for us as of the commencement of such offering period. Our non-employee directors and consultants are not eligible to participate in the 2010 Stock Purchase Plan. Employees participating in the Prior Stock Purchase Plan as of April 30, 2010 may continue participating in the 2010 Stock Purchase Plan without filing new participation forms with the plan administrator. Approximately 193 persons currently are eligible to participate in our Prior Stock Purchase Plan (and approximately 83 persons are currently participating in the Prior Stock Purchase Plan) and are considered to be eligible to participate in the 2010 Stock Purchase Plan.

Operation of the 2010 Stock Purchase Plan. Each employee eligible to participate in the 2010 Stock Purchase Plan may contribute up to 15% of the employee s covered compensation towards the purchase of shares at a purchase price for each offering period (a period of approximately six months) equal to 85% of the fair market value of the shares at the beginning or the close of the offering period, whichever is lower. Employees may increase or decrease the contribution rate for the next offering period, and except when a participant withdraws from the 2010 Stock Purchase Plan (as discussed below), changes in the contribution rate for the current offering period are not permitted.

A participant s contribution percentage will remain in effect for subsequent offering periods until such time as the participant withdraws from the 2010 Stock Purchase Plan or changes the contribution percentage for a subsequent offering period by providing written instructions to the plan administrator. Subject to applicable law, participants, who choose to withdraw from the 2010 Stock Purchase Plan, must wait until a subsequent offering period before they may resume participating in the plan.

The amount contributed by a participant will be (i) deducted from each paycheck on an after-tax basis and (ii) applied towards the purchase of shares at the close of the offering period. These contributions will not bear interest. If a participant terminates employment for any reason (including death or retirement), the participant s contributions for the offering period will cease, and the accumulated amount of such contributions will be refunded to the participant (or the participant s estate or beneficiary, as applicable).

Restrictions on Purchase. No employee may purchase shares under the 2010 Stock Purchase Plan if (i) the employee owns 5% or more of the total combined voting power or value of our stock or (ii) the purchase would permit the employee to acquire shares with a value in excess of \$25,000 (fair market value) in a single year. Additionally, participants may purchase no more than 5,000 shares during each offering period.

Purchased Shares. As promptly as practicable following the purchase date, shares purchased by a participant under the 2010 Stock Purchase Plan will be provided to the participant in his or her designated brokerage account (or as otherwise designated pursuant to the plan). No fractional shares will be purchased under the 2010 Stock Purchase Plan and amounts that otherwise would have been applied to the purchase of fractional shares will be refunded to the participant.

Amendment or Termination of the 2010 Stock Purchase Plan. The Compensation Committee and the Board at any time may amend the 2010 Stock Purchase Plan, in whole or in part, and the company may, by action of the Board, terminate the 2010 Stock Purchase Plan at any time for any reason.

### **Certain Federal Income Tax Consequences**

The following is a general description of the U.S. federal income tax consequences relating to the 2010 Stock Purchase Plan. These tax consequences may vary depending on the personal circumstances of the individual participants.

The 2010 Stock Purchase Plan is intended to qualify as an employee stock purchase plan under Section 423 of the Code. A participant s payroll contributions are made on an after-tax basis, but the enrollment in the plan for a particular offering period and the purchase of shares on the purchase date are not taxable events for the participant. All tax consequences are deferred until the participant disposes of the shares and such tax consequences will depend on how long a participant has held the shares prior to the disposition.

If a participant disposes of the shares in a qualifying disposition a disposition of the shares that occurs (i) at least two years after the beginning of the offering period (the Commencement Date ) and (ii) at least one year after the date such shares are purchased under the 2010 Stock Purchase Plan (or if the participant dies while owning the shares) the participant will have the following tax consequences. In the year of the disposition, the participant will include in his or her taxable ordinary income the lesser of: (i) the excess of fair market value of the shares at the Commencement Date over the shares purchase price (i.e., the 85% discounted price), or (ii) the excess of the fair market value of the shares at the time of the disposition (or death) over the shares purchase price (i.e., the 85% discounted price). Any further gain on the disposition will generally be taxed as long-term capital gain. If the shares are sold at a price that is less than the shares purchase price (i.e., the 85% discounted price), the difference would be a long-term capital loss for the participant. No deduction is allowed to the company in a qualifying disposition.

When shares are disposed of prior to the expiration of either the two-year or one-year holding period described above, the disposition is a disqualifying disposition. In the year of the disqualifying disposition, the participant will include in his or her taxable ordinary income the amount by which the fair market value of the shares on the purchase date exceeds the shares purchase price (i.e., the 85% discounted price). This excess amount is taxable to the participant even if no gain is realized on the disposition and even if the shares are disposed of for less than the fair market value on the purchase date. The participant will have capital gain if there is any gain above the ordinary income amount, and he or she will have capital loss if the shares are sold for an amount that is less than the fair market value on the purchase date. The capital gain or capital loss will be short-term or long-term capital gain or loss depending on the period of time which the participant held the shares. In the event of a disqualifying disposition, the company will be allowed a deduction for federal income tax purposes equal to the ordinary income realized by the participant disposing of the shares.

Currently, the company is not required to withhold employment or income taxes upon the purchase of shares under the 2010 Stock Purchase Plan. However, the Internal Revenue Service may issue guidance in the future requiring the company to withhold employment and income taxes upon the purchase of shares granted under plans that qualify under Section 423 of the Code.

### **New Plan Benefits**

Future benefits that may be awarded under the 2010 Stock Purchase Plan are not currently determinable, because participation in the 2010 Stock Purchase Plan will be voluntary, and the benefit of participating will depend on the terms of the offerings, including the fair market value of the shares on the purchase date. Non-employee directors are not eligible to participate in the 2010 Stock Purchase Plan. No purchases have been made under the 2010 Stock Purchase Plan since its adoption by the Board. For illustrative purposes, the following table sets forth (i) the number of shares of our common stock that were purchased during the last fiscal year under the Prior Stock Purchase Plan, (ii) the average price per share paid for such shares, and (iii) the fair market value at the date of purchase.

Name of Individual or Group	Number of Shares Purchased (#)	Average Per Share Purchase Price (\$)	Fair Market Value at Date of Purchase (\$)
James A. Fontaine,	r ui chaseu (#)	rui chase riice (\$)	at Date of Furchase (\$)
James A. I Ontaine,			
Chief Executive Officer and President			
Albert H. Taddiken,			
,			
Former Chief Operating Officer*			
Jeffrey A. Kupp,			
Former Chief Financial Officer**			
Robert S. Kirk,			
Vice President of Worldwide Sales			
Barry F. Koch,	4,043	1.59	7,541
Executive Vice President and Managing Director, Microtune			
GmbH & Co. KG***			
Justin M. Chapman,			
Chief Financial Officer and Former Interim Chief Financial			
Officer***			
Phillip D. Peterson,			
a 10 1			
General Counsel	4.042	1.50	7.541
All Executive Officers as a Group	4,043	1.59	7,541
Non-Executive Directors as a Group Non-Executive Officer Employees as a Group	534,847	1.62	1,017,371
Non-Executive Officer Employees as a Group	334,047	1.02	1,017,371

- \* Mr. Taddiken resigned from the position of Chief Operating Officer effective June 10, 2009.
- \*\* Mr. Kupp resigned from the position of Chief Financial Officer effective December 4, 2009.
- \*\*\* Mr. Koch was promoted to the position of Executive Vice President effective June 10, 2009. Mr. Koch continues to serve as Managing Director of Microtune GmbH & Co. KG.
- \*\*\*\* Mr. Chapman was appointed to the position of Interim Chief Financial Officer effective December 4, 2009. Mr. Chapman was appointed to the position of Chief Financial Officer effective February 11, 2010.

## **Required Vote**

The affirmative vote of the holders of a majority of the shares of common stock present in person or represented by proxy and entitled to vote is required to approve the 2010 Stock Purchase Plan. An abstention will be counted as a vote against approval since it is one less vote for approval. Broker non-votes will not affect the outcome since they are not considered shares present for voting purposes.

The Board of Directors unanimously recommends a vote FOR the approval of the Microtune, Inc. 2010 Employee Stock Purchase Plan. Unless otherwise instructed, the proxy holders named in each proxy will vote the shares represented thereby FOR the ratification and approval of the Microtune, Inc. 2010 Employee Stock Purchase Plan as provided in Proposal No. 4.

#### PROPOSAL NO. 5

## RATIFICATION OF APPOINTMENT OF INDEPENDENT AUDITOR

The Board of Directors, upon recommendation of the Audit Committee, has selected KPMG LLP as our independent auditor to perform the audit of our financial statements for the fiscal year ending December 31, 2010, and we are asking stockholders to ratify this selection. Representatives of KPMG LLP are expected to be present at our Annual Meeting. They will have the opportunity to make a statement at our Annual Meeting if they wish to do so, and they will be available to respond to appropriate questions from stockholders.

In connection with the settlement of consolidated stockholder derivative litigation in January 2005, we adopted certain corporate governance provisions, including, among others, our agreement to rotate our independent auditor. As a result, on March 13, 2008, the Audit Committee appointed KPMG LLP to serve as Microtune s independent auditor for the fiscal year ended December 31, 2008. This appointment followed a solicitation and review process conducted by us pursuant to the Audit Committee s determination to solicit competitive proposals for audit services from independent accounting firms. During the fiscal years ended December 31, 2006 and 2007, and prior to its engagement, (i) KPMG LLP was not engaged as our principal accountant to audit our financial statements, or as an independent accountant to audit a significant subsidiary, and (ii) we did not consult with KPMG LLP regarding (a) the application of accounting principles to any completed or proposed transaction, (b) the type of audit opinion that might be rendered on our financial statements for such periods, or (c) any other accounting, auditing or financial reporting matter described in Items 304(a)(2)(i) and (ii) of Regulation S-K.

## **Change in Independent Registered Public Accounting Firm**

Effective March 13, 2008 following the solicitation process described above, we replaced Ernst & Young LLP as our independent auditor. Ernst & Young LLP completed its procedures regarding our financial statements for the fiscal year ended December 31, 2007 and the 2007 Annual Report on Form 10-K (in which such financial statements are included) on February 28, 2008, coincident with the filing of our 2007 Annual Report on Form 10-K.

Ernst & Young LLP s reports on our financial statements for the fiscal years ended December 31, 2006 and 2007 did not contain an adverse opinion or a disclaimer of opinion, and were not qualified or modified as to uncertainty, audit scope or accounting principle, except for an explanatory paragraph relating to a change in the method of accounting in 2007 for uncertain tax positions. During the fiscal years ended December 31, 2006 and 2007, and through March 13, 2008, (i) there were no disagreements with Ernst & Young LLP on any matter of accounting principles or practices, financial statement disclosure, or auditing scope or procedure, which, if not resolved to the satisfaction of Ernst & Young LLP, would have caused Ernst & Young LLP to make reference thereto in its reports on our financial statements for such years, and (ii) there have been no reportable events as defined in Item 304(a)(1)(v) of Regulation S-K.

## **Principal Accounting Fees and Services**

In connection with the audit of our financial statements for the fiscal year ended December 31, 2008, we entered into an engagement agreement with KPMG LLP which sets forth the terms by which KPMG LLP performed audit services for Microtune.

The following table sets forth the amount of audit fees, audit-related fees, tax fees and all other fees billed or expected to be billed by KPMG LLP for the years ended December 31, 2008 and December 31, 2009.

	2008	2009
Audit fees <sup>(1)</sup>	\$ 417,522	\$ 475,882(3)
Audit-related fees		
Tax fees <sup>(2)</sup>		
All other fees		
Total Fees	\$ 417,522	\$ 475,882

KPMG LLP did not render professional services relating to financial information systems design and implementation for the years ended December 31, 2008 and 2009.

- (1) Audit fees are generated from services consisting of the annual audits of our consolidated financial statements included in our Annual Reports on Form 10-K, the annual audit on effectiveness of internal control over financial reporting, quarterly reviews of our consolidated financial statements included in our Quarterly Reports on Form 10-Q, as well as any required statutory audits of our foreign subsidiaries, services related to filings made with the SEC and accounting advisory services related to financial accounting matters.
- (2) Tax fees are generated from services, including but not limited to, assistance with certain tax compliance matters.
- (3) Includes fees for the 2009 integrated audit, 2009 quarterly reviews and procedures performed in Plano, Texas, Ingolstadt, Germany and China.

The Audit Committee has determined that the services provided to Microtune are compatible with maintaining KPMG LLP s independence. For more information about KPMG LLP, see the Report of the Audit Committee of the Board of Directors on page 65.

The Audit Committee has adopted policies and procedures for the pre-approval of audit and non-audit services by KPMG LLP. In general, all services must be pre-approved at duly convened meetings of the Audit Committee or by the Chairperson of the Audit Committee. Any services approved by the Audit Committee Chairperson are required to be discussed at the following regular meeting of the Audit Committee. All fees paid in 2009 were approved in accordance with these procedures.

## **Required Vote**

Ratification of KPMG LLP as our independent registered public accounting firm for the fiscal year ending December 31, 2010 requires the affirmative vote of the holders of a majority of the shares of common stock present in person or represented by proxy and entitled to vote. An abstention will be counted as a vote against approval since it is one less vote for approval of the shares present. Broker non-votes will not affect the outcome as they are not considered shares present for voting purposes, however, brokers will have authority to vote shares they hold on the behalf of beneficial holders in favor of approval if they have not been instructed otherwise.

If our stockholders do not ratify the selection of KPMG LLP, the appointment of the independent registered public accounting firm will be reconsidered by our Audit Committee. Even if the selection is ratified, the Audit Committee, in its discretion, may direct the appointment of a different independent accounting firm at any time during the year if the Audit Committee believes that such a change would be in the best interests of Microtune and its stockholders.

### **Recommendation of the Board of Directors**

The Board of Directors and, more specifically, the Audit Committee, unanimously recommends a vote FOR the ratification of the appointment of KPMG LLP to serve as our independent auditor for the year ending December 31, 2010. Unless otherwise instructed, the proxy holders named in each proxy will vote the shares represented thereby FOR the ratification of the appointment of KPMG LLP.

### DIRECTORS, EXECUTIVE OFFICERS AND CORPORATE GOVERNANCE

Information regarding our Corporate Governance can be found beginning on page 8 and information regarding our Board of Directors can be found beginning on page 9.

The following table sets forth certain information with respect to our executive officers as of April 9, 2010:

Name	Age	Position
James A. Fontaine	52	Chief Executive Officer and President
Justin M. Chapman	35	Chief Financial Officer
Robert S. Kirk	49	Vice President of Worldwide Sales
Barry F. Koch	44	Executive Vice President
Phillip D. Peterson	40	General Counsel and Corporate Secretary

Biographical information concerning Mr. Fontaine is set forth under Election of Directors on page 19.

Justin M. Chapman was named Vice President and Chief Financial Officer in February 2010. Mr. Chapman served as the Company s Vice President and Interim Chief Financial Officer from December 2009 to February 2010. He served as Vice President of Accounting of the Company from November 2004 to December 2009 (and was the Principal Financial and Accounting Officer from November 2004 to May 2005). From August 2004 to November 2004, he served as Director of Accounting. From August 2003 to August 2004, he served as Manager of Accounting. From August 2001 to August 2003, he served as Manager of Financial Planning. He is a Certified Public Accountant and holds a BBA degree in Accounting from the University of Oklahoma.

Robert S. Kirk was named Vice President of Worldwide Sales in October 2003. Mr. Kirk served as Vice President of Sales, North America from March 2003 to October 2003. Mr. Kirk was Vice President of Sales, Northern Europe for Avnet from August 2001 to March 2003, and Vice President of Sales, North America for ON Semiconductor from May 1982 to August 2001. Mr. Kirk holds a B.S.E.E. from Purdue University.

Barry F. Koch was named Executive Vice President in June 2009. Mr. Koch also serves as Managing Director of Microtune GmbH & Co. KG, a wholly-owned German subsidiary of Microtune, since May 2000. Mr. Koch joined Microtune GmbH & Co. KG as Director of Advanced Development in January 2000 at the time of Microtune s combination with Temic. Mr. Koch had been with Temic and its predecessors since June 1995. In addition, Mr. Koch has held a number of positions at Microtune concurrently with his positions at Microtune GmbH & Co. KG. Mr. Koch was named Vice President and Co-General Manager of Microtune s Broadband Business Unit in January 2004. Additionally, Mr. Koch has served as Vice President and General Manager of Microtune s Automotive Business Unit since December 2001, and served as Vice President of Systems Engineering from May 2000 until December 2001. Mr. Koch holds a B.S.E.E. from the University of Missouri-Rolla and a M.S.E.E. from Purdue University.

Phillip D. Peterson was named General Counsel in April 2004. Mr. Peterson was an attorney with Cox Smith Matthews Incorporated (formerly Cox & Smith Incorporated) from September 2002 to October 2003 and with DLA Piper (formerly Gray Cary Ware & Freidenrich, LLP) from March 2000 to January 2002. Prior to that, he was an attorney with King & Spalding LLP and Fulbright & Jaworski L.L.P. Mr. Peterson holds a B.A. from Trinity University and a J.D. from the University of Texas School of Law.

# SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT AND RELATED STOCKHOLDER MATTERS

### Security Ownership of Certain Beneficial Owners and Management

The following table shows shares of our common stock that we believe are beneficially owned as of March 1, 2010 by:

each person who we know beneficially owns more than 5% of our common stock;

each of our current directors and director nominees;

each current executive officer named in the Summary Compensation Table; and

all of our current directors, director nominees and current executive officers as a group.

Under the rules of the SEC, beneficial ownership includes voting or investment power with respect to securities and includes the shares issuable under stock options that are exercisable within 60 days of the date set forth above. Accordingly, we have included any options held by each stockholder that are exercisable within 60 days of March 1, 2010 (i.e., April 30, 2010). No RSU awards are currently scheduled to vest within such time period. Shares issuable under stock options are deemed outstanding for purposes of computing the percentage ownership of the person holding options but are not deemed outstanding for purposes of computing the percentage ownership of any other person. As of March 1, 2010, there were 53,998,794 shares of our common stock outstanding.

Name of Beneficial Owner	Shares	Percentage Beneficially Owned
Institutional Venture Partners VII <sup>(1)</sup>	2,738,668	5.1%
FMR LLC <sup>(2)</sup>	6,222,540	11.5%
Simon J. Michael <sup>(3)</sup>	5,079,296	9.4%
Ramius LLC <sup>(4)</sup>	4,950,000	9.2%
Justin M. Chapman <sup>(5)</sup>	120,678	*
Walter S. Ciciora <sup>(6)</sup>	254,138	*
James H. Clardy <sup>(7)</sup>	215,738	*
Steven Craddock <sup>(8)</sup>	140,738	*
James A. Fontaine <sup>(9)</sup>	777,690	1.4%
Robert S. Kirk <sup>(10)</sup>	462,296	*
Barry F. Koch <sup>(11)</sup>	423,642	*
Anthony J. LeVecchio <sup>(12)</sup>	152,738	*
Bernard T. Marren <sup>(13)</sup>	581,800	*
Phillip D. Peterson <sup>(14)</sup>	248,929	*
Michael T. Schueppert <sup>(15)</sup>	107,800	*
William P. Tai <sup>(16)</sup>	2,827,228	5.2%
A. Travis White <sup>(17)</sup>	130,738	*
Robert Rast	0	
Raghavendra (Raghu) Rau <sup>(18)</sup>	3,300	*
Drew Peck	0	
All current directors, director nominees and current executive officers as a group (16 persons) <sup>(19)</sup>	6,447,453	11.4%

- \* Less than 1% of the outstanding shares of common stock.
- (1) Based upon Amendment No. 1 to the Schedule 13G filed with the SEC on February 16, 2010. Includes 54,773 shares held by Institutional Venture Management VII, L.P., 2,597,061 shares held by Institutional Venture Partners VII, L.P. and 86,834 shares held by IVP Founders Fund I, L.P. The address for these funds is 3000 Sandhill Road, Suite 290, Menlo Park, CA 94025.

- (2) Based upon Amendment No. 1 to the Schedule 13G filed with the SEC on February 16, 2010. The address for FMR LLC is 82 Devonshire Street, Boston, MA 02109.
- (3) Based upon Amendment No. 9 to Schedule 13G filed with the SEC on February 10, 2010 by Simon J. Michael, Balch Hill Capital, LLC and Balch Hill Partners, L.P., all whose address is 2778 Green Street, San Francisco, CA 94123. The 13G states that:
  - (i) Simon J. Michael has sole voting power and sole dispositive power with respect to 735,470 shares and shared voting power and shared dispositive power with respect to 4,343,826 shares;
  - (ii) Balch Hill Capital, LLC has shared voting power and shared dispositive power with respect to 4,343,826 shares; and
  - (iii) Balch Hill Partners, L.P. has shared voting power and shared dispositive power with respect to 4,153,826 shares.
- (4) Based upon Amendment No. 3 to the Schedule 13D filed with the SEC on March 26, 2010. The address for Ramius LLC is 599 Lexington Avenue, 20 th Floor, New York, NY 10022.
- (5) Includes 98,783 shares for options which are currently exercisable or become exercisable within 60 days.
- (6) Mr. Ciciora has sole voting power and sole dispositive power with respect to 80,400 shares and shared voting power and shared dispositive power with respect to 38,000 shares. Includes 135,738 shares for options which are currently exercisable or become exercisable within 60 days.
- (7) Includes 25,000 shares of common stock held by trusts, of which Mr. Clardy acts as co-trustee, for the benefit of Mr. Clardy s children, none of whom are dependents of Mr. Clardy. Mr. Clardy has sole voting power and sole dispositive power with respect to 55,000 shares and shared voting power and shared dispositive power with respect to 25,000 shares. Also includes 135,738 shares for options which are currently exercisable or become exercisable within 60 days.
- (8) Includes 140,738 shares for options which are currently exercisable or become exercisable within 60 days.
- (9) Includes 15,653 shares of common stock outstanding held by JSCJ Ventures, Ltd., of which Mr. Fontaine is the general partner. Mr. Fontaine has sole voting power and sole dispositive power with respect to 101,363 shares. Also includes 676,327 shares for options which are currently exercisable or become exercisable within 60 days.
- (10) Mr. Kirk has sole voting power and sole dispositive power with respect to 41,624 shares. Also includes 420,672 shares for options which are currently exercisable or become exercisable within 60 days.
- (11) Mr. Koch has sole voting power and sole dispositive power with respect to 28,240 shares. Also includes 395,402 shares for options which are currently exercisable or become exercisable within 60 days.
- (12) Mr. LeVecchio has sole voting power and sole dispositive power with respect to 23,250 shares. Also includes 129,488 shares for options which are currently exercisable or become exercisable within 60 days.
- (13) Mr. Marren has sole voting power and sole dispositive power with respect to 500,000 shares. Also includes 81,800 shares for options which are currently exercisable or become exercisable within 60 days.
- (14) Mr. Peterson has sole voting power and sole dispositive power with respect to 58,072 shares. Also includes 190,857 shares for options which are currently exercisable or become exercisable within 60 days.
- (15) Mr. Schueppert has sole voting power and sole dispositive power with respect to 50,000 shares. Includes 57,800 shares for options which are currently exercisable or become exercisable within 60 days.
- (16) Includes 54,773 shares held by Institutional Venture Management VII, L.P. and 2,597,061 shares held by Institutional Venture Partners VII, L.P. Mr. Tai is a general partner of each of these partnerships, shares voting and dispositive power with respect to the shares held by each of these entities and disclaims beneficial ownership of the shares held by these entities, except to the extent of his pecuniary interest. Mr. Tai has sole voting power and sole dispositive power with respect to 39,656 shares. Mr. Tai has shared voting power and shared dispositive power with respect to 2,651,834 shares. Also includes 135,738 shares for options which are currently exercisable or become exercisable within 60 days.
- (17) Includes 130,738 shares for options which are currently exercisable or become exercisable within 60 days.
- (18) Mr. Rau has shared voting power and shared dispositive power with respect to 3,300 shares.
- (19) Includes 2,729,819 shares for options which are currently exercisable or become exercisable within 60 days.

### **Equity Compensation Plan Information**

Information regarding stock-based compensation awards outstanding and available for future grants as of December 31, 2009, segregated between stock-based compensation plans approved by stockholders and stock-based compensation plans not approved by stockholders, is presented in the table below:

Plan Category	Number of Securities to be Issued Upon Exercise of Outstanding Options, Warrants and Rights and Vesting of RSU Awards	Exerci Outstand Warı	ed-Average se Price of ling Options, rants and tights	Number of Securities Remaining Available for Future Issuance under Equity Compensation Plans	
Equity compensation plans approved by security holders	11,042,113	\$	3.546	3,632,326	
Equity compensation plans not approved by security holders		·			
Total	11,042,113	\$	3.546	3,632,326	

### SECTION 16(a) BENEFICIAL OWNERSHIP REPORTING COMPLIANCE

Section 16(a) of the Exchange Act requires our officers, directors and persons who own more than 10% of our common stock to file certain reports regarding ownership of, and transactions in, our securities with the SEC. Such officers, directors and 10% stockholders are also required by SEC rules to furnish us with copies of all Section 16(a) forms that they file.

Based solely on our review of such forms furnished to us and written representations from certain reporting persons, we believe that all Section 16(a) filing requirements applicable to our executive officers, directors and more than 10% stockholders were complied with during the fiscal year ended December 31, 2009.

### **EXECUTIVE COMPENSATION**

### COMPENSATION DISCUSSION AND ANALYSIS

This Compensation Discussion and Analysis is intended to provide investors with an understanding of our compensation policies and decisions regarding our named executive officers for 2009. Our named executive officers are our Chief Executive Officer, Chief Financial Officer and our three other most highly compensated executive officers. The following will be discussed and analyzed in this Compensation Discussion and Analysis section:

Microtune Compensation Philosophy;
Executive Compensation Objectives and Policies;
Executive Compensation Components;
Setting Executive Compensation;
Role of Executive Officers in Compensation Decisions;
Base Salary Compensation;
2009 Salary Adjustments;
Annual Incentive Compensation;
Suspension of Annual Incentive Compensation Program in 2009;
2010 Incentive Compensation Program;
Long-Term Equity Incentive Compensation;
Equity Awards and Practices;
Annual RSU Awards;
New Hire RSII Awards

Retirement Benefits;
Other Compensation;
Severance Benefits;
Severance and Change of Control Agreements;
Potential Payments Upon Termination or Change of Control;
Executive Compensation During 2009;
Suspension of Annual Incentive Compensation Program and Base Salary Increases;
Stock Ownership Guidelines and Policy Against Hedging;
Tax Deduction for Executive Compensation; and
Compensation Committee Report.
Microtune Compensation Philosophy
Key to establishing long-term and short-term compensation plans is the establishment of a compensation philosophy, and annually reviewin

44

and adapting that philosophy to match the evolving industry environment and Microtune s place in that industry.

Over the past few years, we believe Microtune has evolved from the emerging growth company stage of its development and is in the process of transitioning to a more mature stage of its development. In 2006, 2007, and 2008, our Compensation Committee spent considerable time discussing and updating our compensation philosophy for our executives to reflect the current evolution of Microtune and to take into account the impact of Financial Accounting Standards Board Accounting Standards Codification Topic 718 Stock Compensation (FASB ASC Topic 718) and our heavy historical reliance on long-term equity incentive compensation. Our current executive compensation philosophy is characterized by the following:

base salaries at market;

annual incentive compensation programs based primarily on the achievement of a combination of revenue and profit goals with annual incentive bonuses comprised of the performance vesting of RSUs; and

long-term equity incentive compensation with the historical use of stock options transitioning to the use of a combination of stock options and RSU awards in 2007, at a reduced rate from historical levels, and further transitioning to 100% RSU awards for executive officers in 2008 and 2009.

Our Compensation Committee has also developed our non-executive compensation philosophy to a more traditional, larger company model in order to better attract and retain professional employees. This new compensation philosophy for our non-executive employees is characterized by the following:

at or above market base salary compensation;

annual incentive compensation programs for key professional employees based primarily on the achievement of a combination of revenue and profit goals;

long-term equity incentive compensation with the historical use of stock options transitioning to the use of a combination of stock options and RSU awards in 2007, at a reduced rate from historical levels, and further transitioning to 100% RSU awards for all employees in 2008 and 2009; and

addition of a 401(k) matching program to encourage employees to fund their own retirement programs.

We believe that any risks arising from our compensation policies and practices for any of our employees are not reasonably likely to have a material adverse effect on the company.

### **Executive Compensation Objectives and Policies**

The Compensation Committee is responsible for setting, administering and reviewing our compensation programs and the policies governing the annual compensation of our named executive officers and the other officers appointed by our Board of Directors. The role of the Compensation Committee is to establish and recommend salaries and other compensation paid to the Chief Executive Officer and the other executive officers

and to administer our stock and benefit plans. For executive officers, the Board approves all base salaries, annual incentive compensation and long-term equity incentive compensation. Our Chief Executive Officer abstains from all voting actions of the Board regarding executive compensation. Any discretionary bonus payments are also approved by our full Board. Our Compensation Committee, which met 12 times during 2009, is composed entirely of independent directors. See Microtune Corporate Governance Board Meetings and Committees; Annual Meeting Attendance Compensation Committee on page 13. Our Board, upon the recommendation of our Nominating and Corporate Governance Committee, designates the members and the Chairperson of our Compensation Committee. Our primary compensation goals are to:

attract and retain highly qualified people;

motivate existing employees who are critical to the achievement of our objectives and reward them for meeting or exceeding our goals;

more closely align the interests of our employees with those of our stockholders; and

simultaneously maximize stockholder value.

Accordingly, our executive compensation philosophy is to tightly link compensation with Microtune sperformance and the creation of stockholder value. We also structure compensation programs so that individuals are held accountable for their performance and our results. Our compensation programs are designed to:

reward executives for long-term strategic performance and enhancement of stockholder value;

support a performance-oriented environment that rewards achievement of our financial and non-financial goals;

promote ownership of Microtune stock by our executives to better align the interests of management and our stockholders;

attract and retain executives whose abilities are considered essential to our long-term success and competitiveness; and

align the long-term financial interests of our executives with those of our stockholders by placing a significant percentage of executive compensation at risk through the use of equity-based compensation components.

Executive Compensation Components. To meet the objectives described above, executive compensation for Microtune s officers is comprised of three main components: (1) base salary; (2) annual incentive compensation and (3) long-term equity incentive compensation. Each of these compensation components is discussed below. The overall compensation package for our executives is variable with actual executive compensation related to and contingent upon Microtune s performance and stock price through the use of annual incentive compensation and long-term equity incentive compensation, respectively. The equity incentive plans, pursuant to which such awards are made, have been approved by our stockholders and are designed to comply with the requirements of United States federal tax laws on deductibility under Section 162(m) of the Code.

Setting Executive Compensation. Our fiscal year aligns with the calendar year and as such, we develop and finalize our annual and long-term goals in the fourth quarter of the prior year or the first quarter of each year. We review Microtune s performance for the prior fiscal year during the first quarter of each year. As part of this review of Microtune s performance, we assess whether any annual incentive compensation has been

earned. We also review progress towards our performance goals each quarter during the year. In determining overall compensation for a specific officer, we consider many factors, including the responsibilities and scope of the officer s particular position, his or her experience and performance in that position, the recent compensation history of the officer (including special or unusual compensation payments), the period of time since his or her last increase in total compensation, the expected value of the officer s contribution to our future success and growth, the compensation levels of all employees within Microtune to ensure internal pay equity is preserved,

executive compensation at certain other companies to ensure that our compensation levels are competitive and our recent overall financial and business performance. We review and consider the compensation of all levels of employees within Microtune to provide an appropriate context for executive compensation decisions and to help ensure that there is internal pay equity within the organization.

We use, among other things, the AON Consulting Radford Executive Survey (the Radford Survey ) to set base salary and annual incentive compensation. Base salary adjustments, if any, are discretionary and typically occur in the second quarter of each year. The companies currently comprising our Radford Peer Group are technology companies with annual revenue from \$50M to under \$200M:

Advanced Analogic Technologies Aeroflex Colorado Springs Cirrus Logic

Cortina Systems

DPIX

Ebara Technologies Entropic Communications Exar
JSR Micro
Magnum Semiconductor
Mindspeed Technologies
Netlogic Microsystems
Pericom Semiconductor
PLX Technology

Rambus Sematech TowerJazz

Trident Microsystems

Volterra

The Compensation Committee sets annual and long-term performance goals for our Chief Executive Officer and evaluates our Chief Executive Officer s performance against such goals and reviews Microtune s stock performance against the performance of the Philadelphia Semiconductor Index. The Compensation Committee meets at least once each calendar year in executive session without our Chief Executive Officer, and he is not present for the portion of any meeting of the Compensation Committee during which his compensation is considered. The aggregate amount of compensation (base salary, annual incentive compensation and long-term equity incentive compensation) of Mr. Fontaine for fiscal 2009 falls below the median of the Radford Peer Group.

The Compensation Committee does not formally retain any consultants to assist it in assessing and determining appropriate and competitive compensation for our executives.

Role of Executive Officers in Compensation Decisions. Our Chief Executive Officer makes a recommendation to the Compensation Committee each year on the appropriate compensation to be paid to our executive officers based on his review of companies with revenue between \$50 million and \$200 million, including a peer group of semiconductor companies, included in the Radford Survey. Although our Chief Executive Officer does not provide recommendations for his base salary compensation, he does provide annual incentive compensation and long-term equity incentive compensation recommendations for all of our executive officers, including himself. Our Board makes the final determination of the amount of compensation to be awarded to each executive officer, including our Chief Executive Officer, based on the Compensation Committee s recommendations and determination of how that compensation achieves the objectives of our compensation policies.

### **Base Salary Compensation**

We target the base salary for each executive officer position to be competitive with the base salaries of executive officers of semiconductor companies of similar size and complexity based on our review of companies included in the Radford Survey. Our Compensation Committee annually reviews salaries recommended by the Chief Executive Officer for each of our executive officers (other than himself) and makes a recommendation to the Board regarding the appropriate salary of each executive officer on a case-by-case basis. Final decisions on base salary adjustments of executive officers other than the Chief Executive Officer are made with the Chief Executive Officer s involvement, however he abstains from all voting actions. In addition, our Compensation Committee annually reviews the salary of our Chief Executive Officer in executive session without management

present and makes a recommendation to the Board. Our Chief Executive Officer is not present during Board discussions or voting regarding his compensation. In determining the appropriate salary levels for our executive officers, we consider, among other factors:

responsibilities and scope of the position held and the officer s experience and performance in that position;

recent compensation history of the officer (including special or unusual compensation payments);

period of time since the last base salary increase;

expected value of the officer s contribution to our future success and growth;

compensation of all levels of employees within Microtune to ensure internal pay equity is preserved;

executive compensation at certain other companies to ensure that our compensation levels are competitive based on our review of companies included in the Radford Survey; and

our recent overall financial and business performance.

In addition, executives and other employees may receive an additional increase in base salary if warranted because of promotions, retention concerns and market conditions. The base salaries paid to our named executive officers in 2009 can be found below in the Summary Compensation Table.

2008 Salary Adjustments

On April 29, 2008, the Board of Directors, upon the recommendation of the Compensation Committee, approved increases in the annual base salaries for the company s named executive officers and officers subject to Section 16 of the Securities and Exchange Act of 1934, as amended, effective May 1, 2008 as set forth in the table below:

<b>Executive Officer</b>	Executive Officer Position							
James A. Fontaine	Chief Executive Officer and President	\$	350,000					
Jeffrey A. Kupp <sup>(1)</sup>	Former Chief Financial Officer	\$	256,025					
Albert Taddiken <sup>(2)</sup>	Former Chief Operating Officer	\$	256,025					
Robert S. Kirk	Vice President of Worldwide Sales	\$	166,002					
Barry F. Koch <sup>(3)</sup>	Executive Vice President and Managing Director of		147,624 <sup>(4)</sup>					
	Microtune GmbH & Co. KG							
Phillip D. Peterson	General Counsel	\$	210,000					

- (1) Mr. Kupp resigned from the position of Chief Financial Officer effective December 4, 2009.
- (2) Mr. Taddiken resigned from the position of Chief Operating Officer effective June 10, 2009.
- (3) Mr. Koch was promoted to the position of Executive Vice President effective June 10, 2009. Mr. Koch continues to serve as Managing Director of Microtune GmbH & Co. KG.

### (4) Mr. Koch is compensated in Euros.

In considering the proposed changes, the Compensation Committee considered each officer s performance measured in accordance with the factors listed above. The Compensation Committee increased the base salaries of the four named executive officers, other than the Chief Executive Officer, an average of 4.5% over their 2007 base salaries.

2009 Salary Adjustments

In 2009, in light of the macroeconomic environment and the associated challenges facing the company, the Board of Directors, upon the recommendations of management and the Compensation Committee, determined not to increase the annual base salaries of our management and to maintain the 2008 base salaries reflected above. As a result, we did not consult the Radford Survey. However, during 2009, we promoted Justin M. Chapman to Chief Financial Officer and, in

connection with such promotion, we consulted the Radford Survey and again reviewed our Radford Peer Group to target his base salary compensation at approximately the 5<sup>th</sup> percentile or \$190,000, because our compensation philosophy is to gradually adjust base salary compensation of newly promoted named executive officers up to the 50<sup>th</sup> percentile range incrementally over a several year period.

On October 13, 2009, the Company s Board of Directors, upon the recommendation of the Compensation Committee and as part of the restructuring plan disclosed on a Current Report on Form 8-K filed on October 16, 2009, approved a five percent (5%) reduction in the base salary of the following officers of the Company:

Executive Officer	Position	Adjuste	ed Base Salary
James A. Fontaine	Chief Executive Officer and President	\$	332,500
Jeffrey A. Kupp <sup>(1)</sup>	Former Chief Financial Officer	\$	243,224
Robert S. Kirk	Vice President of Worldwide Sales	\$	157,702
Barry F. Koch <sup>(2)</sup>	Executive Vice President and Managing Director of		154,267(3)
	Microtune GmbH & Co. KG		
Peter Birch	Vice President and General Manager, DTV Business	\$	215,000

- (1) Mr. Kupp resigned from the position of Chief Financial Officer effective December 4, 2009.
- (2) Mr. Koch was promoted to the position of Executive Vice President effective June 10, 2009 and received a base salary increase upon this promotion. Mr. Koch continues to serve as Managing Director of Microtune GmbH & Co. KG.
- (3) Mr. Koch is compensated in Euros.

The reductions in base salary became effective on October 16, 2009 and were to continue for a period of eighteen (18) months, unless the Board of Directors determined in its discretion to otherwise modify such base salary compensation. The Board of Directors subsequently determined that if the Company is profitable on a Non-GAAP basis in the first quarter of 2010 that the original salaries would be restored.

### **Annual Incentive Compensation**

Prior to 2006, we had not implemented a formal annual incentive compensation program for our executive officers. However, with the approval of our full Board, we had previously awarded discretionary bonus compensation to certain executives for significant accomplishments that contributed to our results. Consistent with our objective to support a pay for performance compensation philosophy, any annual incentive compensation plans for senior executives must link pay to the achievement of financial goals set in advance by the Compensation Committee, which further the interests of Microtune. In determining each individual executive officer s bonus opportunity, the Compensation Committee considers the input that each individual executive officer can have on the profit and revenue scores in a given year. The Compensation Committee reviews the Company s financial performance and considers whether established targets of an annual incentive compensation program have been met in deciding whether to recommend to the Board to award annual incentive compensation and in what amounts.

Suspension of Annual Incentive Compensation Program for 2009

Upon the recommendations of management and the Compensation Committee, the Board of Directors determined not to adopt an annual incentive compensation program for fiscal year 2009 based on the following factors:

The challenges of forecasting future financial performance in the current macroeconomic environment;

The desirability of limiting compensation expense, limiting stockholder dilution, and preserving cash in the current economic environment; and

The compensation philosophy that establishing annual incentive compensation where year over year financial growth is not likely and where the Company faces potential macroeconomic challenges is not appropriate.

### 2010 Incentive Compensation Program

On February 10, 2010, the Compensation Committee and the Board, upon the Compensation Committee s recommendation, approved a new incentive compensation program for fiscal 2010 (the 2010 Program). The 2010 Program covers executive officers, key managers, and other key employees and provides for incentive compensation to be paid (to the extent any such compensation is earned) 100% through the performance vesting of restricted stock unit awards.

All payments and awards under the 2010 Program are conditioned on Microtune meeting the revenue and profitability goals described in the 2010 Program for fiscal 2010. The restricted stock unit awards to be made under the 2010 Program are designed to provide an additional alignment of the interests of our executive management and key employees with the interests of stockholders. In accordance with the terms of the 2010 Program and pursuant to the 2000 Stock Plan, the Board approved the terms of the restricted stock unit awards made to our executive officers and key employees covered by the 2010 Program. Under the 2010 Program, the number of the total restricted stock unit awards that will actually vest and result in the issuance of underlying shares is calculated as described below.

**Profitability**. We have established targets for profitability under the 2010 Program, measured by 2010 Non-GAAP operating income, which is 2010 GAAP operating income, excluding: (1) FASB ASC Topic 718 stock-based compensation expense, amortization of intangibles, restructuring charges, litigation related expenses, and the cost of the 2010 Program itself. We established a series of Non-GAAP operating income levels with corresponding profit scores. The profit scores range from 0% to 70% based upon Microtune s achievement of Non-GAAP operating income goals.

**Revenue**. In addition, we established a series of revenue levels with corresponding revenue scores under the 2010 Program. The revenue scores vary from 0% to 30% based upon Microtune s achievement of these revenue goals.

To determine the total bonus score, the Compensation Committee will add the profit score (i.e. a range from 0% to 70%) to the revenue score (i.e. a range from 0% to 30%). The total bonus score (i.e. 0% to 100%) will then be used to determine the award of each of the participants in the 2010 Program based on predetermined individual bonus percentiles of base salary.

The maximum total bonus score under the 2010 Program requires both: (i) 2010 Non-GAAP operating income of at least \$6.6 million and (ii) 2010 net revenue of approximately \$102 million.

An aggregate number of 1,405,282 restricted stock unit awards were made under the 2010 Program to our executive officers and other key employees in differing amounts based on their respective responsibilities and each individual s unique ability to impact Microtune s financial performance and business objectives. The projected restricted stock unit award vesting at Plan and Maximum for named executive officers are described below:

		RSU Aw	ard Shares
Executive Officer	Position	At Plan	Maximum
James A. Fontaine	Chief Executive Officer	22,167	147,778
Justin M. Chapman	Chief Financial Officer	9,491	63,270
Barry F. Koch	Executive VP and Managing Director (Microtune	14,557	97,044
	GmbH & Co. KG)		
Robert S. Kirk	Vice President of Worldwide Sales	10,513	70,089

Phillip D. Peterson General Counsel 9,800 65,333

Plan describes Microtune s expectations regarding fiscal year 2010 financial performance at the time of the adoption of the 2010 Program.

In addition, we granted restricted stock unit awards under the 2010 Program to other key managers and key employees (who are not named executive officers) totaling 961,768 restricted stock units. Approximately 1,451,000 shares were returned to the 2000 Stock Plan as a result of the recent departure of two senior executives.

The number of restricted stock unit s granted under the 2010 Program that shall vest will be determined based on a review and evaluation by the Compensation Committee of our performance in fiscal 2010 based upon the terms and conditions of the 2010 Program. Any portion of an executive or key employee s restricted stock unit bonus award that does not vest upon the Compensation Committee s determination shall be forfeited. A participant in the 2010 Program must be an employee of Microtune on the date of the bonus payment to receive his bonus compensation under the 2010 Program. The 2010 Program will terminate according to its terms after the vesting and award determination is made by the Compensation Committee. We may adopt other incentive compensation programs in the future.

### **Long-Term Equity Incentive Compensation**

Microtune currently awards long-term equity incentive compensation to executive officers as part of its overall compensation package. These awards differ from annual incentive compensation by rewarding executives for long-term stockholder value creation, rather than short-term annual financial performance. These awards are consistent with Microtune s policies of offering competitive overall compensation packages, while furthering Microtune s pay for performance compensation philosophy. Long-term equity incentive compensation awards are an important and significant element of the total compensation package for Microtune executives and serve to align the financial interests of our executives with the financial interests of our stockholders and to supplement base salaries which we have historically set at or below market base salaries. The number of shares of our common stock underlying an award and any subsequent awards are based in part on the competitive environment of the technology industry, while also maintaining an overall sense of fairness in the compensation distributed to Microtune employees. When granting long-term equity incentive compensation awards, we consider the current rate and amount of annual vesting of equity-based awards for each individual and whether we believe that such vesting rates and amounts are competitive.

Beginning in 2006, we began to consider the use of RSU awards as long-term equity incentive compensation for key executives in place of traditional incentive stock option grants, initially by using performance-based RSU awards in conjunction with our annual incentive compensation program. We believe RSU awards have certain advantages over stock option grants. Generally, fewer shares of RSUs are awarded as compared to comparable stock option grants, resulting in less dilution to our current stockholders. In addition, depending on our future equity-based incentive award practices, RSU awards could potentially have a lesser impact on Microtune s reported financial results as compared to stock option grants.

In April 2008, our Board of Directors, upon the recommendations of management and the Compensation Committee, determined to make 100% of all future long-term equity incentive compensation awards exclusively in the form of RSU awards, rather than a mix of stock option awards and RSU awards. These RSU awards were made in May 2008 and May 2009 and vest in full in May 2012 and 2013, respectively.

Methodology for Setting Long-Term Equity Incentive Compensation Awards. Originally, the annual stock option grant of 200,000 option shares for the Chief Executive Officer was based on the assumption of a 15% return per year on the common stock of the Company, and a return of 100% to 200% of the annual base salary of the Chief Executive Officer four years from the date of the grant. The annual option grant vested over four years and was based on the stock price at the time of the grant. Once this run rate was established, subsequent annual stock option awards were made in the same manner.

As discussed above, we gradually transitioned from granting stock options to granting RSU s at a reduced quantity but with the same compensation philosophy. The annual stock option or RSU awards for the other named executive officers were then determined using a percentage of the annual award made to the Chief Executive Officer. Generally, the other named executive officers receive annual long-term equity incentive compensation awards in the range of 30% to 60% of the annual long-term equity incentive compensation award that the Chief Executive Officer receives, based on such officer s seniority, performance and responsibility.

In May 2009, we made annual RSU awards to our named executive officers and certain other officers pursuant to the 2000 Stock Plan that vest in May 2013. See the Summary Compensation Table and the Grants of Plan-Based Awards in 2009 table on pages 57 and 59, respectively.

Equity Awards and Practices. Our equity award granting procedures were substantially modified in June 2006 and then again in the first quarter of 2007 as a result of the remediation of deficiencies discovered by management and by the Audit Committee in its investigation of our stock option granting practices, and the desire to further strengthen our internal controls. Our Equity Compensation Award Policy was adopted by our Board of Directors on December 28, 2006 to address these deficiencies and to formalize our equity award granting procedures. This policy was substantially modified in July 2008 to reflect our complete transition from stock option awards to RSU awards and remains in effect. This policy has specific procedures governing the granting of equity awards and new hire equity awards, including:

We have designated an employee highly trained in the legal and accounting implications of equity awards (the Equity Plan Administrator ) to be responsible for administering and monitoring equity awards. This employee reports to, and his work is reviewed by, our Chief Financial Officer and General Counsel. His responsibilities include, among other things, (1) reviewing all proposed awards before such awards are submitted to the Board or Compensation Committee to ensure that such awards are being made in accordance with applicable law and corporate authority and to ensure that such awards are complete and accurate in all respects; and (2) ensuring that the exercise price established for any stock option grant is equal to or higher than the fair market value of the underlying common stock on the grant date by comparing such exercise price to the closing price of our common stock on the same date that the grant has occurred;

Except in unusual circumstances, we ensure that all equity awards are reflected in resolutions set forth in the minutes of a meeting of the Board or Compensation Committee. Such minutes are prepared promptly following the action taken at the meeting. If stock option grants are awarded pursuant to a unanimous written consent, we take care to document the circumstances requiring the use of a unanimous written consent. Additionally, the unanimous written consent is signed by the appropriate persons and returned to us for inclusion in our records by the date specified as the grant date:

Except under unusual circumstances, all equity awards to new hires in connection with their offers of employment are awarded once a quarter with an effective grant date of either February 15, May 15, August 15, or November 15 and are approved in a meeting that occurs on or prior to such grant date;

Annual grants to employees for performance or retention are made during the same time periods each year and are awarded with an effective grant date of either May 15 or August 15 and are approved in a meeting that occurs on or prior to such grant date;

No annual stock option grants are made during any period designated as a quarterly blackout period under our insider trading policy;

Our equity award procedures are in writing and are reviewed by our Chief Financial Officer, General Counsel and outside counsel annually to ensure compliance with all applicable laws and regulations; and

In no case are awards made to employees prior to the actual first day of employment.

Our Equity Compensation Award Policy provides that any deviation from the policy must receive prior approval from our Compensation Committee or Board. Such approval may only be given after full consideration of the market timing issues and legal risks associated with any such deviation. The policy generally provides that concerns regarding the timing of any grant should be communicated to our General Counsel or outside counsel by the Chairperson of the Compensation Committee. Finally, the policy provides for documentation of any such deviation.

Annual RSU Awards. We intend to present proposed annual RSU awards to our Compensation Committee and Board of Directors at our upcoming Compensation Committee and Board of Directors meetings in April 2010. Although the proposed annual awards will be approved in April 2010, they will not actually be granted under our 2000 Stock Plan until May 15, 2010, in accordance with the terms of our Equity Compensation Award Policy. We intend to make annual awards for future years, if any, according to a schedule that is substantially similar to that described above. See Equity Awards and Practices above for more information regarding our Equity Compensation Award Policy and procedures put into place with respect to our grants of equity compensation.

The Compensation Committee administers our 2000 Stock Plan, 2000 Director Stock Plan and 2000 Employee Stock Purchase Plan.

*New Hire RSU Awards*. Beginning in the second quarter of 2008 with the implementation of our new RSU award procedures, if an offer to a prospective employee is accepted, the proposed RSU award for the new employee will be submitted to the Compensation Committee for consideration and approval at its next quarterly meeting.

Prior to the quarterly Compensation Committee meeting, the supporting information, including the offer approval form, offer letter and schedule of proposed RSU awards are reviewed by our Equity Plan Administrator. Our General Counsel also prepares form resolutions with an attached schedule (the RSU Schedule ) describing all of the material terms of the RSU awards to be considered for approval by the Compensation Committee.

Our Equity Plan Administrator reviews the RSU Schedule prior to its circulation to the Compensation Committee. Prior to the Compensation Committee meeting, the form resolutions and RSU Schedule to be approved are circulated to the members of the Compensation Committee for their review.

The grant date for new hire RSU awards is one of the following dates immediately after the date that the Compensation Committee or the Board of Directors approves the subject RSU awards: February 15, May 15, August 15, or November 15.

After the RSU awards are approved by our Compensation Committee (or Board of Directors), our General Counsel (or outside counsel) prepares minutes reflecting the RSU awards approved in the meeting. These minutes are then sent to the Equity Plan Administrator for his further review. The Equity Plan Administrator reviews all of the terms of the RSU awards and verifies compliance with the Equity Compensation Award Policy. Once the minutes are approved by the Compensation Committee at its subsequent meeting, they are included in the corporate minute books of the Company by our General Counsel.

Although we are currently using RSU awards rather than stock option grants for both new hire equity awards and annual equity awards, our Compensation Committee and Board of Directors may determine in the future to recommend or to make future awards of stock option grants from time to time in their discretion.

### **Retirement Benefits**

We offer our named executive officers the ability to defer compensation pursuant to a 401(k) plan, which is available to all of our U.S. employees. Our Board of Directors, upon the recommendation of our Compensation Committee, approved a 401(k) matching program for all U.S. employees, effective January 1, 2007. In 2010, we reduced the amount of the Company match from \$2,000 to \$500.

### Other Compensation

Other non-cash benefits that are offered to the other employees are provided to the executive officers in accordance with our established programs. Other elements of executive compensation include medical and life insurance benefits and the ability to defer compensation pursuant to a 401(k) plan as described above. No other special plans or benefits are offered to our executive officers which are not generally made available to all other employees.

### **Severance Benefits**

We are party to severance and change of control agreements with our executive officers that provide for severance benefits under certain circumstances. These agreements are described below under Severance and Change of Control Agreements. The Compensation Committee believes that the benefits provided by these agreements are in the best interests of the Company s stockholders as these agreements ensure management continuity both before and in the event of a change of control transaction, thereby preventing the potential for lost transaction value due to the loss of key management personnel during the pendency of a transaction. Additionally, the Compensation Committee believes that these agreements are important for attracting and retaining executives who are critical to our long-term success and competitiveness.

Severance and Change of Control Agreements. On March 4, 2010, Microtune entered into Amended and Restated Severance and Change of Control Agreements with Messrs. Fontaine, Koch, Chapman, Peterson and Kirk. Pursuant to these agreements, if any of the foregoing officers are terminated without cause or are terminated without cause within six months after a change of control transaction (as defined in such agreements), such officer is entitled to a lump sum payment equal to his base annual compensation plus the maximum bonus (and sales commission in the case of Mr. Kirk) available to him under any bonus program offered by Microtune during the year in which the termination occurs, or if the officer is not eligible for a bonus in the year in which the termination occurs, the maximum bonus available to him under any bonus program offered by Microtune in the most recent prior year for which a bonus was available. In addition, all unvested equity awards held by such officer that would have vested over the twelve month period following such officer s termination will immediately vest. If such officer s employment is terminated more than six months but less than twelve months following a change of control, he is entitled to a lump sum payment equal to 50% of his base annual compensation plus 50% of the maximum bonus available to him under any bonus or commission program offered by Microtune during the year in which the termination occurs, or if such officer is not eligible for a bonus or commission in the year in which the termination occurs, the maximum bonus available to him under any bonus program offered by Microtune in the most recent prior year for which a bonus was available. In addition, all unvested equity awards held by such officer that would have vested over the six month period following such officer s termination will immediately vest.

None of our named executive officers (other than Mr. Barry F. Koch, as is customary in Germany) has any employment agreement with us, other than the Severance and Change of Control Agreements described above. The terms and conditions of the Severance and Change of Control Agreements were determined unilaterally by the Compensation Committee and the Board of Directors and were not the subject of arms-length negotiations between the Company and the executive officers covered under such agreements.

Potential Payments Upon Termination or Change of Control. The table below reflects the amount of compensation to be paid to each of our named executive officers upon termination or upon termination after a change of control. If any of our named executive officers with whom we have entered into a severance and

change of control agreement were to have been terminated by us on December 31, 2009 as the result of a Constructive Termination, as defined in the agreements, or for any reason other than cause, assuming such termination occurred, in the case of Messrs. Fontaine, Chapman, Kirk, Koch, and Peterson either in the absence of a change of control or within six months following a change of control, each such named executive officer would have been entitled to the following:

a lump sum payment representing the named executive officer s annual base salary;

a lump sum payment representing the maximum bonus available to the named executive officer under any annual bonus program applicable to such named executive officer;

the immediate vesting of all equity awards held by the named executive officer on the date of termination as if such officer had been employed for an additional twelve months from the date of termination; and

a lump sum payment representing the pro rata amount of the annual bonus accrued for the named executive officer under the any annual bonus program applicable to such named executive officer, which would be determined by multiplying the maximum bonus available to such officer by the ratio of the number of days in which such officer was employed during the applicable bonus period to the number of days in such bonus period.

								Acc	rued			
		Ba	se Salary		Bonus	E	cquity	Bo	nus	All Othe	r	Total
Executive Officer <sup>(1)</sup>	Position	Con	npensation	Con	npensation	Comp	ensation(V	omper	nsation(3	Compensati	on <sup>(6)</sup>	Severance
James A. Fontaine	Chief Executive Officer	\$	350,000	\$	219,100	\$	65,018	\$	0	\$ 11,3	36	\$ 645,454
Justin M. Chapman	Chief Financial Officer and Former Interim Chief Financial Officer	\$	164,850	\$	81,380	\$	6,780	\$	0	\$	0	\$ 253,010
Robert S. Kirk	Vice President of Worldwide Sales	\$	166,002	\$	257,862 <sup>(4)</sup>	\$	31,640	\$	0	\$ 18,7	28	\$ 474,232
Barry F. Koch <sup>(5)</sup>	Executive Vice President and Managing Director, Microtune GmbH & Co. KG	\$	232,748	\$	118,940	\$	27,120	\$	0	\$	0	\$ 378,808
Phillip D. Peterson	General Counsel	\$	210,000	\$	100,160	\$	18,080	\$	0	\$ 5,6	97	\$ 333,937

- (1) In 2009, we did not have an incentive compensation program. Under the terms of the agreements, if there is no program in effect in the year of the termination of employment, then the most recent bonus program is used. Accordingly, the amounts in this column represent the total dollar value of (i) the maximum cash awards under the 2008 Incentive Compensation Program plus (ii) the maximum RSU awards under the 2008 Incentive Compensation Program, valued at \$2.26 per share, which was the fair market value of our common stock on December 31, 2009.
- (2) The amounts in this column represent the dollar value of the shares underlying restricted stock units valued at \$2.26 per share, which was the fair market value of our common stock on December 31, 2009 which were held by the named executive officers and that would have vested had such officer been employed for an additional twelve months from the date of termination pursuant to the terms of the agreements.
- (3) There is no accrued bonus compensation payable because there was no incentive compensation program in effect in 2009.
- (4) Includes the maximum amount payable to Mr. Kirk pursuant to the Executive Officer Sales Incentive Compensation Plan.
- (5) Mr. Koch is compensated in Euros; this amount reflects the conversion from Euros to U.S. dollars using the exchange rate in effect on December 31, 2009.
- (6) The amounts in this column represent one year of COBRA payments for applicable officers.

## Stock Ownership Guidelines and Policy Against Hedging

We strongly believe that the financial interests of our executives should be aligned with those of our stockholders. Accordingly, we maintain stock ownership guidelines for each of our directors as well as for James A. Fontaine, Chief Executive Officer, Barry F. Koch, Executive Vice President and Justin M. Chapman, Chief Financial Officer. Pursuant to these guidelines, any of the persons covered by the guidelines who acquires our common stock via exercise of options granted after January 1, 2005 must retain 33% of the net shares acquired (after taking into account the sale of shares to pay taxes and the stock option exercise price), for at least 12 months or such earlier time as the individual ceases to

be a director of or covered officer of Microtune as a result of death, resignation, termination or other reason. Shares already owned on the exercise date by any director or covered officer may be used to satisfy the retention requirements.

Short sales, puts, calls and other hedging transactions of Microtune stock by any Microtune director or employee are expressly prohibited by our corporate policies.

### **Tax Deduction for Executive Compensation**

We have considered the potential impact of Section 162(m) of the Code and the regulations thereunder. Section 162(m) disallows a tax deduction for any publicly held corporation for individual compensation exceeding \$1 million in any taxable year for any of the executive officers, unless such compensation is performance-based within the meaning of applicable IRS requirements. Since the cash compensation of each of our executive officers is below \$1 million and because we believe that any stock options or RSU awards granted under the 2000 Stock Plan will meet the IRS requirements of being performance-based, we believe that Section 162(m) will not reduce any tax deduction available to Microtune for the tax year ended December 31, 2009. It is our policy that all executive compensation arrangements shall comply with Section 162(m) of the Code by not providing additional compensation or equity compensation to avoid tax penalization.

### **Compensation Committee Report**

In accordance with its written charter adopted by our Board, the Compensation Committee has oversight of compensation policies designed to align compensation with our overall business strategy, values and management initiatives.

The Compensation Committee has reviewed and discussed the Compensation Discussion and Analysis with our management, which has the responsibility for preparing the Compensation Discussion and Analysis. Based upon this review and discussion, the Compensation Committee recommended to our Board that the Compensation Discussion and Analysis be included in this proxy statement and incorporated by reference in our Annual Report on Form 10-K filed on February 16, 2010 with the SEC for the fiscal year ended December 31, 2009.

COMPENSATION COMMITTEE OF THE BOARD OF DIRECTORS

James H. Clardy, Chairperson Bernard T. Marren Michael T. Schueppert

#### SUMMARY COMPENSATION TABLE

The following table sets forth information concerning compensation awarded to, earned by, or paid to our named executive officers, which include our Chief Executive Officer, Chief Financial Officer and our three other most highly compensated executive officers for the fiscal years ended December 31, 2009, 2008, and 2007:

Name and Principal Position	Year	Salary (\$)	Bonus (\$) <sup>(1)</sup>	Stock Awards (\$) <sup>(2)</sup>	Option Awards (\$)(2)	Non-Equity Incentive Plan Compensation (\$)(3)	Change in Pension Value and Nonquali- fied Deferred Compensation Earnings (\$)	All Other Compensation (\$)	Total (\$)
James A. Fontaine	2009	( )	(Ψ)	172,000	(4)	Ψ)	(4)	10,077 <sup>(4)</sup>	528,431
	2008	343,333		529,700		32,287		15,461 <sup>(5)</sup>	920,781
Chief Executive Officer and									
President	2007	320,000		552,447	475,720	115,500		$1,000^{(6)}$	1,464,667
Albert H. Taddiken Former Chief Operating Officer*	2009 2008 2007	170,683 252,350 238,341		103,200 339,500 351,049	285,432	23,062 81,200		21,057 <sup>(7)</sup> 2,000 <sup>(8)</sup> 6,418 <sup>(9)</sup>	294,940 616,912 962,440
Jeffrey A. Kupp	2009	238,702		64,500				6,406(10)	309,608
Former Chief Financial Officer**		252,350		257,580		22,139		2,000(11)	534,069
	2007	242,085	10,000(12)	269,509	161,547	81,200		1,000(13)	765,341
Robert S. Kirk	2009	- ,		60,200		122,915(14)		10,400(15)	357,788
Vice President of Worldwide		163,367		205,720	166.502	141,631 <sup>(16)</sup>		10,400 <sup>(17)</sup>	521,118
Sales	2007	156,070		207,466	166,502	163,718 <sup>(18)</sup>		9,400(19)	703,156
Barry F. Koch <sup>(20)</sup>	2009	215,274		51,600			1,077(21)	36,820(22)	304,771
Executive Vice President and Managing Director, Microtune GmbH & Co. KG***	2008	214,115		204,980		17,527	1,960 <sup>(23)</sup>	35,216 <sup>(24)</sup>	473,798
Gilloff & Co. KG	2007	191,114		200,864	142,716	56,000	2,164(25)	74,923(26)	667,781
Justin M. Chapman	2009	164,850		12,685				6,438(27)	183,973
Chief Financial Officer and Former Interim Chief Financial Officer****									
Phillip D. Peterson	2009	210,000		34,400				12,702 <sup>(28)</sup>	257,102
General Counsel									

- \* Mr. Taddiken resigned from the position of Chief Operating Officer effective June 10, 2009.
- \*\* Mr. Kupp resigned from the position of Chief Financial Officer effective December 4, 2009.
- \*\*\* Mr. Koch was promoted to the position of Executive Vice President effective June 10, 2009. Mr. Koch continues to serve as Managing Director of Microtune GmbH & Co. KG.
- \*\*\*\* Mr. Chapman was appointed to the position of Interim Chief Financial Officer effective December 4, 2009. Mr. Chapman was appointed to the position of Chief Financial Officer effective February 11, 2010.
- (1) Amounts reported in the Bonus column do not include cash amounts or RSU awards earned under our 2008 Incentive Compensation Program and 2007 Incentive Compensation Program, which are shown in the Non-Equity Incentive Plan Compensation column with respect to the cash awards paid under the 2008 Incentive Compensation Program and the 2007 Incentive Compensation Program and the Stock Awards column with respect to the RSU awards that vested under the 2008 Incentive Compensation Program and the 2007 Incentive Compensation Program.
- (2) The amounts in this column represent the aggregate grant date fair value of awards computed in accordance with FASB ASC Topic 718.
- (3) The cash payments under the 2008 Incentive Compensation Program are deemed to be earned in 2008 as they were based on the Company s financial performance for the fiscal year ended December 31, 2008, even though participants had to be employed by

Microtune on the payment date in February 2009 to receive the cash payments. The cash payments under the 2007 Incentive Compensation Program are deemed to be earned in 2007 as they were based on the Company s financial performance for the fiscal year ended December 31, 2007, even though participants had to be employed by Microtune on the payment date in February 2008 to receive the cash payments.

- (4) \$8,077 of this amount is attributable to an accrued vacation payout and \$2,000 is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Fontaine.
- (5) \$13,461 of this amount is attributable to an accrued vacation payout and \$2,000 is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Fontaine.
- (6) This amount is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Fontaine.
- (7) \$19,057 of this amount is attributable to an accrued vacation payout and \$2,000 is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Taddiken.
- (8) This amount is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Taddiken.
- (9) \$5,418 of this amount is attributable to an accrued vacation payout and \$1,000 is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Taddiken.
- (10) \$4,406 of this amount is attributable to an accrued vacation payout and \$2,000 is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Kupp.
- (11) This amount is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Kupp.
- (12) This amount is attributable to the portion of a special recognition bonus relating to the previously announced restatement of our consolidated financial statements earned in 2007.
- (13) This amount is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Kupp.
- (14) This amount is attributable to sales commissions.
- (15) \$8,400 of this amount is attributable to an automobile allowance and \$2,000 is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Kirk.
- (16) \$126,871 of this amount is attributable to sales commissions and \$14,760 is attributable to the cash payment paid under the 2008 Incentive Compensation Program.
- (17) \$8,400 of this amount is attributable to an automobile allowance and \$2,000 is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Kirk.
- (18) \$114,718 of this amount is attributable to sales commissions and \$49,000 is attributable to the cash payment paid under the 2007 Incentive Compensation Program.
- (19) \$8,400 of this amount is attributable to an automobile allowance and \$1,000 is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Kirk.
- (20) Mr. Koch is compensated in Euros; the salary and all other compensation reflect the conversion from Euros to U.S. dollars using the average exchange rate in effect for 2009, 2008, and 2007, respectively. The change in pension value and nonqualified deferred compensation earnings reflect the conversion from Euros to U.S. dollars using the exchange rates in effect as of December 31, 2009, 2008, 2007 and 2006.
- (21) This amount is attributable to the change in net present value under the Pensor defined benefit plan from December 31, 2008 to December 31, 2009.
- \$25,542 of this amount is attributable to an automobile allowance, \$10,631 is attributable to our contributions to the defined contribution plan and \$647 is attributable to annual life and accident insurance premiums.
- (23) This amount is attributable to the change in net present value under the Pensor defined benefit plan from December 31, 2007 to December 31, 2008.
- \$24,146 of this amount is attributable to an automobile allowance, \$10,426 is attributable to contributions to the defined contribution plan and \$644 is attributable to annual life and accident insurance premiums.
- (25) This amount is attributable to the change in net present value under the Pensor defined benefit plan from December 31, 2006 to December 31, 2007.
- (26) \$42,737 of this amount is attributable to an accrued vacation payout, \$22,303 is attributable to an automobile allowance, \$9,283 is attributable to contributions to a defined contribution plan and \$600 is attributable to annual life and accident insurance premiums.
- \$4,438 of this amount is attributable to an accrued vacation payout and \$2,000 is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Chapman.
- \$10,702 of this amount is attributable to an accrued vacation payout and \$2,000 is attributable to our contribution to the Microtune, Inc. 401(k) Plan on behalf of Mr. Peterson.

### 2009 Grants of Plan-Based Awards

### **GRANTS OF PLAN-BASED AWARDS IN 2009**

		Under No		e Payouts Incentive		Estimated Possible Payouts Under Equity Incentive Plan Awards			All Other Option Awards: Number of Securities	Exercise or Base Price of Option	Grant Date Fair Value of Stock and Option
Name	Grant Date	Threshold (\$)	Target (\$)	Maximum (\$)	Threshold (#)	Target (#)	Maximum (#)	or Units (#) <sup>(1)</sup>	Underlying Options (#)	Awards (\$/Sh)	Awards (\$)(2)
James A. Fontaine	5/15/09	(1)	(.,	(1)	, ,	Ì	, ,	80,000	• ` ` ′	`` ´	172,000
Albert H. Taddiken	5/15/09							48,000			103,200
Jeffrey A. Kupp	5/15/09							30,000			64,500
Justin M. Chapman	5/15/09							5,900			12,685
Robert S. Kirk	5/15/09							28,000			60,200
Barry F. Koch	5/15/09							24,000			51,600
Phillip D. Peterson	5/15/09							16,000			34,400

- (1) These amounts represent grants of RSU awards pursuant to our 2000 Stock Plan as long-term equity incentive compensation.
- (2) This column represents the aggregate grant date fair value of such RSU awards, excluding any estimates of future forfeitures, computed in accordance with FASB ASC Topic 718. For a discussion of the assumptions used to calculate the value of stock awards, see Note 1 to our Consolidated Financial Statements in the Form 10-K for the year ended December 31, 2009, as filed with the SEC.

### Discussion of Summary Compensation and Plan-Based Awards Tables

Our executive compensation policies and practices, pursuant to which the compensation set forth in the Summary Compensation Table and the Grants of Plan-Based Awards in 2009 table was paid or awarded, are described above under Compensation Discussion and Analysis.

The Company cautions that the amounts reported in the Summary Compensation Table for Stock Awards and Option Awards may not represent the amounts that the named executive officers will actually realize from these equity awards. Whether, and to what extent, a named executive officer realizes value from these awards will depend on whether the shares actually vest, and if the shares do actually vest, the stock price on the date the named executive officer actually disposes of the shares.

The amounts reported in the Summary Compensation Table for Stock Awards and Option Awards are difficult to compare between the named executive officers and also for each executive officer, year over year. This is largely due to the fact that the amounts represent the aggregate grant date fair value of awards in accordance with FASB ASC Topic 718 and not the actual cash value realized, if any, by the executive from equity awards in a given year. The amounts also are affected by the probability that performance conditions, if any, for outstanding equity awards under annual incentive compensation programs will be met. Furthermore, these amounts are not impacted by the current stock price of our common stock. The Summary Compensation Table should be read and analyzed with these factors in mind.

For information regarding the grant date fair value of equity awards made to the named executive officers in 2009, refer to the Grants of Plan-Based Awards in 2009 table above.

For information regarding the value actually received or earned by the named executive officers in 2009, refer to the Compensation Discussion and Analysis on beginning on page 44.

Additional information on all outstanding equity awards is reflected in the Outstanding Equity Awards as of December 31, 2009 table on beginning on page 60. The Company has granted stock options and RSU awards under the 2000 Stock Plan.

## 2009 Outstanding Equity Awards at Fiscal Year-End

The following table sets forth information concerning outstanding equity awards for our named executive officers at December 31, 2009. Option awards were granted for ten-year terms, ending on the option expiration date set forth in the table. Stock awards were granted as indicated in the footnotes to the table.

## Outstanding Equity Awards as of December 31, 2009

		Option Awards				Stock Awards			Eauite
Name	Number of Securities Underlying Unexercised Options (#) Exercisable	Number of Securities Underlying Unexercised Options (#) Unexercisable	Equity Incentive Plan Awards; Number of Securities Underlying Unexercised Unearned Options (#)	Option Exercise Price (\$)(1)	Option Expiration Date	Number of Shares or Units of Stock That Have Not Vested (#)	Market Value of Shares or Units of Stock That Have Not Vested (\$)(2)	Equity Incentive Plan Awards: Number of Unearned Shares, Units or Other Rights That Have Not Vested (#)	Equity Incentive Plan Awards: Market or Payout Value of Unearned Shares, Units or Other Rights That Have Not Vested (\$)
James A. Fontaine	740,000 <sup>(6)</sup> 27,163 <sup>(7)</sup> 66,668 <sup>(8)</sup> 424,511 <sup>(9)</sup>	200,000(3)(10)		2.70 3.20 4.99 3.84 4.53	8/12/2013 6/4/2014 8/19/2014 5/25/2015 5/10/2017	240,000(4)(11)	542,400		
Albert H. Taddiken							·		
Justin M. Chapman	7,334 667 1,333 3,000 5,000 4,000 1,250 34,680 37,595 3,924	15,000 <sup>(3)</sup>		1.79 2.20 2.75 2.40 2.50 3.95 5.48 5.31 3.84 4.53	5/1/2013 7/9/2013 7/9/2013 8/1/2013 3/31/2014 6/14/2014 8/19/2014 1/18/2015 5/25/2015 5/10/2017	27,900(5)	63,054		
Jeffrey A. Kupp	332,500 19,167			3.61 4.53	5/9/2015 5/10/2017				
Robert S. Kirk	8,575 15,925 1,000 2,000 20,333 37,079 89,578 16,516 4,000 60,000 165,666	70,000 <sup>(3)</sup>		1.27 2.21 2.20 2.75 2.40 2.85 2.80 3.20 4.47 4.99 3.84 4.53	3/10/2013 3/10/2013 7/9/2013 7/9/2013 8/1/2013 8/1/2013 10/21/2013 6/4/2014 8/19/2014 8/19/2014 5/25/2015 5/10/2017				

					84,000(4)	189,840
Barry F. Koch	14,667 161,462 43,195 16,337 25,000 134,741	60,000(3)	4.95 2.44 2.40 3.20 4.47 3.84 4.53	5/11/2010 4/14/2013 8/1/2013 6/4/2014 8/19/2014 5/25/2015 5/10/2017		
					72,000(4)	162,720
Phillip D. Peterson	110,000 6,248 44,608 30,001	40,000 <sup>(3)</sup>	2.71 3.20 3.84 4.53	4/8/2014 6/4/2014 5/25/2015 5/10/2017	48,000(4)	108,480

- (1) The 2000 Stock Plan provides that the option exercise price per share for incentive stock options shall be no less than 100 percent of the closing price per share of our common stock on the date of grant.
- (2) The values shown in this column are based on our December 31, 2009 closing stock price of \$2.26 per share.
- (3) This amount includes two grants of equal size, one of which will vest on May 11, 2010 and the other on May 11, 2011.
- (4) This amount includes four grants which vest according to the following schedule: 16.66% will vest on May 11, 2010, 16.66% will vest on May 11, 2011, 33.33% will vest on May 12, 2012 and 33.33% will vest on May 15, 2013.
- (5) This amount includes four grants which vest according to the following schedule: 3,000 shares will vest on each of May 11, 2010 and May 11, 2011, 16,000 shares will vest on May 12, 2012 and 5,900 shares will vest on May 15, 2013.
- (6) This amount includes 370,001 option shares transferred in a divorce settlement.
- (7) This amount includes 13,582 option shares transferred in a divorce settlement.
- (8) This amount includes 33,334 option shares transferred in a divorce settlement.
- (9) This amount includes 165,098 option shares transferred in a divorce settlement.
- (10) This amount includes 49,144 option shares transferred in a divorce settlement.
- (11) This amount includes 26,448 shares transferred in a divorce settlement.

#### 2009 Option Exercises and Stock Vested

The following table sets forth information concerning stock options exercised and stock awards vested during 2009 for our named executive officers:

#### Option Exercises and Stock Awards Vested in 2009

	Option .	Awards	Stock Awards			
Name	Number of Shares Acquired on Exercise(#)	Value Realized on Exercise(\$)	Number of Shares Acquired on Vesting(#) <sup>(1)</sup>	Value Realized on Vesting(\$)(2)		
James A. Fontaine			8,071	15,416		
Albert H. Taddiken			5,765	11,011		
Jeffrey A. Kupp			5,534	10,570		
Justin M. Chapman			2,998	5,726		
Robert S. Kirk			3,689	7,046		
Barry F. Koch			4,381	8,368		
Phillip D. Peterson			3,689	7,046		

- (1) The amounts in this column reflect RSU awards vested under the 2008 Incentive Compensation Program on February 3, 2009.
- (2) The value realized upon vesting was the number of shares vesting multiplied by our closing stock price on the vest date which was \$1.91.

## 2009 Nonqualified Deferred Compensation

The following table sets forth information concerning nonqualified defined contribution and deferred compensation plans for our named executive officers during 2009:

Nonqualified Deferred Compensation in 2009

Name	Executive Contributions in Last FY(\$)	Registrant Contributions in Last FY(\$)	Aggregate Earnings in Last FY(\$) <sup>(1)</sup>	Aggregate Withdrawals/ Distributions(\$)	Aggregate Balance at Last FYE(\$)
James A. Fontaine					
Albert H. Taddiken					
Jeffrey A. Kupp					
Justin M. Chapman					
Robert S. Kirk					
Barry F. Koch		10,631(2)	$1,077^{(3)}$		70,032(4)
Phillip D. Peterson					

(1) Aggregate earnings are included in the Summary Compensation Table under Change in Pension Value and Nonqualified Deferred Compensation Earnings.

- (2) Microtune made contributions under the Allianz Unterstützungskasse defined contribution plan of 7,623 Euros. This amount was converted to \$10,631, using a 1.39463 exchange rate, which was the average rate for 2009. Microtune did not make contributions to the Pensor defined benefit plan in 2009.
- (3) The change in net present value under the Pensor defined benefit plan from December 31, 2008 to December 31, 2009.
- (4) The aggregate balance of the Allianz Unterstützungskasse defined contribution plan and the Pensor defined benefit plan at December 31, 2009.

Under the Pensor defined benefit plan and the current Allianz Unterstützungskasse defined contribution plan, payouts are made on Mr. Koch s 65th birthday. If he elects to retire at an earlier age, deductions to the payout amount are made. Mr. Koch does not have discretion to select investments under the Allianz Unterstützungskasse defined contribution plan. There is a guaranteed rate of return of 3.25% with additional earnings not guaranteed. We no longer make contributions to the Pensor defined benefit plan.

#### **Compensation of Directors**

#### **Director Compensation in 2009**

					Change in		
	Fees				Pension Value and		
	Earned or Paid in Cash	Stock Awards	Option Awards	Non-Equity Incentive Plan Compensation	Nonqualified Deferred Compensation	All Other Compensation	Total
Name	(\$)	(\$)(1)	(\$)	(\$)	Earnings (\$)	(\$)	(\$)
Walter S. Ciciora <sup>(2)</sup>	20,000	16,380					36,380
James H. Clardy <sup>(3)</sup>	25,000	16,380					41,380
Steven Craddock <sup>(4)</sup>	20,000	16,380					36,380
Anthony J. LeVecchio <sup>(5)</sup>	30,000	16,380					46,380
Bernard T. Marren <sup>(6)</sup>	18,000	16,380					34,380
Michael T. Schueppert <sup>(7)</sup>	20,000	16,380					36,380
William P. Tai <sup>(8)</sup>	15,000	16,380					31,380
A. Travis White <sup>(9)</sup>	30,000	16,380					46,380

- (1) Each non-employee director received 8,400 stock awards for shares of common stock on April 24, 2009. This column represents the aggregate grant date fair value of such stock awards, excluding any estimates of future forfeitures, computed in accordance with FASB ASC Topic 718. For a discussion of the assumptions used to calculate the value of stock awards, see Note 1 to our Consolidated Financial Statements in the Form 10-K for the year ended December 31, 2009, as filed with the SEC.
- (2) Mr. Ciciora held 140,938 options and 8,400 stock awards for shares of common stock as of December 31, 2009.
- (3) Mr. Clardy held 140,938 options and 8,400 stock awards for shares of common stock as of December 31, 2009.
- (4) Mr. Craddock held 145,938 options and 8,400 stock awards for shares of common stock as of December 31, 2009.
- (5) Mr. LeVecchio held 134,688 options and 8,400 stock awards for shares of common stock as of December 31, 2009.
- (6) Mr. Marren held 87,000 options and 8,400 stock awards for shares of common stock as of December 31, 2009.
- (7) Mr. Schueppert held 63,000 options and 8,400 stock awards for shares of common stock as of December 31, 2009.
- (8) Mr. Tai held 140,938 options and 8,400 stock awards for shares of common stock as of December 31, 2009.
- (9) Mr. White held 135,938 options and 8,400 stock awards for shares of common stock as of December 31, 2009.

Our outside directors participate in our 2000 Director Stock Plan. The 2000 Director Stock Plan, as adopted and approved by our stockholders, became effective as of August 4, 2000, and was amended effective as of April 17, 2002, February 27, 2004, April 13, 2005, March 27, 2007, March 11, 2008, and March 10, 2009. Under the 2000 Director Stock Plan, restricted stock units may be granted only to non-employee directors. A non-employee director shall receive a restricted stock unit award of 5,250 shares of common stock on the date he or she first becomes a non-employee director. As currently in effect, each non-employee director will also receive a restricted stock units award of 8,400 shares of common stock on the date of our annual stockholders meeting of each year, provided that he or she is a non-employee director on such date and he or she has served as director for at least the preceding six months. All grants under the 2000 Director Stock Plan are automatic and nondiscretionary. The aggregate number of shares of common stock that may be issued under the plan shall not exceed 1,187,500. Our Board of Directors evaluates director compensation on an annual basis. As part of such evaluation, our Board of Directors reviews board compensation practices of our peer companies. The 2000 Director Stock Plan shall continue until August 4, 2010, unless terminated earlier by the Board of Directors. However, following receipt of stockholder approval of the 2010 Director Stock Plan at the 2010 Annual Meeting, we intend to terminate the 2000 Director Stock Plan.

In October 2008, the Board determined to reduce its cash compensation effective January 1, 2009, and to convert its annual stock option awards

to annual awards of a lesser number of RSU awards. These changes to director compensation are outlined in greater detail below: A transition from stock options to RSUs: 5,250 RSUs awarded to directors upon initial appointment, with such awards vesting over three years; and 8,400 RSUs awarded to directors annually, with such awards also vesting over three years. Such changes were effected through an amendment to the 2000 Director Stock Plan as approved by the stockholders of the Company at the 2009 Annual Meeting of Stockholders. Fees for meeting attendance: \$4,000 per regular board meeting for in person attendance (5 regular meetings); \$2,000 per regular board meeting for telephonic attendance; \$1,000 per regular board meeting if no attendance; \$10,000 per year for Audit Committee Chair; \$10,000 per year for Lead Independent Director;

\$5,000 per year for Compensation Committee Chair;

\$0 per year for Nominating and Corporate Governance Committee Chair;

\$0 per year for Technical Advisory Committee Chair, but \$1,500 per meeting attended; and

No changes were made to the compensation paid for excess meetings, however, our Lead Independent Director and Compensation Committee Chair can make a determination that any special Board or committee meeting should qualify as either a regular or as an excess meeting for compensation purposes based on the time and effort required to prepare for the meeting and the duration of the meeting.

### CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS, AND DIRECTOR INDEPENDENCE

Except as set forth below, we have not been a party to any transaction that involved more than \$120,000 with an executive officer, a nominee for director, a director or a 5% stockholder or any of their immediate family members since the beginning of fiscal 2009 or that involved indebtedness to or payments from us during fiscal

2009 other than as described under the caption Executive Compensation and the transactions described below. The discussion does not address compensation paid in connection with employment or Board service.

Drew Peck, a nominee for director of our Company, is a founder of Steel Horse Advisors, a strategic advisory and consulting firm and a division of Green Manning & Bunch, Ltd., and is an employee of Green, Manning & Bunch. The Company engaged Steel Horse Advisors in January 2010 as a strategic consultant to assist the Company in various corporate and business development activities. Pursuant to this engagement the Company has paid Steel Horse Advisors consulting fees of approximately \$50,000 in the aggregate. In addition, the Company has agreed that it will pay Steel Horse Advisors customary advisory fees in connection with transactions arising out of its engagement and will reimburse Steel Horse Advisors for its out-of-pocket expenses incurred in connection with providing such consulting and advisory services. The amount of fees that may become payable to Steel Horse Advisors in the future, if any, is not currently determinable but could be material to Steel Horse Advisors.

We believe that all transactions between us and our officers, nominees for director, directors, principal stockholders and other affiliates have been on terms no less favorable to us than could be obtained from unaffiliated third parties.

The Audit Committee of our Board reviews all related party transactions for potential conflict of interest situations on an ongoing basis and approves any such transactions. However, any such transaction may be alternatively approved by another independent body of the Board in accordance with NASDAQ Listing Rule 5630(a).

Our Board of Directors has determined that eight of our current Board members, Messrs. Ciciora, Clardy, Craddock, LeVecchio, Marren, Schueppert, Tai and White, and that director nominees Messrs. Rast and Rau, are independent directors as defined under the current listing standards of The NASDAQ Stock Market. For more information about our corporate governance, including information regarding our board of directors and committees, see Microtune Corporate Governance beginning on page 8.

#### **Compensation Committee Interlocks and Insider Participation**

Our Compensation Committee currently consists of Messrs. Clardy (Chairperson), Marren, and Schueppert. None of the members of the Compensation Committee were officers or employees of Microtune at any time during 2009 or at any other time. During 2009, no current executive officer of Microtune served as a member of the Board of Directors or compensation committee of any other entity whose executive officer(s) served on Microtune s Board of Directors or Compensation Committee.

#### REPORT OF THE AUDIT COMMITTEE OF THE BOARD OF DIRECTORS

We, Anthony J. LeVecchio, Steven Craddock and A. Travis White, are the current members of the Audit Committee and each of us is a non-employee director. The Board of Directors has determined that Mr. LeVecchio is an audit committee financial expert and that each of the committee members is independent as defined in NASDAQ Listing Rule 5605(a)(2) and meets the independence requirements of Rule 10A-3(b)(i) of the Exchange Act, as well as the requirements of NASDAQ Listing Rule 5605(c)(2).

The Audit Committee discusses general financial and accounting-related matters at its regular quarterly meetings that coincide with full Board meetings, and it discusses Microtune s quarterly financial performance and associated earnings announcements at such meetings. The committee met seven times during fiscal year 2009.

We assist the Board of Directors in its oversight of Microtune s financial accounting, reporting and controls. We operate under a written charter that both the Board of Directors and we have approved. We also evaluate the performance and independence of Microtune s independent auditor.

Management is responsible for the preparation, presentation and integrity of Microtune s financial statements, including setting the accounting and financial reporting principles and establishing the internal controls and procedures designed to ensure compliance with accounting standards and applicable laws and regulations. Our independent auditor is responsible for performing an independent audit of the consolidated financial statements in accordance with generally accepted auditing standards and issuing a report on the consolidated financial statements. We oversee these processes. As part of that oversight, we require that the independent auditor report directly to the Audit Committee.

We reviewed and discussed the audited financial statements for the fiscal year ended December 31, 2009 with management and the independent auditor. We also discussed with the independent auditor the matters required to be discussed by Statement on Auditing Standards 61, Communication with Audit Committees, as amended or supplemented. We received the written disclosures and the letter from the independent auditor required by applicable requirements of the Public Company Accounting Oversight Board for independent auditor communications with Audit Committees concerning independence. We discussed with KPMG LLP that firm s independence and considered whether the provision of non-audit services by the independent auditor was compatible with maintaining the auditor s independence. We approve any non-audit services on a case-by-case basis before any work is performed. Based on the reports, discussions and review described in this report, and subject to the limitations on our role and responsibilities, we recommended to the Board of Directors that the audited financial statements be included in Microtune s Annual Report on Form 10-K for fiscal year 2009, and the Board of Directors approved such inclusion.

AUDIT COMMITTEE OF THE BOARD OF DIRECTORS

Anthony J. LeVecchio, Chairperson Steven Craddock A. Travis White

#### OTHER MATTERS

Form 10-K for 2009

On February 16, 2010, we filed with the SEC an Annual Report on Form 10-K for the year ended December 31, 2009. The Annual Report on Form 10-K has been provided concurrently with this Proxy Statement to all stockholders entitled to notice of, and to vote at, the Annual Meeting.

Stockholders may also obtain a copy of the Annual Report on Form 10-K and any of our other SEC reports, free of charge, (1) from the SEC s website at www.sec.gov, (2) from our website at www.microtune.com, or (3) by writing to Investor Relations, Microtune, Inc., 2201 10<sup>th</sup> Street, Plano, Texas 75074. The Annual Report on Form 10-K is not incorporated into this Proxy Statement and is not considered proxy solicitation material. Information contained on our website, other than this Proxy Statement, is not part of the proxy solicitation material and is not incorporated by reference herein.

#### **Annual Meeting**

Our Board of Directors does not currently intend to bring any other business before our Annual Meeting and is not aware of any other business to be brought before our Annual Meeting. If any other business is properly brought before our Annual Meeting, including consideration of a motion to adjourn or postpone our Annual Meeting in order to, among other things, solicit additional proxies, the proxy holders will vote the proxies based on their judgment.

Any stockholder or stockholder is representative who, because of a disability, may need special assistance or accommodation to allow him or her to participate at our Annual Meeting may request reasonable assistance or accommodation from us by contacting Barbara Ureste at Microtune, Inc., 2201 10<sup>th</sup> Street, Plano, Texas 75074. The telephone number at that location is (972) 673-1600. To provide us sufficient time to arrange for reasonable assistance or accommodation, please submit requests by May 13, 2010.

It is important that your shares be represented at the meeting, regardless of the number of shares that you hold. You are, therefore, urged to complete and return the accompanying proxy card in the envelope that has been enclosed, at your earliest convenience.

FOR THE BOARD OF DIRECTORS

DATED: April 9, 2010

#### APPENDIX A

#### MICROTUNE, INC.

#### 2010 STOCK PLAN

(Effective as of May 20, 2010)

Microtune, Inc. (the Company ), a Delaware corporation, hereby establishes and adopts the following Microtune, Inc. 2010 Stock Plan (the Plan ).

#### 1. PURPOSE OF THE PLAN

The purpose of the Plan is to assist the Company and its Subsidiaries in attracting and retaining selected individuals to serve as employees, directors and consultants of the Company and its Subsidiaries who are expected to contribute to the Company success and to achieve long-term objectives which will inure to the benefit of all stockholders of the Company through the additional incentives inherent in the Awards hereunder.

#### 2. **DEFINITIONS**

- **2.1.** Affiliate shall have the meaning set forth in the last paragraph of Section 11.3.
- **2.2.** Award means any Option, Stock Appreciation Right, Restricted Stock Award, Restricted Stock Unit Award, Other Share-Based Award, Performance Award or any other right, interest or option relating to Shares or cash granted pursuant to the provisions of the Plan.
- **2.3.** Award Agreement means any agreement, contract or other instrument or document evidencing any Award hereunder, including through an electronic medium.
- **2.4.** Board means the board of directors of the Company.
- **2.5.** Cause shall have the same meaning as set forth in an Award Agreement, if applicable.
- **2.6.** Change in Control shall have the same meaning as set forth in Section 11.3.
- 2.7. Code means the Internal Revenue Code of 1986, as amended.

<b>2.8.</b> <i>Committee</i> means the Compensation Committee of the Board or a subcommittee thereof formed by the Compensation Committee to act as
the Committee hereunder. The Committee shall consist of (i) independent directors (i.e., Directors who are neither officers nor Employees of the
Company or its Subsidiaries, nor have a relationship which, in the judgment of the Board, would interfere with the exercise of independent
judgment in carrying out their responsibilities, and who are otherwise independent under the rules of NASDAQ), (ii) non-employee directors
within the meaning of Rule 16b-3 promulgated under the Exchange Act, and (iii) outside directors within the meaning of Code Section 162(m).

- **2.9.** Company shall have the same meaning as set forth in the introductory paragraph of this document.
- 2.10. Consultant means any person, including an advisor, engaged by the Company or any Subsidiary to render services to such entity.
- **2.11.** Covered Employee means an employee of the Company or its Subsidiaries who is a covered employee within the meaning of Code Section 162(m).
- **2.12.** Covered Family Members shall have the same meaning as set forth in Section 12.3.

<b>2.13.</b> <i>Director</i> means a member of the Board.
<b>2.14.</b> <i>Dividend Equivalents</i> shall have the meaning set forth in Section 12.5.
<b>2.15.</b> <i>Employee</i> means any employee of the Company or any Subsidiary, including officers and employee Directors, and any prospective employee conditioned upon, and effective not earlier than, such person becoming an employee of the Company or any Subsidiary.
<b>2.16.</b> Exchange Act means the Securities Exchange Act of 1934, as amended.
<b>2.17.</b> Executive Officer means an officer of the Company within the meaning of the rules under Section 16 of the Exchange Act.
<b>2.18.</b> Fair Market Value means the per Share closing price of the Shares as reported on The NASDAQ Global Market or The NASDAQ Capit Market of The NASDAQ Stock Market as of the relevant date (or if there were no reported prices on such date, on the last preceding date on which the prices were reported) or, if the Company is not then listed on The NASDAQ Stock Market, on such other principal securities exchange on which the Shares are traded, and if the Company is not listed on The NASDAQ Stock Market or any other securities exchange, the Fair Market Value of Shares shall be determined by the Committee in its sole discretion.
<b>2.19.</b> <i>Good Reason</i> shall have the meaning as set forth in an Award Agreement, if applicable.
<b>2.20.</b> <i>Limitations</i> shall have the meaning set forth in Section 10.6.
<b>2.21.</b> <i>Option</i> means any right granted to a Participant under the Plan allowing such Participant to purchase Shares at such price or prices and during such period or periods as the Committee shall determine and not inconsistent with the provisions of the Plan.
<b>2.22.</b> Other Share-Based Award shall have the meaning set forth in Section 8.1.
<b>2.23.</b> <i>Participant</i> means an Employee, Director or Consultant who is selected by the Committee to receive an Award under the Plan.
<b>2.24.</b> Payee shall have the meaning set forth in Section 13.2.
<b>2.25.</b> <i>Performance Award</i> means any Award of Performance Cash or Performance Share Units granted pursuant to Article 9.

- **2.26.** *Performance Cash* means any cash incentives granted pursuant to Article 9 which will be paid to the Participant upon the achievement of such performance goals as the Committee shall establish.
- **2.27.** *Performance Period* means the period established by the Committee during which any performance goals specified by the Committee with respect to such Award are to be measured.
- **2.28.** *Performance Share Unit* means any grant pursuant to Article 9 of a unit valued by reference to a designated number of Shares, which value will be paid to the Participant upon achievement of such performance goals as the Committee shall establish.
- **2.29.** *Permitted Assignee* shall have the meaning set forth in Section 12.3.
- **2.30.** *Plan* shall have the same meaning as set forth in the introductory paragraph of this document.
- **2.31.** *Plan Administrator* means the individual or committee appointed by the Committee to handle the day-to-day administration of the Plan. If the Committee does not appoint an individual or committee to serve as the Plan Administrator, the Committee will be the Plan Administrator.

<b>2.32.</b> <i>Protection Period</i> shall have the meaning as set forth in an Award Agreement, if applicable.
<b>2.33.</b> <i>Qualifying Termination</i> shall have the meaning as set forth in an Award Agreement, if applicable.
<b>2.34.</b> Restricted Stock means any Share issued with the restriction that the holder may not sell, transfer, pledge or assign such Share and with such other restrictions as the Committee, in its sole discretion, may impose (including any restriction on the right to vote such Share and the right to receive any dividends), which restrictions may lapse separately or in combination at such time or times, in installments or otherwise, as the Committee may deem appropriate.
<b>2.35.</b> Restricted Stock Award shall have the meaning set forth in Section 7.1.
<b>2.36.</b> Restricted Stock Unit means an Award that is valued by reference to a Share, which value may be paid to the Participant by delivery, as the Committee shall determine, of cash, Shares, or any combination thereof, and that has such restrictions as the Committee, in its sole discretion, may impose, including without limitation, any restriction on the right to retain such Awards, to sell, transfer, pledge or assign such Awards, and/or to receive any cash Dividend Equivalents with respect to such Awards, which restrictions may lapse separately or in combination at such time or times, in installments or otherwise, as the Committee may deem appropriate.
<b>2.37.</b> Restricted Stock Unit Award shall have the meaning set forth in Section 7.1.
<b>2.38.</b> <i>Shares</i> means the shares of common stock of the Company.
<b>2.39.</b> <i>Stock Appreciation Right</i> means the right granted to a Participant pursuant to Article 6.
<b>2.40.</b> <i>Subsidiary</i> means any corporation (other than the Company) in an unbroken chain of corporations beginning with the Company if, at the relevant time each of the corporations other than the last corporation in the unbroken chain owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in the chain.
<b>2.41.</b> Substitute Awards means Awards granted or Shares issued by the Company in assumption of, or in substitution or exchange for, awards previously granted, or the right or obligation to make future awards, in each case by a company acquired by the Company or any Subsidiary or with which the Company or any Subsidiary combines.
<b>2.42.</b> <i>Vesting Period</i> shall have the meaning set forth in Section 7.1.

SHARES SUBJECT TO THE PLAN

3.1. Number of Shares.

(a) Subject to the adjustment provided for in Section 12.2, a total of 3,000,000 Shares shall be authorized for grant under the Plan. Any Shares that are subject to Awards shall be counted against this limit as one (1) Share for every one (1) Share granted.

(b) If (i) any Shares subject to an Award are forfeited, cancelled or expire or (ii) an Award is settled for cash (in whole or in part), the Shares subject to such Award shall, to the extent of such forfeiture, cancellation, expiration or cash settlement, again be available for Awards under the Plan. Notwithstanding anything to the contrary contained herein, the following Shares shall be added back to the Shares authorized for grant under the Plan: (A) Shares tendered by the Participant or withheld by the Company in payment of the purchase price of an Option, (B) Shares tendered by the Participant or withheld by the Company to satisfy any tax withholding obligation (or portion thereof) with respect to an Award, and (C) Shares subject to a Stock Appreciation Right that are not issued in connection with the stock settlement of the Stock Appreciation

Right on exercise thereof. Any Shares that again become available for grant pursuant to this Article shall be authorized for grant as one (1) Share for any one (1) Share that may be added back pursuant to this Section 3.1(b).

- (c) Substitute Awards shall not reduce the Shares authorized for grant under the Plan or authorized for grant to a Participant under Section 10.6. Additionally, in the event that a company acquired by the Company or any Subsidiary or with which the Company or any Subsidiary combines has shares available under a pre-existing plan approved by stockholders and not adopted in contemplation of such acquisition or combination, the shares available for grant pursuant to the terms of such pre-existing plan (as adjusted, to the extent appropriate, using the exchange ratio or other adjustment or valuation ratio or formula used in such acquisition or combination to determine the consideration payable to the holders of common stock of the entities party to such acquisition or combination) may be used for Awards under the Plan and shall not reduce the Shares authorized for grant under the Plan; provided, that Awards using such available shares shall not be made after the date awards or grants could have been made under the terms of the pre-existing plan, absent the acquisition or combination, and shall only be made to individuals who were not Employees, Directors or Consultants prior to such acquisition or combination.
- (d) No Award may be granted if the number of Shares to be delivered in connection with such Award exceeds the number of Shares remaining available under this Plan minus the number of Shares issuable in settlement of or related to then-outstanding Awards. The Committee may adopt reasonable counting procedures to ensure appropriate counting, avoid double counting and make adjustments if the number of Shares actually delivered differs from the number of Shares previously counted in connection with an Award.
- **3.2.** Character of Shares. Any Shares issued hereunder may consist, in whole or in part, of authorized and unissued shares, treasury shares or shares purchased in the open market or otherwise.

#### 4. ELIGIBILITY AND ADMINISTRATION

- 4.1. Eligibility. Any Employee, Director or Consultant shall be eligible to be selected by the Committee as a Participant.
- 4.2. Administration
- (a) The Plan shall be administered by the Committee. The Committee shall have full power and authority, subject to the provisions of the Plan and subject to such orders or resolutions not inconsistent with the provisions of the Plan as may from time to time be adopted by the Board, to: (i) select the Employees, Directors and Consultants to whom Awards may from time to time be granted hereunder; (ii) determine the type or types of Awards, not inconsistent with the provisions of the Plan, to be granted to each Participant hereunder; (iii) determine the number of Shares to be covered by each Award granted hereunder; (iv) determine the terms and conditions, not inconsistent with the provisions of the Plan, of any Award granted hereunder; (v) determine whether, to what extent and under what circumstances, Awards may be settled in cash, Shares or other property; (vi) determine whether, to what extent, and under what circumstances cash, Shares, other property and other amounts payable with respect to an Award made under the Plan shall be deferred either automatically or at the election of the Participant; (vii) determine whether, to what extent and under what circumstances any Award shall be canceled or suspended; (viii) interpret and administer the Plan and any instrument or agreement entered into under or in connection with the Plan, including any Award Agreement; (ix) correct any defect, supply any omission or reconcile any inconsistency in the Plan or any Award Agreement in the manner and to the extent that the Committee shall deem desirable to carry it into effect; (x) establish such rules and regulations and appoint such agents as it shall deem appropriate for the proper administration of the Plan; (xi) determine whether any Award will have Dividend Equivalents and the time and form of payment of such Dividend Equivalents; and (xii) make any other determination and take any other action that the Committee deems necessary or desirable for administration of the Plan.

-4-

Notwithstanding the foregoing, to the extent, and during such periods that, the Committee is unavailable or otherwise unable to satisfy its responsibilities set forth in this Section 4.2(a) (or any other provision under the Plan), the Board or its delegate shall assume such responsibilities until such time as the Committee again becomes available to resume satisfying such responsibilities.

- (b) Decisions of the Committee shall be final, conclusive and binding on all persons or entities, including the Company, any Participant, and any Subsidiary. A majority of the members of the Committee may determine its actions, including fixing the time and place of its meetings. Notwithstanding the foregoing, the determination of the Directors to whom Awards may be granted, the time(s) at which Awards may be granted to Directors and the number of Shares subject to Awards to Directors shall be made by the Board.
- (c) To the extent not inconsistent with applicable law, including Code Section 162(m), or the rules and regulations of The NASDAQ Stock Market (or such other principal securities exchange on which the Shares are traded), the Committee may delegate to one or more Executive Officers or a committee of Executive Officers the right to grant Awards to Employees (who are not Directors or Executive Officers) and Consultants, the authority to take action on behalf of the Committee pursuant to the Plan to cancel or suspend Awards to Employees (who are not Directors or Executive Officers) and Consultants, and the authority to take any of the other actions described in Section 4.2(a).
- (d) The Committee may appoint the Plan Administrator, who will have the responsibility and duty to administer the Plan on a daily basis. The Committee may remove the Plan Administrator with or without cause at any time. The Plan Administrator will have all the day-to-day responsibilities of administering the Plan but for those duties retained by the Committee as set forth above in Section 4.2(c) and not otherwise delegated to such Plan Administrator.

Notwithstanding the foregoing, to the extent, and during such periods that, the Plan Administrator is unavailable or otherwise unable to satisfy its responsibilities set forth in this Section 4.2(d) to the Plan (or any other provision under the Plan), the Committee or its delegate shall assume such responsibilities until such time as the Plan Administrator again becomes available to resume satisfying such responsibilities.

#### 5. OPTIONS

- **5.1.** *Grant of Options*. Options may be granted hereunder to Participants either alone or in addition to other Awards granted under the Plan. Any Option shall be subject to the terms and conditions of this Article and to such additional terms and conditions, not inconsistent with the provisions of the Plan, as the Committee shall deem desirable.
- **5.2.** Award Agreements. All Options granted pursuant to this Article shall be evidenced by a written Award Agreement in such form and containing such terms and conditions as the Committee shall determine which are not inconsistent with the provisions of the Plan. Such Award Agreement shall be exempt from the requirements of Code Section 409A. The terms of Options need not be the same with respect to each Participant. Granting an Option pursuant to the Plan shall impose no obligation on the recipient to exercise such Option. Any individual who is granted an Option pursuant to this Article may hold more than one (1) Option granted pursuant to the Plan at the same time.
- **5.3.** Option Price. Other than in connection with Substitute Awards, the option price per each Share purchasable under any Option granted pursuant to this Article shall not be less than 100% of the Fair Market Value of one (1) Share on the date of grant of such Option. Other than pursuant to Section 12.2, the Committee shall not without the approval of the Company s stockholders (a) lower the option price per Share of an Option after it is granted, (b) cancel an Option in exchange for cash or another Award (other than in connection with Substitute Awards), or (c) take any other action with respect to an Option that would be treated as a repricing under the rules and regulations of the principal securities exchange on which the Shares are traded.

**5.4.** *Option Term.* The term of each Option shall be fixed by the Committee in its sole discretion; provided, that no Option shall be exercisable after the expiration of ten (10) years from the date the Option is granted, except in the event of death or disability.

#### **5.5.** Exercise of Options.

- (a) Vested Options granted under the Plan shall be exercised by the Participant or by a Permitted Assignee thereof (or by the Participant s executors, administrators, guardian or legal representative, as may be provided in an Award Agreement) as to all or part of the Shares covered thereby, by giving notice of exercise to the Company or its designated agent, specifying the number of Shares to be purchased. The notice of exercise shall be in such form, made in such manner, and in compliance with such other requirements consistent with the provisions of the Plan as the Committee may prescribe from time to time.
- (b) Unless otherwise provided in an Award Agreement, full payment of such purchase price shall be made at the time of exercise and shall be made (i) in cash or cash equivalents (including certified check or bank check or wire transfer of immediately available funds), (ii) by tendering previously acquired Shares (either actually or by attestation, valued at their then Fair Market Value), (iii) with the consent of the Committee, by delivery of other consideration (including, where permitted by law and the Committee, other Awards) having a Fair Market Value on the exercise date equal to the total purchase price, (iv) with the consent of the Committee, by withholding Shares otherwise issuable in connection with the exercise of the Option, (v) through any other method specified in an Award Agreement (including same-day sales through a broker except by Executive Officers), or (vi) any combination of any of the foregoing. The notice of exercise, accompanied by such payment, shall be delivered to the Company at its principal business office or such other office as the Committee may from time to time direct, and shall be in such form, containing such further provisions consistent with the provisions of the Plan, as the Committee may from time to time prescribe. In no event may any Option granted hereunder be exercised for a fraction of a Share. No adjustment shall be made for cash dividends or other rights for which the record date is prior to the date of such issuance.
- **5.6.** Form of Settlement. In its sole discretion, the Committee may provide that the Shares to be issued upon an Option s exercise shall be in the form of Restricted Stock or other similar securities.
- **5.7.** *Incentive Stock Options.* The Committee may grant Options intended to qualify as incentive stock options as defined in Code Section 422, to any employee of the Company or any Subsidiary, subject to the requirements of Code Section 422. Solely for purposes of determining whether Shares are available for the grant of incentive stock options under the Plan, the maximum aggregate number of Shares that may be issued pursuant to incentive stock options granted under the Plan shall be the number of Shares set forth in the first sentence of Section 3.1(a), subject to adjustments provided for in Section 12.2. Incentive stock options shall not be granted more than ten (10) years after the earlier of the adoption of this Plan or the approval of this Plan by the Company s stockholders. In addition, the Fair Market Value of Shares subject to an incentive stock option and the aggregate Fair Market Value of Shares of any parent corporation or subsidiary corporation (within the meaning of Code Sections 424(e) and (f)) subject to any other incentive stock option (within the meaning of Code Sections 424(e) and (f)) that first becomes purchasable by a Participant in any calendar year may not (with respect to that Participant) exceed \$100,000, or such other amount as may be prescribed under Code Section 422 or applicable regulations or rulings from time to time. As used in the previous sentence, Fair Market Value shall be determined as of the date the incentive stock options are granted. Failure to comply with this provision shall not impair the enforceability or exercisability of any Option, but shall cause the excess amount of Shares to be reclassified in accordance with the Code.

#### 6. STOCK APPRECIATION RIGHTS

**6.1.** Grant and Exercise. The Committee may provide Stock Appreciation Rights (a) in conjunction with all or part of any Option granted under the Plan or at any subsequent time during the term of such Option, (b) in

-6-

conjunction with all or part of any Award (other than an Option) granted under the Plan or at any subsequent time during the term of such Award, or (c) without regard to any Option or other Award in each case upon such terms and conditions as the Committee may establish in its sole discretion.

- **6.2.** *Terms and Conditions*. Stock Appreciation Rights shall be subject to such terms and conditions, not inconsistent with the provisions of the Plan, as shall be determined from time to time by the Committee, including the following:
- (a) Upon the exercise of a Stock Appreciation Right, the holder shall have the right to receive the excess of (i) the Fair Market Value of one (1) Share on the date of exercise (or such amount less than such Fair Market Value as the Committee shall so determine at any time during a specified period before the date of exercise) over (ii) the grant price of the Stock Appreciation Right on the date of grant, which, except in the case of Substitute Awards or in connection with an adjustment provided for in Section 12.2, shall not be less than the Fair Market Value of one (1) Share on such date of grant of the Stock Appreciation Right.
- (b) The Committee shall determine in its sole discretion whether payment of a Stock Appreciation Right shall be made in cash, in whole Shares, or any combination thereof.
- (c) The Award Agreement evidencing a grant of Stock Appreciation Rights shall be exempt from the requirements of Code Section 409A.
- (d) The provisions of Stock Appreciation Rights need not be the same with respect to each recipient.
- (e) The Committee may impose such other conditions or restrictions on the terms of exercise and the grant price of any Stock Appreciation Right, as it shall deem appropriate. A Stock Appreciation Right shall have (i) a grant price not less than Fair Market Value on the date of grant (subject to the requirements of Code Section 409A with respect to a Stock Appreciation Right granted in conjunction with, but subsequent to, an Option), and (ii) a term not greater than ten (10) years except in the event of death or disability.
- (f) Without the approval of the Company s stockholders, other than pursuant to Section 12.2, the Committee shall not (i) reduce the grant price of any Stock Appreciation Right after the date of grant, (ii) cancel any Stock Appreciation Right in exchange for cash or another Award (other than in connection with Substitute Awards), or (iii) take any other action with respect to a Stock Appreciation Right that would be treated as a repricing under the rules and regulations of the principal securities market on which the Shares are traded.
- (g) The Committee may impose such other terms and conditions on Stock Appreciation Rights granted in conjunction with any Award as the Committee shall determine in its sole discretion.

#### 7. RESTRICTED STOCK AND RESTRICTED STOCK UNITS

**7.1.** *Grants.* Awards of Restricted Stock and of Restricted Stock Units may be issued hereunder to Participants either alone or in addition to other Awards granted under the Plan (a Restricted Stock Award or Restricted Stock Unit Award respectively), and such Restricted Stock Awards and Restricted Stock Unit Awards shall also be available as a form of payment of Performance Awards and other earned cash-based incentive compensation. A Restricted Stock Award or Restricted Stock Unit Award shall be subject to vesting restrictions imposed by the Committee

covering a period of time determined by the Committee (the Vesting Period ) and specified in the applicable Award Agreement. The Committee has absolute discretion to determine whether any consideration (other than services) is to be received by the Company or any Subsidiary as a condition precedent to the issuance of Restricted Stock or Restricted Stock Units.

**7.2.** Award Agreements. The terms of any Restricted Stock Award or Restricted Stock Unit Award granted under the Plan shall be set forth in a written Award Agreement which shall contain provisions determined by the Committee and not inconsistent with the Plan. Such Award Agreement shall either comply with, or be exempt from, the requirements of Code Section 409A. The terms of Restricted Stock Awards and Restricted Stock Unit Awards need not be the same with respect to each Participant.

- **7.3.** Rights of Holders of Restricted Stock and Restricted Stock Units. Unless otherwise provided in the Award Agreement, beginning on the date of grant of the Restricted Stock Award and subject to execution of the Award Agreement, the Participant shall become a stockholder of the Company with respect to all Shares subject to the Award Agreement and shall have all of the rights of a stockholder, including the right to vote such Shares and the right to receive distributions made with respect to such Shares. A Participant receiving a Restricted Stock Unit Award shall not possess voting rights with respect to such Award. Except as otherwise provided in an Award Agreement, any Shares or any other property (other than cash) distributed as a dividend or otherwise with respect to any Restricted Stock Award or Restricted Stock Unit Award as to which the restrictions have not yet lapsed shall be subject to the same restrictions as such Restricted Stock Award or Restricted Stock Unit Award.
- **7.4.** *Issuance of Shares.* Any Restricted Stock granted under the Plan may be evidenced in such manner as the Committee may deem appropriate, including book-entry registration or issuance of a stock certificate or certificates, which certificate or certificates shall be held by the Company. Any certificate or certificates shall be registered in the name of the Participant and shall bear an appropriate legend referring to the restrictions applicable to such Restricted Stock.

#### 8. OTHER SHARE-BASED AWARDS

- **8.1.** *Grants.* Other Awards of Shares and other Awards that are valued by reference to, or are otherwise based on, Shares (Other Share-Based Awards) may be granted hereunder to Participants either alone or in addition to other Awards granted under the Plan. Other Share-Based Awards shall be subject to the Vesting Periods set forth in the applicable Award Agreements. Other Share-Based Awards shall also be available as a form of payment of other Awards granted under the Plan and other earned cash-based compensation.
- **8.2.** Award Agreements. The terms of an Other Share-Based Award granted under the Plan shall be set forth in a written Award Agreement which shall contain provisions determined by the Committee and not inconsistent with the Plan. Such Award Agreement shall either comply with, or be exempt from, the requirements of Code Section 409A. The terms of such Awards need not be the same with respect to each Participant.
- **8.3.** *Payment.* Except as may be provided in an Award Agreement, Other Share-Based Awards may be paid in cash, Shares, or any combination thereof in the sole discretion of the Committee. Other Share-Based Awards may be paid in a lump sum or in installments or, in accordance with procedures established by the Committee, on a deferred basis subject to the requirements of Code Section 409A.

#### 9. PERFORMANCE AWARDS

- **9.1.** *Grants.* Performance Awards in the form of Performance Cash or Performance Share Units, as determined by the Committee in its sole discretion, may be granted hereunder to Participants, for no consideration or for such minimum consideration as may be required by applicable law, either alone or in addition to other Awards granted under the Plan. The performance goals to be achieved for each Performance Period shall be conclusively determined by the Committee and may be based upon the criteria set forth in Section 10.2.
- **9.2.** Award Agreements. The terms of any Performance Award granted under the Plan shall be set forth in a written Award Agreement which shall contain provisions determined by the Committee and not inconsistent with the Plan, including whether such Awards shall have Dividend Equivalents. Such Award Agreement shall either comply with, or be exempt from, the requirements of Code Section 409A. The terms of Performance Awards need not be the same with respect to each Participant.

**9.3.** *Terms and Conditions.* The performance criteria to be achieved during any Performance Period and the length of the Performance Period shall be determined by the Committee upon the grant of each Performance Award; provided, however, that a Performance Period shall not be shorter than twelve (12) months. The amount of the Award to be distributed shall be conclusively determined by the Committee.

**9.4.** *Payment.* Except as provided in Article 11 or as may be provided in an Award Agreement, Performance Awards will be distributed only after the end of the relevant Performance Period. Performance Awards may be paid in cash, Shares, or any combination thereof in the sole discretion of the Committee. Performance Awards may be paid in a lump sum or in installments following the close of the Performance Period or, in accordance with procedures established by the Committee, on a deferred basis subject to the requirements of Code Section 409A.

### 10. CODE SECTION 162(m) PROVISIONS

- **10.1.** Covered Employees. Notwithstanding any other provision of the Plan, if the Committee determines at the time a Restricted Stock Award, a Restricted Stock Unit Award, a Performance Award or an Other Share-Based Award is granted to a Participant who is, or is likely to be, as of the end of the tax year in which the Company would claim a tax deduction in connection with such Award, a Covered Employee, then the Committee may provide that this Article 10 is applicable to such Award.
- **10.2.** *Performance Criteria.* If the Committee determines that a Restricted Stock Award, a Restricted Stock Unit, a Performance Award or an Other Share-Based Award is intended to be subject to this Article 10, the lapsing of restrictions thereon and the distribution of cash, Shares or other property pursuant thereto, as applicable, shall be subject to the achievement of one or more objective performance goals established by the Committee, which shall be based on the attainment of specified levels of one or any combination of the following:
- (a) Basic or diluted earnings per share of common stock, which may be calculated (i) as income calculated in accordance with Section 10.2(d), divided by (A) the weighted average number of Shares, in the case of basic earnings per share, and (B) the weighted average number of Shares and Share equivalents of common stock, in the case of diluted earnings per Share, or (ii) using such other method as may be specified by the Committee;
- (b) Cash flow, which may be calculated or measured in any manner specified by the Committee;
- (c) Economic value added, which is after-tax operating profit less the annual total cost of capital;
- (d) Income, which may include, without limitation, net income, operating income, volume measures and expense control measures, and which may be calculated or measured (i) before or after income taxes, including or excluding interest, depreciation and amortization, minority interests, extraordinary items and other material non-recurring items, discontinued operations, the cumulative effect of changes in accounting policies and the effects of any tax law changes; or (ii) using such other method as may be specified by the Committee;
- (e) Business performance or return measures (including, but not limited to, market share, debt reduction, return on assets, capital, equity, or sales), which may be calculated or measured in any manner specified by the Committee;
- (f) The price of the Company s common stock (including, but not limited to, growth measures and total stockholder return), which may be calculated or measured in any manner specified by the Committee; or

(g) Any of the above performance criteria, determined on an absolute or relative basis or as compared to the performance of a published or special index deemed applicable by the Committee including, but not limited to, the Russell 2000 or the Russell 3000 Index, as applicable, or a group of companies deemed by the Committee to be comparable to the Company.

Such performance goals also may be based solely by reference to the Company s performance or the performance of a Subsidiary, division, business segment or business unit of the Company, or based upon the relative performance of other companies or upon comparisons of any of the indicators of performance relative to other companies. As and to the extent permitted by Code Section 162(m), in the event of (i) a change in corporate capitalization, a corporate transaction or a complete or partial corporate liquidation,

(ii) a natural disaster or other significant unforeseen event that materially impacts the operation of the Company, (iii) any extraordinary gain or loss or other event that is treated for accounting purposes as an extraordinary item under generally accepted accounting principles, or (iv) any material change in accounting policies or practices affecting the Company and/or the performance goals, then, to the extent any of the foregoing events was not anticipated at the time the performance goals were established, the Committee may make adjustments to the performance goals, based solely on objective criteria, so as to neutralize the effect of the event on the applicable Award.

**10.3.** *Timing for Establishing Performance Criteria.* Performance goals shall be established not later than 90 days after the beginning of any Performance Period applicable to such Awards, or at such other earlier date as may be required or permitted for performance-based compensation under Code Section 162(m).

10.4. Settlement and Adjustments. The Committee shall at the end of the applicable Performance Period, determine whether the applicable performance goals were satisfied and the amount payable with respect to any Restricted Stock Award, Restricted Stock Unit Award, Performance Award or Other Share-Based Award. Notwithstanding any provision of the Plan (other than Article 11), with respect to any such Award that is subject to this Section 10, the Committee may adjust downwards, but not upwards, the amount payable pursuant to such Award; provided, however, that no such adjustment shall be made if it would cause the Plan or an Award to fail to comply with or be exempt from the requirements of Code Section 409A. The Committee may not waive the achievement of the applicable performance goals, except in the case of the death or disability of the Participant or as otherwise determined by the Committee in special circumstances, subject to the requirements of Code Section 162(m). All such determinations by the Committee shall be in writing and the Committee may not delegate any responsibility relating to Awards subject to this Section 10.

**10.5.** *Restrictions*. The Committee shall have the power to impose such other restrictions on Awards subject to this Article as it may deem necessary or appropriate to ensure that such Awards satisfy all requirements for performance-based compensation within the meaning of Code Section 162(m).

**10.6.** Limitations on Grants to Individual Participants. Subject to adjustment as provided for in Section 12.2, no Participant may (i) be granted Options or Stock Appreciation Rights during any period of five (5) consecutive fiscal years with respect to more than an average of 700,000 Shares per year over such five (5) consecutive fiscal year period, and (ii) earn more than an average of 700,000 Shares per year under Restricted Stock Awards, Restricted Stock Unit Awards, Performance Awards and/or Other Share-Based Awards in any period of five (5) consecutive fiscal years that are intended to comply with the performance-based exception under Code Section 162(m) and are denominated in Shares (collectively, the Limitations). In addition to the foregoing, the maximum dollar value that may be earned by any Participant in any period of five (5) consecutive fiscal years with respect to Performance Awards that are intended to comply with the performance-based exception under Code Section 162(m) and are denominated in cash is an annual average of \$5,000,000 during such five (5) consecutive fiscal year period. If an Award is cancelled, the cancelled Award shall continue to be counted toward the applicable Limitations.

#### 11. CHANGE IN CONTROL PROVISIONS

11.1. Impact on Certain Awards. Award Agreements may provide that in the event of a Change in Control of the Company: (a) Options and Stock Appreciation Rights outstanding as of the date of the Change in Control shall be cancelled and terminated without payment therefor if the Fair Market Value of one (1) Share as of the date of the Change in Control is less than the per Share Option exercise price or Stock Appreciation Right grant price, and (b) all Performance Awards shall be considered to be earned and payable (either in full or pro rata based on the portion of Performance Period completed as of the date of the Change in Control), and any limitations or other restriction shall lapse and such Performance Awards shall be immediately settled or distributed.

**11.2.** Assumption or Substitution of Certain Awards.

(a) Unless otherwise prohibited by Code Section 162(m), in the event of a Change in Control of the Company in which the successor company assumes or substitutes for an Option, Stock Appreciation Right, Restricted Stock Award, Restricted Stock Unit Award or Other Share-Based Award, if a Participant incurs a Qualifying Termination with such successor company (or a subsidiary thereof) within the Protection Period (or such other period set forth in the Award Agreement, including a period prior thereto if applicable), then, if and to the extent provided in the Award Agreement, the following shall occur: (i) Options and Stock Appreciation Rights outstanding as of the date of such termination of employment will immediately vest (i.e., immediately vest on the termination date), become fully exercisable, and may thereafter be exercised for twenty-four (24) months (or the period of time set forth in the Award Agreement), (ii) restrictions, limitations and other conditions applicable to Restricted Stock and Restricted Stock Units shall lapse and the Restricted Stock and Restricted Stock Units shall become free of all restrictions, limitations and conditions and become fully vested on the termination date, and (iii) the restrictions, limitations and other conditions applicable to any Other Share-Based Awards or any other Awards shall lapse, and such Other Share-Based Awards or such other Awards shall become free of all restrictions, limitations and conditions and become fully vested and transferable, to the full extent of the original grant, on the termination date. For the purposes of this Section 11.2, an Option, Stock Appreciation Right, Restricted Stock Award, Restricted Stock Unit Award or Other Share-Based Award shall be considered assumed or substituted for if following the Change in Control the Award confers the right to purchase or receive, for each Share subject to the Option, Stock Appreciation Right, Restricted Stock Award, Restricted Stock Unit Award or Other Share-Based Award immediately prior to the Change in Control, the consideration (whether stock, cash or other securities or property) received in the transaction constituting a Change in Control by holders of Shares for each Share held on the effective date of such transaction (and if holders were offered a choice of consideration, the type of consideration chosen by the holders of a majority of the outstanding shares); provided, however, that if such consideration received in the transaction constituting a Change in Control is not solely common stock of the successor company, the Committee may, with the consent of the successor company, provide that the consideration to be received upon the exercise or vesting of an Option, Stock Appreciation Right, Restricted Stock Award, Restricted Stock Unit Award or Other Share-Based Award, for each Share subject thereto, will be solely common stock of the successor company substantially equal in fair market value to the per share consideration received by holders of Shares in the transaction constituting a Change in Control. The determination of such substantial equality of value of consideration shall be made by the Committee in its sole discretion, and its determination shall be final, conclusive and binding.

(b) Unless otherwise provided in an Award Agreement or, to the extent applicable, prohibited by Code Section 162(m), in the event of a Change in Control of the Company to the extent the successor company does not assume or substitute for an Option, Stock Appreciation Right, Restricted Stock Award, Restricted Stock Unit Award or Other Share-Based Award: (i) those Options and Stock Appreciation Rights outstanding as of the date of the Change in Control that are not assumed or substituted for shall vest and become fully exercisable immediately prior to the Change in Control, (ii) restrictions, limitations and other conditions on Restricted Stock and Restricted Stock Units that are not assumed or substituted for shall lapse and the Restricted Stock and Restricted Stock Units shall become free of all restrictions, limitations and conditions and become fully vested immediately prior to the Change in Control, and (iii) the restrictions, limitations and other conditions applicable to any Other Share-Based Awards or any other Awards that are not assumed or substituted for shall lapse, and such Other Share-Based Awards or such other Awards shall become free of all restrictions, limitations and conditions and become fully vested and transferable, to the full extent of the original grant, immediately prior to the Change in Control.

(c) The Committee, in its discretion, and to the extent applicable, consistent with Code Section 162(m), may determine that, upon the occurrence of a Change in Control of the Company, each Option and Stock Appreciation Right outstanding shall terminate within a specified number of days after notice to the Participant, and/or that each Participant shall receive, with respect to each Share subject to such Option or Stock Appreciation Right, an amount equal to the excess of the Fair Market Value of such Share immediately prior to the occurrence of such Change in Control over the exercise price per share of such

Option and/or Stock Appreciation Right; such amount to be payable in cash, in one or more kinds of stock or property (including the stock or property, if any, payable in the transaction) or in a combination thereof, as the Committee, in its discretion, shall determine.

- 11.3. Change in Control. For purposes of the Plan, Change in Control means the occurrence of any one of the following events:
- (a) A change in the ownership of the Company which will occur on the date that any one person, or more than one person acting as a group within the meaning of Code Section 409A, acquires ownership of stock in the Company that, together with stock held by such person or group, constitutes more than fifty percent (50%) of the total Fair Market Value or total voting power of the stock of the Company. However, if any one person or more than one person acting as a group, is considered to own more than fifty percent (50%) of the total Fair Market Value or total voting power of the stock of the Company, the acquisition of additional stock by the same person or persons will not be considered a change in the ownership of the Company (or to cause a change in the effective control of the Company within the meaning of paragraph (b) below). Further, an increase of the effective percentage of stock owned by any one person, or persons acting as a group, as a result of a transaction in which the Company acquires its stock in exchange for property will be treated as an acquisition of stock for purposes of this paragraph; provided, that for purposes of this Section 11.3(a), the following acquisitions of Company stock will not constitute a Change in Control: (i) any acquisition by any employee benefit plan (or related trust) sponsored or maintained by the Company or an Affiliate (as defined below), (ii) any acquisition directly from the Company or (iii) any acquisition by the Company. This paragraph (a) applies only when there is a transfer of the stock of the Company (or issuance of stock) and stock in the Company remains outstanding after the transaction.
- (b) A change in the effective control of the Company which will occur on the date that either:
- (i) any one person, or more than one person acting as a group within the meaning of Code Section 409A, acquires (or has acquired during the twelve (12) month period ending on the date of the most recent acquisition by such person or persons) ownership of stock of the Company possessing thirty-five percent (35%) or more of the total voting power of the stock of the Company (not considering stock owned by such person or group prior to such twelve (12) month period) (<u>i.e.</u>, such person or group must acquire within a twelve (12) month period stock possessing thirty-five percent (35%) of the total voting power of the stock of the Company) except for (A) any acquisition by any employee benefit plan (or related trust) sponsored or maintained by the Company or an Affiliate (as defined below), (B) any acquisition directly from the Company or (C) any acquisition by the Company; or
- (ii) a majority of the members of the Board are replaced during any twelve (12) month period by Directors whose appointment or election is not endorsed by a majority of the members of the Board prior to the date of the appointment or election.

For purposes of a change in the effective control of the Company, if any one person, or more than one person acting as a group, is considered to effectively control the Company within the meaning of this paragraph (b), the acquisition of additional control of the Company by the same person or persons is not considered a change in the effective control of the Company, or to cause a change in the ownership of the Company within the meaning of paragraph (a) of this Section.

(c) A change in the ownership of a substantial portion of the Company s assets which will occur on the date that any one person, or more than one person acting as a group, acquires (or has acquired during the twelve (12) month period ending on the date of the most recent acquisition by such person or persons) assets of the Company that have a total gross fair market value equal to or more than forty percent (40%) of the total gross fair market value of all the assets of the Company immediately prior to such acquisition or acquisitions. For this purpose, gross fair market value means the value of the assets of the Company, or the value of the assets being disposed of, determined without regard to any liabilities associated with such

assets. Any transfer of assets to an entity that is controlled by the stockholders of the Company immediately after the transfer, as provided in guidance issued pursuant to Code Section 409A, will not constitute a Change in Control.

(d) A liquidation or dissolution of the Company that is approved by a majority of the Company s stockholders.

For purposes of this Section 11.3, the provisions of Code Section 318(a) regarding the constructive ownership of stock will apply to determine stock ownership; provided that, stock underlying unvested options (including options exercisable for stock that is not substantially vested) will not be treated as owned by the individual who holds the option. The term Affiliate for purposes of this Section 11.3 means a corporation that is a member of a controlled group of corporations (as defined in Code Section 414(b)) that includes the Company, any trade or business (whether or not incorporated) that is under common control (as defined in Code Section 414(c)) with the Company, or any entity that is a member of the same affiliated service group (as defined in Code Section 414(m)) as the Company.

#### 12. GENERALLY APPLICABLE PROVISIONS

**12.1.** *Amendment and Termination of the Plan.* 

- (a) The Committee and the Board may, from time to time, modify alter or amend the Plan as it shall deem advisable, subject to any requirement for stockholder approval imposed by applicable law, including the rules and regulations of the principal securities market on which the Shares are traded; provided, that the Plan may not be modified, altered or amended in any manner that would result in noncompliance with Rule 16b-3 of the Exchange Act; and further provided that, without the approval of the Company s stockholders, Plan may not be modified, altered or amended in any manner to (i) increase the number of Shares that may be the subject of Awards under the Plan (except for adjustments pursuant to Section 12.2), (ii) expand the types of Awards available under the Plan, (iii) materially expand the class of persons eligible to participate in the Plan, (iv) amend any provision of Section 5.3, Section 6.2(e) or Section 6.2(f) (regarding changes in the exercise price of Options and Stock Appreciation Rights), (v) increase the maximum permissible term of any Option specified by Section 5.4 or the maximum permissible term of a Stock Appreciation Right specified by Section 6.2(e), or (vi) increase the Limitations set forth in Section 10.6. The Committee may not, without the approval of the Company s stockholders, take any other action with respect to an Option or Stock Appreciation Right that would be treated as a repricing under the rules and regulations of the principal securities exchange on which the Shares are traded, including a reduction of the exercise price of an Option or the grant price of a Stock Appreciation Right or the exchange of an Option or Stock Appreciation Right for cash or another Award.
- (b) The Company may, by action of the Board, terminate the Plan at any time and for any reason, subject to any requirement for stockholder approval imposed by applicable law, including the rules and regulations of the principal securities market on which the Shares are traded; provided, that the Plan may not be modified, altered or amended in any manner that would result in noncompliance with Rule 16b-3 of the Exchange Act.
- (c) No amendments to, or termination of, the Plan shall impair in any material respect the rights of a Participant under any Award previously granted without such Participant s consent except as required to comply with applicable securities laws or Code Section 409A.
- 12.2. Adjustments. In the event of any merger, reorganization, consolidation, recapitalization, dividend or distribution (whether in cash, Shares or other property, other than a regular cash dividend), stock split, reverse stock split, spin-off or similar transaction or other change in corporate structure affecting the Shares or the value thereof, such adjustments and other substitutions shall be made to the Plan and to Awards as the Committee deems equitable or appropriate taking into consideration the accounting and tax consequences, including such adjustments in the aggregate number, class and kind of securities that may be delivered under the Plan, the

Limitations, the maximum number of Shares that may be issued as incentive stock options and, in the aggregate or to any one Participant, in the number, class, kind and option or exercise price of securities subject to outstanding Awards granted under the Plan (including, if the Committee deems appropriate, the substitution of similar options to purchase the shares of, or other awards denominated in the shares of, another company) as the Committee may determine to be appropriate; provided, however, that no such adjustment or other substitution shall be made if it would cause the Plan or an Award to fail to comply with or be exempt from the requirements of Code Section 409A and provided, further, that the number of Shares subject to any Award shall always be a whole number.

12.3. Transferability of Awards. Except as provided below, no Award and no Shares subject to Awards that have not been issued or as to which any applicable restriction, performance or deferral period has not lapsed, may be sold, assigned, transferred, pledged or otherwise encumbered, other than by will or the laws of descent and distribution, and such Award may be exercised during the life of the Participant only by the Participant or the Participant s guardian or legal representative. To the extent and under such terms and conditions as determined by the Committee, a Participant may assign or transfer an Award (each transferee thereof, a Permitted Assignee ) to (a) the Participant s spouse, children or grandchildren (including any adopted and step children or grandchildren), parents, grandparents or siblings (the Covered Family Members ), (b) to a trust for the benefit of one or more of the Participant or the Covered Family Members, (c) to a partnership, limited liability company or corporation in which the Participant or one or more of the or the Covered Family Members are the only partners, members or stockholders, (d) for charitable donations or (e) pursuant to a domestic relations order entered or approved by a court of competent jurisdiction; provided, that such Permitted Assignee shall be bound by and subject to all of the terms and conditions of the Plan and the Award Agreement relating to the transferred Award and shall execute an agreement satisfactory to the Company evidencing such obligations; and provided further that such Participant shall remain bound by the terms and conditions of the Plan. The Company shall cooperate with any Permitted Assignee and the Company s transfer agent in effectuating any transfer permitted under this Section.

**12.4.** *Termination of Employment.* Subject to Article 11, the Committee shall determine and set forth in each Award Agreement whether any Awards granted in such Award Agreement will (a) in the case of Options or Stock Appreciation Rights, continue to be or become exercisable and, if so, the terms of exercise, and (b) in the case of Restricted Stock, Restricted Stock Units, Performance Awards or Other Share-Based Awards, cease to be subject to any applicable restrictions, limitations and other conditions, and if so, the timing of the removal of such restrictions, limitations and conditions, after the date that a Participant ceases to be employed by or to provide services to the Company or any Subsidiary (including as a Director or Consultant), whether by reason of death, disability, voluntary or involuntary termination of employment or services, or otherwise. The date of termination of a Participant s employment or services will be determined by the Committee, which determination will be final.

**12.5.** *Deferral*; *Dividend Equivalents*. The Committee shall be authorized to establish procedures pursuant to which the payment of any Award may be deferred. Such procedures may include, without limitation, provisions for the payment or crediting of reasonable interest on installment or deferred payments or the grant or crediting of Dividend Equivalents or other amounts in respect of deferred payments denominated in Shares. Any deferral shall only be allowed as is provided in a separate deferred compensation plan adopted by the Company. This Plan shall not constitute an employee benefit plan for purposes of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended.

Subject to the provisions of the Plan and any Award Agreement, the recipient of an Award (including any deferred Award) may, if so determined by the Committee, be entitled to receive, currently or on a deferred basis, cash, stock or other property dividends, or cash payments in amounts equivalent to cash, stock or other property dividends on Shares ( Dividend Equivalents ) with respect to the number of Shares covered by the Award, as determined by the Committee, in its sole discretion. The Committee may provide that such Dividend Equivalents (if any) shall be either be (a) paid with respect to such Award on the dividend payment date in cash or in

unrestricted Shares having a Fair Market Value equal to the amount of such dividends or (b) deferred and the amount or value thereof automatically reinvested in additional Shares, other Awards or otherwise reinvested and may provide that such Dividend Equivalents are subject to the same vesting or performance conditions as the underlying Award.

#### 13. MISCELLANEOUS

- 13.1. Award Agreements. Each Award Agreement shall either be (a) in writing in a form approved by the Committee and executed by the Company by an officer duly authorized to act on its behalf, or (b) an electronic notice in a form approved by the Committee and recorded by the Company (or its designee) in an electronic recordkeeping system used for the purpose of tracking one or more types of Awards as the Committee may provide; in each case and if required by the Committee, the Award Agreement shall be executed or otherwise electronically accepted by the recipient of the Award in such form and manner as the Committee may require. The Committee may authorize any officer of the Company to execute any or all Award Agreements on behalf of the Company. The Award Agreement shall set forth the material terms and conditions of the Award as established by the Committee consistent with the provisions of the Plan.
- 13.2. Tax Withholding. The Company shall have the right to make all payments or distributions pursuant to the Plan to a Participant (or a Permitted Assignee thereof) (any such person, a Payee) net of any applicable federal, state and local taxes required to be paid or withheld as a result of (a) the grant of any Award, (b) the exercise of an Option or Stock Appreciation Right, (c) the delivery of Shares or cash, (d) the lapse of any restrictions in connection with any Award or (e) any other event occurring pursuant to the Plan. The Company or any Subsidiary shall have the right to withhold from wages or other amounts otherwise payable to such Payee such withholding taxes as may be required by law, or to otherwise require the Payee to pay such withholding taxes. If the Payee shall fail to make such tax payments as are required, the Company or its Subsidiaries shall, to the extent permitted by law, have the right to deduct any such taxes from any payment of any kind otherwise due to such Payee or to take such other action as may be necessary to satisfy such withholding obligations. The Committee shall be authorized to establish procedures for election by Participants to satisfy such obligation for the payment of such taxes by tendering previously acquired Shares (either actually or by attestation, valued at their then Fair Market Value), or by directing the Company to retain Shares (up to the Participant s minimum required tax withholding rate or such other rate that will not trigger a negative accounting impact) otherwise deliverable in connection with the Award.
- 13.3. Right of Discharge Reserved; Claims to Awards. Nothing in the Plan nor the grant of an Award hereunder shall confer upon any Employee, Director or Consultant the right to continue in the employment or service of the Company or any Subsidiary or affect any right that the Company or any Subsidiary may have to terminate the employment or service of (or to demote or to exclude from future Awards under the Plan) any such Employee, Director or Consultant at any time for any reason. Except as specifically provided by the Committee, the Company shall not be liable for the loss of existing or potential profit from an Award granted in the event of termination of an employment or other relationship. No Employee, Director, Consultant or Participant shall have any claim to be granted any Award under the Plan, and there is no obligation for uniformity of treatment of Employees, Directors, Consultants or Participants under the Plan.
- **13.4.** Substitute Awards. Notwithstanding any other provision of the Plan, the terms of Substitute Awards may vary from the terms set forth in the Plan to the extent the Committee deems appropriate to conform, in whole or in part, to the provisions of the awards in substitution for which they are granted.
- 13.5. Cancellation of Award. Notwithstanding anything to the contrary contained herein, an Award Agreement may provide that the Award shall be canceled if the Participant, without the consent of the Company, while employed by the Company or any Subsidiary or after termination of such employment or service, establishes a relationship with a competitor of the Company or any Subsidiary or engages in activity that is in conflict with or adverse to the interest of the Company or any Subsidiary, as determined by the Committee in its

sole discretion. The Committee may provide in an Award Agreement that if within the time period specified in the Agreement the Participant establishes a relationship with a competitor or engages in an activity referred to in the preceding sentence, the Participant will forfeit any gain realized on the vesting or exercise of the Award and must repay such gain to the Company.

- 13.6. Stop-Transfer Orders. Any Shares delivered under the Plan pursuant to any Award shall be subject to such stop-transfer orders and other restrictions as the Committee may deem advisable under the rules, regulations and other requirements of the Securities and Exchange Commission, any stock exchange upon which the Shares are then listed, and any applicable federal or state securities law, and the Committee may cause a legend or legends to be put on any such Shares (or certificates for such Shares, if applicable) to make appropriate reference to such restrictions.
- **13.7.** *Nature of Payments.* All Awards made pursuant to the Plan are in consideration of services performed or to be performed for the Company or any Subsidiary, division or business unit of the Company. Any income or gain realized pursuant to Awards under the Plan constitutes a special incentive payment to the Participant and shall not be taken into account, to the extent permissible under applicable law, as compensation for purposes of any of the employee benefit plans of the Company or any Subsidiary except as may be determined by the Committee or by the Board or board of directors of the applicable Subsidiary.
- **13.8.** Other Plans. Nothing contained in the Plan shall prevent the Board from adopting other or additional compensation arrangements, subject to stockholder approval if such approval is required; and such arrangements may be either generally applicable or applicable only in specific cases.
- 13.9. Severability. If any provision of the Plan shall be held unlawful or otherwise invalid or unenforceable in whole or in part by a court of competent jurisdiction, such provision shall (a) be deemed limited to the extent that such court of competent jurisdiction deems it lawful, valid and/or enforceable and as so limited shall remain in full force and effect, and (b) not affect any other provision of the Plan or part thereof, each of which shall remain in full force and effect. If the making of any payment or the provision of any other benefit required under the Plan shall be held unlawful or otherwise invalid or unenforceable by a court of competent jurisdiction, such unlawfulness, invalidity or unenforceability shall not prevent any other payment or benefit from being made or provided under the Plan, and if the making of any payment in full or the provision of any other benefit required under the Plan in full would be unlawful or otherwise invalid or unenforceable, then such unlawfulness, invalidity or unenforceability shall not prevent such payment or benefit from being made or provided in part, to the extent that it would not be unlawful, invalid or unenforceable shall be made or provided under the Plan.
- **13.10.** Construction. As used in the Plan, the words include and including, and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words without limitation.
- **13.11.** *Unfunded Status of the Plan.* The Plan is intended to constitute an unfunded plan for incentive compensation. With respect to any payments not yet made to a Participant by the Company, nothing contained herein shall give any such Participant any rights that are greater than those of a general creditor of the Company. In its sole discretion, the Committee may authorize the creation of trusts or other arrangements to meet the obligations created under the Plan to deliver the Shares or payments in lieu of or with respect to Awards hereunder; provided, however, that the existence of such trusts or other arrangements is consistent with the unfunded status of the Plan.
- **13.12.** *Governing Law.* The Plan and all determinations made and actions taken thereunder, to the extent not otherwise governed by the Code or the laws of the United States, shall be governed by the laws of the State of Texas, without reference to principles of conflict of laws, and construed accordingly.

**13.13.** *Effective Date of Plan; Termination of Plan.* The Plan shall be effective on the date of the approval of the Plan by the holders of the shares entitled to vote at a duly constituted meeting of the stockholders of the

Company. The Plan shall be null and void and of no effect if the foregoing condition is not fulfilled and in such event, each Award shall, notwithstanding any of the preceding provisions of the Plan, be null and void and of no effect. Awards may be granted under the Plan at any time and from time to time on or prior to the tenth anniversary of the effective date of the Plan, on which date the Plan will expire except as to Awards then outstanding under the Plan. Such outstanding Awards shall remain in effect until they have been exercised or terminated, or have expired.

**13.14.** Foreign Employees. Awards may be granted to Participants who are foreign nationals or employed outside the United States, or both, on such terms and conditions different from those applicable to Awards to Employees employed in the United States as may, in the judgment of the Committee, be necessary or desirable in order to recognize differences in local law or tax policy. The Committee also may impose conditions on the exercise or vesting of Awards in order to minimize the Company s obligation with respect to tax equalization for Employees on assignments outside their home country.

13.15. Compliance with Code Section 409A. This Plan is intended to comply, and shall be administered in a manner that is intended to comply, with Code Section 409A and shall be construed and interpreted in accordance with such intent. To the extent that an Award or the payment, settlement or deferral thereof is subject to Code Section 409A, the Award shall be granted, paid, settled or deferred in a manner that will comply with Code Section 409A, including regulations or other guidance issued with respect thereto, except as otherwise determined by the Committee. Any provision of this Plan that would cause the grant of an Award or the payment, settlement or deferral thereof to fail to satisfy Code Section 409A shall be amended to comply with Code Section 409A on a timely basis, which may be made on a retroactive basis, in accordance with regulations and other guidance issued under Code Section 409A. Further, if, on the date of a Participant s separation from service (as defined in Code Section 409A) other than upon the occurrence of death, the Participant is a specified employee (as defined in Code Section 409A), no payments of any amounts hereunder that are subject to Code Section 409A shall be made until the earliest date on which payment is permissible under 409A(a)(2)(B)(i) (the six (6) month delay rule for specified employees).

**13.16.** Captions. The captions in the Plan are for convenience of reference only, and are not intended to narrow, limit or affect the substance or interpretation of the provisions contained herein.

#### APPENDIX B

#### MICROTUNE, INC.

#### 2010 DIRECTOR STOCK PLAN

(Effective as of May 20, 2010)

Microtune, Inc. (the Company ), a Delaware corporation, hereby establishes and adopts the following Microtune, Inc. 2010 Director Stock Plan (the Plan ).

#### 1. PURPOSE OF THE PLAN

The purpose of the Plan is to assist the Company in attracting and retaining the best available personnel for service as Outside Directors (as defined herein) of the Company, to provide additional incentive to the Outside Directors of the Company to serve as Directors (as defined herein), and to encourage their continued service on the Board (as defined herein).

#### 2. **DEFINITIONS**

- **2.1.** Affiliate shall have the meaning set forth in the last paragraph of Section 6.2.
- 2.2. Award means any award of Restricted Stock Units granted pursuant to the provisions of the Plan.
- **2.3.** Award Agreement means any agreement, contract or other instrument or document evidencing any Award hereunder, including through an electronic medium.
- **2.4.** *Board* means the board of directors of the Company.
- **2.5.** Change in Control shall have the same meaning as set forth in Section 6.2.
- 2.6. Code means the Internal Revenue Code of 1986, as amended.
- 2.7. Committee means the Board or a committee thereof formed by the Board to act as the Committee hereunder.

<b>2.8.</b> <i>Company</i> shall have the same meaning as set forth in the introductory paragraph of this document.
<b>2.9.</b> Covered Family Members shall have the same meaning as set forth in Section 7.3.
<b>2.10.</b> <i>Director</i> means a member of the Board.
<b>2.11.</b> <i>Employee</i> means any employee of the Company or any Subsidiary, including officers and Inside Directors, and any prospective employee conditioned upon, and effective not earlier than, such person becoming an employee of the Company or any Subsidiary. The payment of a Director s fee by the Company shall not be sufficient in and of itself to constitute employment by the Company.
<b>2.12.</b> Exchange Act means the Securities Exchange Act of 1934, as amended.
<b>2.13.</b> Fair Market Value means the per Share closing price of the Shares as reported on The NASDAQ Global Market or The NASDAQ Capita Market of The NASDAQ Stock Market as of the relevant date (or if there were no reported prices on such date, on the last preceding date on which the prices were reported) or, if the Company is not then listed on The NASDAQ Stock Market, on such other principal securities exchange on which the Shares are traded, and if the Company is not listed on The NASDAQ Stock Market or any other securities exchange, the Fair Market Value of Shares shall be determined by the Committee in its sole discretion.
<b>2.14.</b> <i>Inside Director</i> means a Director who is an Employee.
<b>2.15.</b> <i>Outside Director</i> means a Director who is not an Employee.

<b>2.16.</b> <i>Participant</i> means a Director who receives an Award under the Plan.
<b>2.17.</b> <i>Payee</i> shall have the meaning set forth in Section 8.2.
<b>2.18.</b> <i>Permitted Assignee</i> shall have the meaning set forth in Section 7.3.
<b>2.19.</b> <i>Plan</i> shall have the same meaning as set forth in the introductory paragraph of this document.
<b>2.20.</b> Restricted Stock Unit means an Award that is valued by reference to a Share, which value shall be paid to the Participant by delivery of Shares, and that has such restrictions as set forth herein.
<b>2.21.</b> <i>RSU Shares</i> shall have the meaning set forth in Section 5.4.
<b>2.22.</b> <i>Shares</i> means the shares of common stock of the Company.
<b>2.23.</b> Subsidiary means any corporation (other than the Company) in an unbroken chain of corporations beginning with the Company if, at the relevant time each of the corporations other than the last corporation in the unbroken chain owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in the chain.
<b>2.24.</b> <i>Vesting Period</i> shall have the meaning set forth in Section 5.3.
3. SHARES SUBJECT TO THE PLAN
3.1. Number of Shares.
(a) Subject to the adjustment provided for in Section 7.2, a total of 250,000 Shares shall be authorized for grant under the Plan. Any Shares that are subject to Awards shall be counted against this limit as one (1) Share for every one (1) Share granted.
(b) If (i) any Shares subject to an Award are forfeited, cancelled or expire or (ii) an Award is settled for cash (in whole or in part), the Shares subject to such Award shall, to the extent of such forfeiture, cancellation, expiration or cash settlement, again be available for Awards under the Plan. Notwithstanding anything to the contrary contained herein, Shares tendered by the Participant or withheld by the Company to satisfy any tax withholding obligation (or portion thereof) with respect to an Award shall be added back to the Shares authorized for grant under the Plan. Any Shares that again become available for grant pursuant to this Section 3.1 shall be authorized for grant as one (1) Share for any one (1) Share that may be added back pursuant to this Section 3.1(b).

(c) In the event that a company acquired by the Company or any Subsidiary or with which the Company or any Subsidiary combines has shares available under a pre-existing plan approved by stockholders and not adopted in contemplation of such acquisition or combination, the shares available for grant pursuant to the terms of such pre-existing plan (as adjusted, to the extent appropriate, using the exchange ratio or other adjustment or valuation ratio or formula used in such acquisition or combination to determine the consideration payable to the holders of common stock of the entities party to such acquisition or combination) may be used for Awards under the Plan and shall not reduce the Shares authorized for grant under the Plan; provided, that Awards using such available shares shall not be made after the date awards or grants could have been made under the terms of the pre-existing plan, absent the acquisition or combination, and shall only be made to individuals who were not Outside Directors prior to such acquisition or combination.

(d) In the event that any Award granted under the Plan would otherwise cause the number of Shares subject to outstanding Awards plus the number of Shares previously issued and distributed under Restricted Stock Units to exceed the number of Shares set forth in Section 3(a), then the remaining Shares available for Awards shall be granted to the Outside Directors pursuant to Article 5 on a pro rata basis. No further grants

shall be made until such time, if any, as additional Shares become available for grant under the Plan through action of the Board or the stockholders to increase the number of Shares which may be issued under the Plan or through cancellation or expiration of Awards previously granted hereunder. The Committee may adopt reasonable counting procedures to ensure appropriate counting, avoid double counting and make adjustments if the number of Shares actually delivered differs from the number of Shares previously counted in connection with an Award.

**3.2.** Character of Shares. Any Shares issued hereunder may consist, in whole or in part, of authorized and unissued shares, treasury shares or shares purchased in the open market or otherwise.

#### 4. ELIGIBILITY AND ADMINISTRATION

- **4.1.** *Eligibility*. Awards may be granted only to Outside Directors. All Awards shall be automatically granted in accordance with the applicable terms set forth in Section 5.2 hereof.
- 4.2. Administration.
- (a) The Plan shall be administered on a limited basis by the Committee. The Committee shall have full power and authority, subject to the provisions of the Plan and subject to such orders or resolutions not inconsistent with the provisions of the Plan, to: (i) determine whether, to what extent and under what circumstances any Award shall be canceled or suspended; (ii) interpret and administer the Plan and any instrument or agreement entered into under or in connection with the Plan, including any Award Agreement; (iii) correct any defect, supply any omission or reconcile any inconsistency in the Plan or any Award Agreement in the manner and to the extent that the Committee shall deem desirable to carry it into effect; (iv) establish such rules and regulations and appoint such agents as it shall deem appropriate for the proper administration of the Plan; and (v) make any other determination and take any other action that the Committee deems necessary or desirable for administration of the Plan.
- (b) Decisions of the Committee shall be final, conclusive and binding on all persons or entities, including the Company and any Participant. A majority of the members of the Committee may determine its actions, including fixing the time and place of its meetings.

#### 5. GRANTS OF AWARDS OF RESTRICTED STOCK UNITS

- **5.1.** *Grants in General.* All grants of Awards to Outside Directors under this Plan shall be automatic and nondiscretionary and shall be made strictly in accordance with this Article 5. No person shall have any discretion to select which Outside Directors shall be granted Awards or to determine the number of Shares to be covered by Awards.
- **5.2.** *Procedure for Grants.* Subject to Section 5.1 above, during any period in which Awards shall be granted to Outside Directors (which period shall commence after May 20, 2010), all such grants of Awards to Outside Directors under this Plan shall be made strictly in accordance with the following provisions:
- (a) Each Outside Director shall be granted an Award of 5,250 Restricted Stock Units on the date on which such person first becomes an Outside Director, whether through election by the stockholders of the Company or appointment by the Board to fill a vacancy; provided, however, that an Inside Director who ceases to be an Inside Director but who remains a Director shall not receive a grant of the Award described in this

Section 5.2(a).

- (b) Each Outside Director shall be granted an Award of 8,400 Restricted Stock Units on the date of the Company s annual stockholders meeting of each year; provided, that he or she is then an Outside Director and if as of such date, he or she shall have served on the Board for at least the preceding six (6) months.
- **5.3.** Vesting Period. Each Award shall become vested as to one-third (1/3rd) of the Restricted Stock Units subject to such Award on each anniversary of its date of grant; provided, that the Participant continues to serve as

a Director on such dates (the Vesting Period). Except as otherwise provided in the Plan, an Award Agreement or any other agreement between the Participant and the Company, if the termination of the Participant s continuous status as a Director occurs for any reason (including, but not limited to, the Participant s death or disability) or there is a liquidation or dissolution of the Company, all unvested Restricted Stock Units shall be forfeited as of the termination, liquidation or dissolution date (as applicable).

- **5.4.** Settlement of Awards. Except as otherwise provided in the Plan or an Award Agreement, to the extent the Restricted Stock Units subject to an Award become vested hereunder, the Company shall settle such vested Restricted Stock Units, and as a result thereof: (a) issue and deliver to the Participant one Share for each such vested Restricted Stock Unit (the RSU Shares) and (b) have the Participant s name entered as a shareholder of record with respect to the RSU Shares on the books of the Company or its duly authorized transfer agent. RSU Shares shall be issued to the Participant as soon as practicable after the vesting of the vested Restricted Stock Units, but in no event later than seventy-five (75) days after such vesting.
- **5.5.** Rights of Holders of Restricted Stock Units. A Participant receiving an Award shall not possess voting rights with respect to such Award. Until the issuance of the RSU Shares (as evidenced by the appropriate entry on the books of the Company or its duly authorized transfer agent), no right to vote or receive dividends or any other rights as a stockholder shall exist with respect to any Restricted Stock Units, notwithstanding the occurrence of the time or event that gives rise to the settlement of such Restricted Stock Units. No adjustment shall be made for a dividend or other right for which the record date is prior to the date the RSU Shares are issued.
- **5.6.** *Recordkeeping.* The number of vested and unvested Restricted Stock Units subject to an Award and issued RSU Shares pursuant to the applicable terms of this Article 5 shall be credited (and adjusted thereafter as necessary) by the Committee to a bookkeeping account established in the name of the Outside Director.

#### 6. CHANGE IN CONTROL PROVISIONS

- **6.1.** Assumption and Substitution of Awards.
- (a) In the event of a Change in Control of the Company in which the successor company assumes or substitutes an Award, and (i) a Participant s status as a director of the successor company (or a subsidiary thereof) terminates other than upon the voluntary resignation of the Participant, or (ii) the Participant is not offered a director-level position with the successor company prior to the Change in Control, then, all unvested Restricted Stock Units subject to the Participant s Awards shall become fully vested on the termination date or immediately prior to the Change in Control, as applicable. RSU Shares shall be issued to the Participant (i) as soon as practicable after the vesting of the Restricted Stock Units upon termination, but in no event later than seventy-five (75) days after such vesting, and (ii) concurrent with the vesting of the Restricted Stock Units that occurs immediately prior to the Change in Control.
- (b) In the event of a Change in Control of the Company and to the extent the successor company does not assume or substitute for an Award, the restrictions, limitations and other conditions on the Restricted Stock Units that are not assumed or substituted for shall lapse and such Restricted Stock Units shall become fully vested immediately prior to the Change in Control. RSU Shares shall be issued to the Participant as soon as practicable after the vesting of the vested Restricted Stock Units, but in no event later than seventy-five (75) days after such vesting; provided, that RSU Shares shall be issued to the Participant concurrent with the vesting of the Restricted Stock Units that occurs immediately prior to the Change in Control.
- (c) For the purposes of this Section 6.1, an Award shall be considered assumed or substituted for if following the Change in Control the Award would confer the right to purchase or receive, for each outstanding Restricted Stock Unit, the consideration (whether stock, cash or other

securities or property) received in the transaction constituting a Change in Control by holders of Shares for each Share held on the

-4-

effective date of such transaction (and if holders were offered a choice of consideration, the type of consideration chosen by the holders of a majority of the outstanding Shares); provided, however, that if such consideration received in the transaction constituting a Change in Control is not solely common stock of the successor company, the Committee may, with the consent of the successor company, provide that the consideration to be received upon the vesting of an Award of Restricted Stock Units will be solely common stock of the successor company substantially equal in fair market value to the per Share consideration received by holders of Shares in the transaction constituting a Change in Control. The determination of such substantial equality of value of consideration shall be made by the Committee in its sole discretion and, its determination shall be conclusive and binding.

- **6.2.** Change in Control. For purposes of the Plan, Change in Control means the occurrence of any one of the following events:
- (a) A change in the ownership of the Company which will occur on the date that any one person, or more than one person acting as a group within the meaning of Code Section 409A, acquires ownership of stock in the Company that, together with stock held by such person or group, constitutes more than fifty percent (50%) of the total Fair Market Value or total voting power of the stock of the Company. However, if any one person or more than one person acting as a group, is considered to own more than fifty percent (50%) of the total Fair Market Value or total voting power of the stock of the Company, the acquisition of additional stock by the same person or persons will not be considered a change in the ownership of the Company (or to cause a change in the effective control of the Company within the meaning of paragraph (b) below). Further, an increase of the effective percentage of stock owned by any one person, or persons acting as a group, as a result of a transaction in which the Company acquires its stock in exchange for property will be treated as an acquisition of stock for purposes of this paragraph; provided, that for purposes of this Section 6.2(a), the following acquisitions of Company stock will not constitute a Change in Control: (i) any acquisition by any employee benefit plan (or related trust) sponsored or maintained by the Company or an Affiliate (as defined below), (ii) any acquisition directly from the Company or (iii) any acquisition by the Company. This paragraph (a) applies only when there is a transfer of the stock of the Company (or issuance of stock) and stock in the Company remains outstanding after the transaction.
- (b) A change in the effective control of the Company which will occur on the date that either:
- (i) Any one person, or more than one person acting as a group within the meaning of Code Section 409A, acquires (or has acquired during the twelve (12) month period ending on the date of the most recent acquisition by such person or persons) ownership of stock of the Company possessing thirty-five percent (35%) or more of the total voting power of the stock of the Company (not considering stock owned by such person or group prior to such twelve (12) month period) (i.e., such person or group must acquire within a twelve (12) month period stock possessing thirty-five percent (35%) of the total voting power of the stock of the Company) except for (A) any acquisition by any employee benefit plan (or related trust) sponsored or maintained by the Company or an Affiliate (as defined below), (B) any acquisition directly from the Company or (C) any acquisition by the Company; or
- (ii) A majority of the members of the Board are replaced during any twelve (12) month period by Directors whose appointment or election is not endorsed by a majority of the members of the Board prior to the date of the appointment or election.

For purposes of a change in the effective control of the Company, if any one person, or more than one person acting as a group, is considered to effectively control the Company within the meaning of this paragraph (b), the acquisition of additional control of the Company by the same person or persons is not considered a change in the effective control of the Company, or to cause a change in the ownership of the Company within the meaning of paragraph (a) of this Section.

(c) A change in the ownership of a substantial portion of the Company s assets which will occur on the date that any one person, or more than one person acting as a group, acquires (or has acquired during the

twelve (12) month period ending on the date of the most recent acquisition by such person or persons) assets of the Company that have a total gross fair market value equal to or more than forty percent (40%) of the total gross fair market value of all the assets of the Company immediately prior to such acquisition or acquisitions. For this purpose, gross fair market value means the value of the assets of the Company, or the value of the assets being disposed of, determined without regard to any liabilities associated with such assets. Any transfer of assets to an entity that is controlled by the stockholders of the Company immediately after the transfer, as provided in guidance issued pursuant to Code Section 409A, will not constitute a Change in Control.

For purposes of this Section 6.2, the provisions of Code Section 318(a) regarding the constructive ownership of stock will apply to determine stock ownership; provided, that stock underlying unvested options (including options exercisable for stock that is not substantially vested) will not be treated as owned by the individual who holds the option. The term Affiliate for purposes of this Section 6.2 means a corporation that is a member of a controlled group of corporations (as defined in Code Section 414(b)) that includes the Company, any trade or business (whether or not incorporated) that is under common control (as defined in Code Section 414(c)) with the Company, or any entity that is a member of the same affiliated service group (as defined in Code Section 414(m)) as the Company.

#### 7. GENERALLY APPLICABLE PROVISIONS

- **7.1.** Amendment and Termination of the Plan.
- (a) The Committee and the Board may, from time to time, modify alter or amend the Plan as it shall deem advisable, subject to any requirement for stockholder approval imposed by applicable law, including the rules and regulations of the principal securities market on which the Shares are traded; provided, that the Plan may not be modified, altered or amended in any manner that would result in noncompliance with Rule 16b-3 of the Exchange Act; and further provided that, without the approval of the Company s stockholders, the Plan may not be modified, altered or amended in any manner to (i) increase the number of Restricted Stock Units that may be the subject of Awards under the Plan (except for adjustments pursuant to Section 7.2), (ii) expand the types of Awards available under the Plan or (iii) expand the class of persons eligible to participate in the Plan.
- (b) The Company may, by action of the Board, terminate the Plan at any time and for any reason, subject to any requirement for stockholder approval imposed by applicable law, including the rules and regulations of the principal securities market on which the Shares are traded; provided, that the Plan may not be modified, altered or amended in any manner that would result in noncompliance with Rule 16b-3 of the Exchange Act.
- (c) No amendments to, or termination of, the Plan shall impair in any material respect the rights of a Participant under any Award previously granted without such Participant s consent except as required to comply with applicable securities laws or Code Section 409A.
- **7.2.** *Adjustments.* In the event of any merger, reorganization, consolidation, recapitalization, dividend or distribution (whether in cash, shares or other property, other than a regular cash dividend), stock split, reverse stock split, spin-off or similar transaction or other change in corporate structure affecting the Shares or the value thereof, such adjustments and other substitutions shall be made to the Plan and to Awards as the Committee deems equitable or appropriate taking into consideration the accounting and tax consequences, including such adjustments in the aggregate number, class and kind of securities that may be delivered under the Plan; provided, however, that no such adjustment or other substitution shall be made if it would cause the Plan or an Award to fail to comply with or be exempt from the requirements of Code Section 409A and provided, further, that the number of Shares subject to any Award shall always be a whole number.

**7.3.** *Transferability of Awards*. Except as provided below, no Award and no Shares subject to Awards that have not been issued or as to which any applicable restriction, performance or deferral period has not lapsed, may be sold, assigned, transferred, pledged or otherwise encumbered, other than by will or the laws of descent and

-6-

distribution, and such Award may be settled during the life of the Participant only by the Participant or the Participant s guardian or legal representative. To the extent and under such terms and conditions as determined by the Committee, a Participant may assign or transfer an Award (each transferee thereof, a Permitted Assignee ) to (a) the Participant s spouse, children or grandchildren (including any adopted and step children or grandchildren), parents, grandparents or siblings (the Covered Family Members ), (b) to a trust for the benefit of one or more of the Participant or the Covered Family Members, (c) to a partnership, limited liability company or corporation in which the Participant or one or more of the Covered Family Members are the only partners, members or stockholders, (d) for charitable donations or (e) pursuant to a domestic relations order entered or approved by a court of competent jurisdiction; provided, that such Permitted Assignee shall be bound by and subject to all of the terms and conditions of the Plan and the Award Agreement relating to the transferred Award and shall execute an agreement satisfactory to the Company evidencing such obligations; and provided further that such Participant shall remain bound by the terms and conditions of the Plan. The Company shall cooperate with any Permitted Assignee and the Company s transfer agent in effectuating any transfer permitted under this Section.

**7.4.** *Deferral.* The Committee shall be authorized to establish procedures pursuant to which the payment of any Award may be deferred. Any deferral shall only be allowed as is provided in a separate deferred compensation plan adopted by the Company. This Plan shall not constitute an employee benefit plan for purposes of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended.

#### 8. MISCELLANEOUS

- **8.1.** Award Agreements. The material conditions and terms of any Award granted under the Plan shall be set forth in an Award Agreement which shall contain provisions that are consistent with the provisions of the Plan. Each Award Agreement shall either be (a) in writing in a form approved by the Committee and executed by the Company by an officer duly authorized to act on its behalf, or (b) an electronic notice in a form approved by the Committee and recorded by the Company (or its designee) in an electronic recordkeeping system used for the purpose of tracking the Awards; in each case and if required by the Committee, the Award Agreement shall be executed or otherwise electronically accepted by the recipient of the Award in such form and manner as the Committee may require. The Committee may authorize any officer of the Company to execute any or all Award Agreements on behalf of the Company. Such Award Agreement shall either comply with, or be exempt from, the requirements of Code Section 409A.
- **8.2.** *Tax Withholding*. The Company shall have the right to make all payments or distributions pursuant to the Plan to a Participant who becomes an Inside Director (or a Permitted Assignee thereof) (any such person, a Payee) net of any applicable federal, state and local taxes required to be paid or withheld as a result of (a) the grant of any Award, (b) the delivery of Shares, (c) the lapse of any restrictions in connection with any Award or (d) any other event occurring pursuant to the Plan. The Company shall have the right to withhold from wages or other amounts otherwise payable to such Payee such withholding taxes as may be required by law, or to otherwise require the Payee to pay such withholding taxes. If the Payee shall fail to make such tax payments as are required, the Company or its Subsidiaries shall, to the extent permitted by law, have the right to deduct any such taxes from any payment of any kind otherwise due to such Payee or to take such other action as may be necessary to satisfy such withholding obligations. The Committee shall be authorized to establish procedures for election by Participants to satisfy such obligation for the payment of such taxes by tendering previously acquired Shares (either actually or by attestation, valued at their then Fair Market Value), or by directing the Company to retain Shares (up to the Participant's minimum required tax withholding rate or such other rate that will not trigger a negative accounting impact) otherwise deliverable in connection with the Award.
- **8.3.** Right of Discharge Reserved; Claims to Awards. Nothing in the Plan nor the grant of an Award hereunder shall confer upon any Director the right to continue in the service of the Company or any Subsidiary (in the capacity as a Director or otherwise) or affect any right that the Company may have to terminate the service of (or to demote or to exclude from future Awards under the Plan) any such Director at any time for any

reason. Except as specifically provided by the Committee, the Company shall not be liable for the loss of existing or potential profit from an Award granted in the event of termination of the Director s service.

- **8.4.** Substitute Awards. Notwithstanding any other provision of the Plan, the terms of an award issued in substitution for any outstanding Awards hereunder, may vary from the terms set forth in the Plan to the extent the Committee deems appropriate to conform, in whole or in part, to the provisions of the awards in substitution for which they are granted.
- **8.5.** Cancellation of Award. Notwithstanding anything to the contrary contained herein, an Award Agreement may provide that the Award shall be canceled if the Participant, without the consent of the Company, while providing services to the Company or any Subsidiary or after termination of such service, establishes a relationship with a competitor of the Company or any Subsidiary or engages in activity that is in conflict with or adverse to the interest of the Company or any Subsidiary, as determined by the Committee in its sole discretion. The Committee may provide in an Award Agreement that if within the time period specified in the Agreement the Participant establishes a relationship with a competitor or engages in an activity referred to in the preceding sentence, the Participant will forfeit any gain realized on the vesting of the Award and must repay such gain to the Company.
- **8.6.** Stop-Transfer Orders. All Shares delivered under the Plan pursuant to any Award shall be subject to such stop-transfer orders and other restrictions as the Committee may deem advisable under the rules, regulations and other requirements of the Securities and Exchange Commission, any stock exchange upon which the Shares are then listed, and any applicable federal or state securities law, and the Committee may cause a legend or legends to be put on any such Shares (or certificates for such Shares, if applicable) to make appropriate reference to such restrictions.
- **8.7.** *Nature of Payments.* All Awards made pursuant to the Plan are in consideration of services performed or to be performed for the Company. Any income or gain realized pursuant to Awards under the Plan constitutes a special incentive payment to the Participant and shall not be taken into account, to the extent permissible under applicable law, as compensation for purposes of any of the employee benefit plans of the Company or any Subsidiary except as may be determined by the Committee.
- **8.8.** Other Plans. Nothing contained in the Plan shall prevent the Board from adopting other or additional compensation arrangements, subject to stockholder approval if such approval is required; and such arrangements may be either generally applicable or applicable only in specific cases.
- **8.9.** Severability. If any provision of the Plan shall be held unlawful or otherwise invalid or unenforceable in whole or in part by a court of competent jurisdiction, such provision shall (a) be deemed limited to the extent that such court of competent jurisdiction deems it lawful, valid and/or enforceable and as so limited shall remain in full force and effect, and (b) not affect any other provision of the Plan or part thereof, each of which shall remain in full force and effect. If the making of any payment or the provision of any other benefit required under the Plan shall be held unlawful or otherwise invalid or unenforceable by a court of competent jurisdiction, such unlawfulness, invalidity or unenforceability shall not prevent any other payment or benefit from being made or provided under the Plan, and if the making of any payment in full or the provision of any other benefit required under the Plan in full would be unlawful or otherwise invalid or unenforceable, then such unlawfulness, invalidity or unenforceability shall not prevent such payment or benefit from being made or provided in part, to the extent that it would not be unlawful, invalid or unenforceable shall be made or provided under the Plan.
- **8.10.** Construction. As used in the Plan, the words include and including, and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words without limitation.

**8.11.** *Unfunded Status of the Plan.* The Plan is intended to constitute an unfunded plan for incentive compensation. With respect to any payments not yet made to a Participant by the Company, nothing contained

herein shall give any such Participant any rights that are greater than those of a general creditor of the Company. In its sole discretion, the Committee may authorize the creation of trusts or other arrangements to meet the obligations created under the Plan to deliver the RSU Shares with respect to Awards hereunder; provided, however, that the existence of such trusts or other arrangements is consistent with the unfunded status of the Plan.

- **8.12.** *Governing Law.* The Plan and all determinations made and actions taken thereunder, to the extent not otherwise governed by the Code or the laws of the United States, shall be governed by the laws of the State of Texas, without reference to principles of conflict of laws, and construed accordingly.
- **8.13.** Effective Date of Plan; Termination of Plan. The Plan shall be effective on the date of the approval of the Plan by the holders of the shares entitled to vote at a duly constituted meeting of the stockholders of the Company. The Plan shall be null and void and of no effect if the foregoing condition is not fulfilled and in such event, each Award shall, notwithstanding any of the preceding provisions of the Plan, be null and void and of no effect. Awards may be granted under the Plan at any time and from time to time on or prior to the tenth anniversary of the effective date of the Plan, on which date the Plan will expire except as to Awards then outstanding under the Plan. Such outstanding Awards shall remain in effect until they have been terminated or have expired.
- **8.14.** Compliance with Code Section 409A. This Plan is intended to shall either comply with, or be exempt from, Code Section 409A and shall be administered in a manner that is intended to comply with, or be exempt from, Code Section 409A and shall be construed and interpreted in accordance with such intent. To the extent that an Award or the payment, settlement or deferral thereof is subject to Code Section 409A, the Award shall be granted, paid, settled or deferred in a manner that will comply with Code Section 409A, including regulations or other guidance issued with respect thereto, except as otherwise determined by the Committee. Any provision of this Plan that would cause the grant of an Award or the payment, settlement or deferral thereof to fail to satisfy Code Section 409A shall be amended to comply with Code Section 409A on a timely basis, which may be made on a retroactive basis, in accordance with regulations and other guidance issued under Code Section 409A. Further, if, on the date of a Participant s separation from service (as defined in Code Section 409A) other than upon the occurrence of death, the Participant is a specified employee (as defined in Code Section 409A), no payments of any amounts hereunder that are subject to Code Section 409A shall be made until the earliest date on which payment is permissible under 409A(a)(2)(B)(i) (the six (6) month delay rule for specified employees).
- **8.15.** Captions. The captions in the Plan are for convenience of reference only, and are not intended to narrow, limit or affect the substance or interpretation of the provisions contained herein.

## **APPENDIX C**

# MICROTUNE, INC. 2010 EMPLOYEE STOCK PURCHASE PLAN

(Effective as of April 30, 2010)

Subject to Stockholder Approval

## TABLE OF CONTENTS

ARTICL	<u>LE I PURPOSE AND COMMENCEMENT</u>	1	
1.1	<u>Purpose</u>	1	
1.2	Commencement	1	
<u>ARTICI</u>	<u>LE II DEFINITIONS</u>	2	
2.1	<u>Definitions</u>	2	
ARTICI	E III ELIGIBILITY AND PARTICIPATION	5	
3.1	<u>Eligibility</u>	5	
3.2	<u>Limitations</u>	5	
3.3	<u>Participation</u>	5	
3.4	<u>Termination of Employment or Death</u>	6	
ARTICLE IV PURCHASE OF COMMON STOCK			
4.1	<u>Determination and Purchase of Common Stock</u>	7	
4.2	Notice of Purchase; Stock Delivery	7	
4.3	Notification of Disposition of Stock	8	
<u>ARTICI</u>	<u>LE V MISCELLANEOUS PROVISIONS</u>	9	
5.1	Shares Subject to Plan; Adjustments	9	
5.2	Administration of the Plan	9	
5.3	Amendment of the Plan	11	
5.4	<u>Termination of the Plan</u>	11	
5.5	Governing Law; Compliance With Law	11	
5.6	No Assignment	12	
5.7	No Contract of Employment	12	
5.8	No Rights as Stockholder	12	

i

#### ARTICLE I

#### PURPOSE AND COMMENCEMENT

#### 1.1 Purpose

The purpose of the Plan is to provide the eligible employees of the Company and its Subsidiaries with an added incentive to continue in their employment and to encourage increased efforts to promote the best interests of the Company by permitting eligible employees to purchase shares of Common Stock at prices less than the current market price thereof. The Plan is intended to qualify as an employee stock purchase plan under section 423 of the Code and will be interpreted and construed in accordance with such purpose.

#### 1.2 Commencement

By this instrument, the Company desires to adopt the Plan effective April 30, 2010, subject to the approval of the Company s stockholders on or before October 29, 2010. In the event that the Company s stockholders do not approve of the Plan on or before October 29, 2010, the provisions of this Plan will be null and void, and any contributions made by Participants to purchase shares of Common Stock will be refunded to such Participants as provided herein.

1

#### ARTICLE II

#### DEFINITIONS

#### 2.1 Definitions

As used in the Plan, the following terms and phrases will have the following meanings:

- (a) Additional Shares means the shares of Common Stock which may become available for purchase under the Plan, pending Stockholder Approval.
- (b) Authorized Shares means any Additional Shares for which Stockholder Approval has been obtained.
- (c) **Board** means the board of directors of the Company.
- (d) Closing Market Price means (i) if the Common Stock is traded on a national securities exchange, the Closing Market Price will be the closing price reported by the applicable composite transactions report on the applicable Trading Day, or (ii) if the foregoing provision is inapplicable, the Closing Market Price will be determined by the Committee in good faith on such basis as it deems appropriate.
- (e) **Code** means the Internal Revenue Code of 1986, as amended.
- (f) Commencement Date means the first Trading Day of an Offering Period.
- (g) **Committee** means the Compensation Committee of the Board or a subcommittee thereof formed by the Compensation Committee to act as the Committee hereunder.
- (h) **Common Stock** means the common stock of the Company.
- (i) **Company** means Microtune, Inc., a Delaware corporation.
- (j) **Contribution Account** means the bookkeeping account established on behalf of a Participant pursuant to Article III to which will be credited his or her Participant Contributions reduced by the amount of any funds used for purchases of shares of Common Stock pursuant to Section 4.1, including, without limitation, conditional purchases of Additional Shares (whether or not such shares have been delivered).
- (k) **Contribution Rate** means the percentage of a Participant s Covered Compensation during each payroll period designated by each such Participant to be contributed by regular payroll deductions to his or her Contribution Account as set forth in Section 3.3.
- (l) Covered Compensation means:

- (i) The entire amount paid to an Employee by a Sponsoring Employer for the performance of duties including base salaries, wages paid on an hourly or other time basis, commissions, overtime and certain other amounts of cash compensation paid during the Offering Period, excluding bonuses, foreign service pay, hardship withdrawal allowances and any other pay intended to reimburse the Employee for the higher cost of living outside the United States, awards and payments under the Microtune, Inc. 2010 Stock Plan, automobile allowances, housing allowances, relocation payments, deemed income, income payable under the stock incentive plans, Christmas gifts, insurance premiums and other imputed income, pensions, retirement benefits, prizes or awards.
- (ii) The entire amount paid to an Employee by a Sponsoring Employer on account of a period of time during which no duties are performed, including salaries, wages paid on an hourly or other time basis, commissions and salary or wage continuation paid during vacation, holiday, illness, jury duty, military duty or leave of absence, regardless of the form of payment, **excluding** (A) any payments made or due under a plan maintained solely for the purpose of complying with workers compensation or unemployment compensation or disability insurance laws, (B) any payment which solely reimburses the Employee for expenses incurred by the Employee, (C) severance pay, or (D) imputed income.

- (iii) For purposes of Subparagraphs (i) and (ii) above, Covered Compensation for any Offering Period will also include amounts described in Subparagraph (i) and (ii) which are deferred by a Participant under the Microtune, Inc. 401(k) Plan, as amended, in accordance with section 401(a) of the Code, under a cafeteria plan maintained by the Company or a Subsidiary in accordance with section 125 of the Code, or under the Company s deferred compensation arrangements.
- (m) **Effective Date** means April 30, 2010; provided, however, if Stockholder Approval is not obtained on or before October 29, 2010, the provisions of this Plan will be of no effect.
- (n) **Employee** means each employee of a Sponsoring Employer whose customary employment is at least twenty (20) hours a week and more than five (5) months in a calendar year and who has reached the age of majority in his or her state of residence. For purposes of the Plan, employment will be determined in accordance with the provisions of section 1.421-1(h) of the Treasury Regulations (or any successor regulations).
- (o) **Offering Period** means the following periods:
  - (i) The period commencing on May 1 and terminating on October 31, provided, however, (A) if May 1 is not a Trading Day, the period shall commence on the first Trading Day thereafter, and (B) if October 31 is not a Trading Day, the period shall end on the first Trading Day prior thereto.
  - (ii) The period commencing on November 1 and terminating on April 30, provided, however, (A) if November 1 is not a Trading Day, the period shall commence on the first Trading Day thereafter, and (B) if April 30 is not a Trading Day, the period shall end on the first Trading Day prior thereto.
- (p) **Participant** means any Employee of a Sponsoring Employer who has met the conditions and provisions for becoming a Participant set forth in Article III.
- (q) Participant Contributions means the aggregate dollars actually contributed by a Participant to his or her Contribution Account.
- (r) Plan means the Microtune, Inc. 2010 Employee Stock Purchase Plan as set forth herein, as it may be amended from time to time.
- (s) **Plan Administrator** means the individual(s), if any, appointed by the Committee to handle the day-to-day administration of the Plan on a ministerial basis as specified in Section 5.2(b); provided, that, during any period in which no such individual(s) are appointed by the Committee, the Committee shall serve as the Plan Administrator.
- (t) **Permanent Disability** means an illness, injury or other physical or mental condition of an Employee that continues for at least one hundred and eighty (180) consecutive days and results in the Employee s inability to provide in all material respects the duties previously performed in his or her capacity as an Employee of a Sponsoring Employer.
- (u) **Prior Plan** means the Amended and Restated Microtune, Inc. 2000 Employee Stock Purchase Plan.
- (v) **Purchase Date** means the last Trading Day of an Offering Period.
- (w) **Purchase Price** means the purchase price for a share of Common Stock to be paid by a Participant on a Purchase Date, which will equal eighty-five percent (85%) of the Closing Market Price on the Purchase Date or the Commencement Date, whichever is lower, with respect to such Offering Period.

- (x) **Request for Participation** means the form prescribed by the Plan Administrator for distribution to Employees in connection with participation in the Plan.
- (y) **Stockholder Approval** means the actual approval by the Company s stockholders of the shares of Common Stock available for purchase under Section 5.1.
- (z) **Sponsoring Employers** means the Company and each Subsidiary that has been designated by the Committee as a Sponsoring Employer under the Plan.

- (aa) **Subsidiary** means a subsidiary of the Company which is treated as a subsidiary corporation under section 424(f) of the Code.
- (bb) **Total Shares** means the Authorized Shares plus any Additional Shares that may be made available for purchase under the Plan at a subsequent date.
- (cc) Trading Day means a day on which national stock exchanges and the NASDAQ system are open for trading.

4

#### ARTICLE III

#### ELIGIBILITY AND PARTICIPATION

#### 3.1 Eligibility

Each Employee will become eligible to be a Participant in the Plan and may participate in the Plan as of the Commencement Date of an Offering Period if such Employee is an Employee on such Commencement Date. The date on which Employees may become eligible to become Participants of the Plan may be amended, from time to time, by the Committee.

#### 3.2 Limitations

Notwithstanding anything to the contrary contained in the Plan, no right to purchase Common Stock will accrue under the Plan in favor of any person who is not an Employee eligible to participate in the Plan under Section 3.1, and no Employee will acquire the right to purchase shares of Common Stock:

- (a) If immediately after receiving such right to purchase Common Stock, such Employee would own five percent (5%) or more of the total combined voting power or value of all classes of stock of the Company or any Subsidiary, taking into account in determining stock ownership any stock attributable to such Employee under section 424(d) of the Code;
- (b) If to do so would permit such Employee s right to purchase stock under all employee stock purchase plans (to which section 423 of the Code applies) of the Company and its Subsidiaries, as those plans are in effect from time to time, to accrue at a rate which exceeds Twenty-Five Thousand Dollars (\$25,000) of fair market value of such stock (as determined as of each Commencement Date) for each calendar year, all as specified in the manner provided by section 423(b)(8) of the Code; or
- (c) If to do so would permit such Employee the right to purchase more than Five Thousand (5,000) shares (or such other number as may be determined in advance for any Offering Period by the Compensation Committee) of Common Stock in any Offering Period.

#### 3.3 Participation

(a) Summary of Plan and Request for Participation Form. Each Employee eligible to be a Participant in the Plan may enroll by completing a Request for Participation and filing it with the Plan Administrator prior to a Commencement Date of an Offering Period. The completed Request for Participation must indicate the Contribution Rate authorized by the Participant. If any Employee does not elect to participate in the Plan during any given Offering Period, such Employee may elect to participate on any future Commencement Date so long as he or she continues to be an eligible Employee. An eligible Employee who enrolls in the Plan will be furnished a summary of the Plan by the Plan Administrator.

Notwithstanding the foregoing, an Employee that is participating in the Prior Plan as of April 30, 2010, will be deemed as having delivered a Request for Participation with the Plan Administrator prior to the first Commencement Date under the Plan based on his or her elections set forth in the applicable subscription agreement that relates to such Prior Plan, and, subject to the discretion of the Plan Administrator, such elections will remain in effect until such time (i) as he or she delivers a new Request for Participation to the Plan Administrator in accordance with this Article III or (ii) the applicable terms and conditions of his or her participation otherwise change pursuant to the Plan.

(b) Payroll Deduction Authorization. On his or her Request for Participation, an Employee must authorize his or her Sponsoring Employer to deduct through a payroll deduction the amount of such Employee s Participant Contribution. The payroll deduction specified in a Request for Participation for each payroll period will be at a Participant Contribution Rate of no more than fifteen percent (15%) of such Employee s Covered Compensation during such payroll period paid to him or her by his or her Sponsoring Employer. Such deductions will begin as of the first pay period ending after the Commencement Date of an Offering Period. Participant Contributions will not be permitted to begin at

any time other than immediately after the Commencement Date of an Offering Period. Participant Contributions will be credited to the Participant s Contribution Account. A Participant s Contribution Rate specified in the Request for Participation will remain in effect for subsequent Offering Periods. No interest will accrue on amounts credited to Participant Contribution Accounts, unless and until the Committee approves such accrual of interest on terms that it specifies and applies on a uniform basis as to all Participants.

- (c) Changes in Contribution Rate. The Participant s Contribution Rate, once established, will remain in effect for all Offering Periods unless changed by the Participant in an updated Request for Participation delivered to the Plan Administrator prior to the Commencement Date of the next Offering Period. Such change in the Participant s Contribution Rate will be effective on the Commencement Date of the next Offering Period. Subject to Section 3.3(d) below, a Participant s Contribution Rate for an Offering Period may not be increased, decreased or otherwise modified at any time during such Offering Period.
- (d) **Discontinuation of Contributions**. A Participant may notify the Plan Administrator of such Participant s desire to discontinue his or her Participant Contributions by delivering to the Plan Administrator written notice on such forms as may be provided by the Plan Administrator. If the Participant desires to cease Participant Contributions for the current Offering Period, the Participant must indicate that on the form and such form must be returned to the Plan Administrator before the last day of the current Offering Period. Upon such request, the entire cash balance in the Participant s Contribution Account will be refunded to him or her as soon as practicable. Alternatively, the Participant may elect to cease Participant Contributions for the next Offering Period by so indicating on the form and returning the form to the Plan Administrator before the last day of the current Offering Period, in which case the balance in his Participant Contribution Account will be used to purchase shares of Common Stock in accordance with Article IV but no Participant Contributions or stock purchases will be made following the current Offering Period unless the Participant reenrolls in the Plan.

If a Participant discontinues his or her Participant Contributions pursuant to the foregoing paragraph, (i) such Participant will be terminated from the Plan effective upon the date specified in the Participant s notice to the Plan Administrator, and (ii) subject to applicable law, such Participant will not be permitted to resume participation in the Plan until such subsequent Offering Period for which he or she files a Request for Participation with the Plan Administrator in accordance with Section 3.3(a) above.

In the event a Participant s payroll deduction is prevented by legal process, the Participant will be deemed to have terminated from the Plan and the balance in such Participant s Participant Contribution Account will either be refunded or used to purchase additional shares of Common Stock depending on whether such termination of payroll deductions occurs on or before or after the last day of the relevant Offering Period (i.e., if the termination is on or before the Purchase Date of the relevant Offering Period, the balance in the Participant s Participant Contribution Account will be refunded and if the termination is after the Purchase Date of the relevant Offering Period, the balance in his Participant Contribution Account will be used to purchase shares of Common Stock in accordance with Article IV).

(e) Establishment of Brokerage Account. By enrolling in the Plan, each Participant will be deemed to have authorized the establishment of a brokerage account in his or her name at a securities brokerage firm or other financial institution selected by the Committee in its discretion.

#### 3.4 Termination of Employment or Death

Any Participant (a) whose employment by a Sponsoring Employer is terminated for any reason (including death or retirement), or (b) who ceases to be an Employee under the Plan, will cease being a Participant as of the date of such termination of employment or cessation of Employee status, and the entire balance credited to the Participant s Contribution Account will be refunded to him or her (or, if applicable, his or her estate or beneficiary as set forth in the Request for Participation) as soon as practicable.

#### ARTICLE IV

#### PURCHASE OF COMMON STOCK

#### 4.1 Determination and Purchase of Common Stock

- (a) Purchase on Each Purchase Date. On each Purchase Date each Participant s Contribution Account will be used to purchase the maximum number of shares of Common Stock permitted (subject to the limits of Section 3.2 and the adjustment provisions of Section 4.1(b)) determined by dividing (i) the Participant s Contribution Account as of such Purchase Date by (ii) the Purchase Price in respect of such Offering Period; provided, however, that the delivery of any Additional Shares is contingent upon Stockholder Approval. If Stockholder Approval is not obtained with respect to any such Additional Shares, the balance of the Participant s Contribution Account as of each such Purchase Date that was applied or allocated to the purchase of Additional Shares will be refunded to the Participant as soon as administratively practicable. Only whole shares of Common Stock may be purchased by the Participant under the Plan, and any fractional shares of Common Stock that are not eligible for purchase will be paid to the Participant in cash.
- (b) Adjustment of Shares. If, in any Offering Period, the total number of shares of Common Stock to be purchased pursuant to the Plan by all Participants exceeds the Total Shares, then each Participant will purchase his or her pro rata portion of the shares of Common Stock remaining available under the Plan based on the ratio that (i) the balance of each Participant s Contribution Account as of the Purchase Date bears to (ii) the total balance of all Participants Contribution Accounts as of the Purchase Date; provided, however, that, in no event, will any fractional shares of Common Stock be issued pursuant to the Plan or this Section 4.1(b). For purposes of this Section 4.1, the shares of Common Stock available for purchase under the Plan will be subject to an ordering rule pursuant to which the Authorized Shares will be purchased first and only when all such shares have been purchased will Participant Contributions be used to purchase the Additional Shares.

## 4.2 Notice of Purchase; Stock Delivery

- (a) **Notice of Purchase**. A report will be sent at least annually by the Plan Administrator, or the agent designated pursuant to Section 5.2(d), to each Participant stating (i) the entries made to his or her Contribution Account (including the cash balance of such Contribution Account), (ii) the number of shares of Common Stock purchased, including with respect to the Additional Shares prior to Stockholder Approval that delivery of such shares of Common Stock is contingent upon such Stockholder Approval, and (iii) the applicable Purchase Price.
- (b) **Stock Delivery**. As promptly as practicable after each Purchase Date, the Company will arrange for the delivery to each Participant in his or her designated brokerage account (or as otherwise designated by the Participant, subject to the approval of the Plan Administrator), the shares of Common Stock purchased under the Plan by each such Participant; provided, that no such delivery will occur with respect to any purchase of Additional Shares until Stockholder Approval is obtained. Shares of Common Stock to be delivered hereunder will be registered in the name of the Participant or in the name of the Participant and his or her spouse.
- (c) The Company will seek Stockholder Approval with respect to the Additional Shares and in the event the Company is required to obtain from any commission or agency authority to issue any such shares, the Company will seek to obtain such authority. Inability of the Company to obtain Stockholder Approval with respect to the Additional Shares, to obtain from any such commission or agency authority or to take such other actions, which the Committee or counsel for the Company deems necessary or appropriate for the lawful issuance of any such shares, will relieve the Company and the Committee from liability to any Participant in the Plan except to return to him or her the amount of the balance in his or her Participant Contribution Account with respect to the shares at issue. In addition, neither the Company nor the Committee will have any liability with respect to a delay in the delivery of any shares of Common Stock pursuant to this paragraph.

## 4.3 Notification of Disposition of Stock

If a Participant or former Participant disposes of a share of Common Stock purchased under the Plan prior to two (2) years after the Commencement Date of the Offering Period during which such share was purchased, then such Participant or former Participant will notify the Plan Administrator immediately of such disposition in writing.

#### ARTICLE V

#### MISCELLANEOUS PROVISIONS

#### 5.1 Shares Subject to Plan; Adjustments

- (a) **Maximum Number of Shares**. Effective as of April 30, 2010, the maximum number of shares of Common Stock which may be purchased under the Plan is one million (1,000,000) shares (which shares shall be Additional Shares until such time as Stockholder Approval, if any, is obtained, at which time such shares shall become Authorized Shares), subject, however, to adjustment as hereinafter set forth. The shares of Common Stock to be purchased under the Plan will be made available, at the discretion of the Board or the Committee, either from authorized but unissued shares of Common Stock or from previously issued shares of Common Stock reacquired by the Company, including shares purchased on the open market.
- (b) Adjustment of Shares. If the outstanding shares of Common Stock of the Company are increased, decreased, or exchanged for a different number or kind of shares or other securities, or if additional shares or new or different shares or other securities are distributed with respect to such shares of Common Stock or other securities, through merger, consolidation, spin off, sale of all or substantially all of the property of the Company, reorganization, recapitalization, reclassification, stock dividend, stock split, reverse stock split or other distribution with respect to such shares of Common Stock, or other securities, an appropriate and proportionate adjustment may be made in the maximum number and kind of shares provided in Section 5.1(a), and the Company or its successor, in its discretion, may modify any Offering Period with respect to such Common Stock, subject in the case of certain corporate reorganizations to the requirements of section 424(a) of the Code.

#### 5.2 Administration of the Plan

(a) **Responsible Party**. The Committee will be responsible for the administration of the Plan including, but not limited to, the determination of eligibility to participate in the Plan and limitations on the number of shares of Common Stock eligible for purchase under the Plan. In carrying out such responsibilities, the Committee will have the discretionary authority to interpret and construe the Plan and determine all questions arising in the administration, application and operation of the Plan, including all questions of fact and all questions of interpretation of the provisions of the Plan and will correct any defect or supply any omission or reconcile any inconsistency in the Plan in the manner and to the extent that the Committee deems desirable to carry out the terms of the Plan. All such determinations by the Committee will be conclusive and binding on all persons. The Committee, from time to time, may adopt, amend and rescind rules and regulations not inconsistent with the Plan for carrying out the administration of the Plan, and may approve the forms of any documents or writings provided for in the Plan.

The Committee will have full discretionary authority to delegate certain of its administrative functions under this Section 5.2(a) to the Plan Administrator, who will have the responsibility and duty to administer the Plan on a daily basis. In addition, the Committee or, if applicable, the Plan Administrator will have full discretionary authority to delegate ministerial functions in the administration of the Plan to employees of the Company.

Notwithstanding the foregoing, to the extent, and during such periods that, the Committee is unavailable or otherwise unable to satisfy its responsibilities set forth in this Section 5.2(a) to the Plan (or any other provision under the Plan), the Board or its delegate shall assume such responsibilities until such time as the Committee again becomes available to resume satisfying such responsibilities.

(b) **Delegation of Duties**. The Committee may delegate certain of its administrative functions under Section 5.2(a) to the Plan Administrator as set forth in this document.

(i)	<b>Duties of the Plan Administrator</b> . If the Committee delegates administrative functions to the Plan Administrator, the Plan Administrator will have the following administrative responsibilities, except to the extent, and during such periods that, the Committee reserves any of the responsibilities for itself:		
	(A)	To determine an Employee s eligibility to participate in the Plan;	
	(B)	To provide eligible Employees with a summary of the provisions of the Plan and a Request for Participation and to prescribe the procedures to be followed by such Employees in order to commence participation in the Plan;	
	(C)	To maintain Participant Contribution Accounts pursuant to Article III;	
	(D)	To determine the number of shares of Common Stock that may be purchased by each Participant as of each Purchase Date;	
	(E)	To implement a Participant s cessation of Participant Contributions and the associated suspension in Plan participation pursuant to Section 3.3(d);	
	(F)	To process withdrawals with respect to a Participant s Contribution Account upon the Participant s termination;	
	(G)	To refund the balance of a Participant s Contribution Account upon the Participant s death;	
	(H)	To determine the proportionate number of shares of Common Stock that may be purchased by each Participant pursuant to Section 4.1;	
	(I)	To receive notice from each Participant of a disposition of shares of Common Stock purchased under the Plan within two (2) years as provided under Section 4.3;	
	(J)	To receive such information from the Company, the Sponsoring Employers and Participants as may be necessary in order to administer the Plan;	
	(K)	To furnish the Company with information that the Company may require for tax or other purposes;	
	(L)	To engage the service of counsel (who may, if appropriate, be counsel for the Company) and agents whom it may deem advisable to assist it with the performance of its duties;	
	(M)	To establish and maintain, or cause to be maintained, the individual accounts described in Section 4.2;	
	(N)	To create and maintain such records and forms as are required for the efficient administration of the Plan;	
	(O)	To comply with applicable federal, state and local tax withholding requirements;	
	(P)	To enforce the terms of the Plan and any rules and regulations adopted by the Committee;	

- (Q) To implement any rules or restrictions imposed by the Committee pursuant to Section 5.2(e) regarding section 16 of the Securities and Exchange Act of 1934;
- (R) To utilize the services of third parties to carry out the duties specified in this Section 5.2(b)(i) to the Plan; and
- (S) To comply with all applicable reporting and disclosure obligations imposed on the Plan.

The foregoing list of express powers is not intended to be either complete or conclusive, and the Plan Administrator, if applicable, will, in addition, have such powers as it may reasonably determine to be necessary or appropriate in the performance of its powers and duties under the Plan.

Notwithstanding the foregoing, to the extent, and during such periods that, the Plan Administrator is unavailable or otherwise unable to satisfy its responsibilities set forth in this Section 5.2(b)(i) to the Plan (or any other provision under the Plan), the Committee or its delegate shall assume such responsibilities until such time as the Plan Administrator again becomes available to resume satisfying such responsibilities.

- (c) Liability. No member of the Board, the Committee, or the Plan Administrator will be liable for any action, determination or omission taken or made in good faith with respect to the Plan or any right granted under the Plan.
- (d) **Custodial and Recordkeeping Services**. The Committee may in its discretion engage a bank trust department, securities brokerage firm or other financial institution as agent to perform custodial and recordkeeping functions for the Plan, such as holding record title to the Participant s shares, maintaining an individual investment account for each Participant and providing periodic account status reports to Participants.
- (e) **Exchange Act Requirements.** The Committee will have the authority to adopt and enforce such special rules and restrictions under the Plan to be applicable to Participants who are subject to section 16 of the Securities and Exchange Act of 1934, as amended, as the Committee will deem are necessary or appropriate to exempt certain Plan transactions from the requirements of such section 16.
- (f) Costs of Administration. The Company and the Sponsoring Employers will bear the cost of administering the Plan, including any fees, costs and expenses relating to the purchase of shares of Common Stock under the Plan. Notwithstanding the foregoing, Participants will be responsible for all fees, costs and expenses incurred in connection with the disposition of shares of Common Stock purchased under the Plan.

#### 5.3 Amendment of the Plan

The Committee and the Board will have the authority to modify, alter or amend the Plan at any time and from time to time to any extent that it may deem advisable, including, without limiting the generality of the foregoing, any amendment deemed necessary to ensure compliance of the Plan with section 423 of the Code. Notwithstanding the foregoing, no amendment of the Plan will operate to reduce any amounts previously allocated to a Participant s Contribution Account nor to reduce a Participant s rights with respect to shares of Common Stock previously purchased and held on his or her behalf under the Plan.

#### 5.4 Termination of the Plan

The Company may, by action of the Board, terminate the Plan at any time and for any reason. The Plan will automatically terminate upon the purchase by Participants of all shares of Common Stock subject to the Plan under Section 5.1, unless such number of shares is increased by the Board and such increase is approved by the stockholders of the Company. Upon termination of the Plan, as soon as practicable, each Participant will be refunded the entire cash balance in his or her Contribution Account and forwarded all shares of Common Stock otherwise held under the Plan for the account of such Participant. The Board may suspend operation of the Plan for any period as it may deem advisable. Unless terminated sooner pursuant to the foregoing, the Plan will remain in effect until the tenth anniversary of the Effective Date, on which date the Plan will terminate.

## 5.5 Governing Law; Compliance With Law

The Plan will be construed in accordance with the laws of the State of Texas. The Company s obligation to sell and deliver shares of Common Stock hereunder will be subject to all applicable federal and state laws, rules and regulations and to such approvals by any regulatory or governmental agency as may, in the opinion of counsel for the Company, be required. The Company may make such provisions as it may deem appropriate for the withholding of any taxes or payment of any taxes which it determines it may be required to withhold or pay in connection with a Participant s participation in the Plan.

#### 5.6 No Assignment

The purchase rights granted hereunder are not assignable or transferable by the Participants, other than by will or the laws of descent and distribution, and are exercisable during the Participant s lifetime only by the Participant. Any attempted assignment, transfer or alienation not in compliance with the terms of the Plan will be null and void for all purposes and respects.

#### 5.7 No Contract of Employment

The Plan will not be deemed to constitute a contract between a Sponsoring Employer and any Participant nor to be consideration nor an inducement for the employment of any Participant or Employee. Nothing contained in the Plan will be deemed to give any Participant or Employee the right to be retained in the service of a Sponsoring Employer or to interfere with the right of a Sponsoring Employee to discharge any Participant or Employee at any time regardless of the effect which such discharge will have upon him or her as a Participant of the Plan.

#### 5.8 No Rights as Stockholder

No eligible Employee or Participant will by reason of participation in the Plan have any rights of a stockholder of the Company until he or she acquires shares of Common Stock as herein provided. No stockholder rights will apply with respect to any Additional Shares purchased by a Participant until Stockholder Approval has been obtained.